

# City of Sidney, MT City Council Regular Meeting 4-21-2025 April 21, 2025 6:30 PM 115 2nd Street SE | Sidney, MT 59270

The City Council meetings are open to the public attending in person, with masks encouraged when social distancing cannot be accomplished. If the public does not wish to participate in person, they are also invited to participate via a Zoom meeting. You can participate via phone:

Meeting ID: 713 080 5898 Passcode: 4332809 Call: 1-346-248-7799

- 1. Call to Order
- 2. Pledge of Allegiance
- 3. Aldermen Present
- 4. Correction or Approval of Minutes
  - a. City Council Regular Meeting Minutes- 4-7-2025
- 5. Visitors
  - a. Scotty Strugis- Baseball Camp May 10th, 2025, in Moose Park
  - b. David Seitz- Seitz Insurance Workers Comp Policy
  - c. Larry Christensen- Sportsman Club
- 6. Public Hearing
  - <u>a.</u> Amended Plat-Homestead Subdivision Phase 2 (tabled at 3-17-25 meeting) (Tabled until the 4-21-2025 Meeting)
- 7. Mayor Norby
- 8. Committee Meeting Work
- 9. Alderman Requests and Committee Reports

<u>Budget and Finance</u> – Chairman Christensen – DiFonzo, Koffler. <u>Police and Fire</u> – Chairman DiFonzo – Rasmussen, Larson. <u>Parks and Recreation</u> – Chairman Christensen – Larson, Rasmussen. <u>Water</u> and Sewer – Chairman Godfrey – Christensen, DiFonzo. <u>Street and Alley</u> – Chairman Rasmussen –

#### **Unfinished Business** 10.

## 11. New Business

- a. Cory Wieland- Variance Meeting
- b. Growth Policy Kick-off Coordination

# 12. City Planner

a. Homestead Subdivision Phase 2 Amended Plat

# 13. City Attorney

- a. Tax Exempt Bond Reimbursement Resolution
- b. Contract for B&B Builders for Facilites and Grounds Maintenace
- c. Second Reading of Ordinance 601- Amending Snow Removal Code
- d. Second Reading of Ordinance 603- Amending Zoning Code

# 14. Chief of Police

#### 15. **Public Works Director**

- a. March 2025 Compliance Officer Report
- b. Morrison Maierle- Meadows Village Storm Water Project

# 16. Fire Marshal/Building Inspector

a. March 2025 Fire Department Run Report

# 17. City Clerk/Treasurer

- a. Quarter 1 (January through March 2025) Water & Sewer Bank Transfer
- b. March 2025 JV Report
- c. March 2025 Treasurer's Report
- d. FY23-24 Audit Planning and Governance Letter and Auditor Bob Denning Retiring

#### 18. Consent Agenda

<u>a.</u> Claims to be approved: \$6,017.97 +\$148,468.93 = \$154,486.90

#43795 CARDMEMBER SERVICE \$ 6,017.97

<u>b.</u> Building Permits to be approved: 2025-057 and RC2025-20 & RC2025-24 & RC2025-25 & RC2025-26

# 19. Adjournment



# City of Sidney, MT City Council Regular Meeting 4-7-25 April 07, 2025 6:30 PM 115 2nd Street SE | Sidney, MT 59270

The City Council meetings are open to the public attending in person, with masks encouraged when social distancing cannot be accomplished. If the public does not wish to participate in person, they are also invited to participate via a Zoom meeting. You can participate via phone:

Meeting ID: 713 080 5898 Passcode: 4332809 Call: 1-346-248-7799

# 1. Call to Order

Mayor Norby called the regular meeting of the Sidney City Council to order at 6:30pm.

# 2. Pledge of Allegiance

The Pledge of Allegiance was stated by all present.

## 3. Aldermen Present

Christensen, Buxbaum, Larson, Koffler, and DiFonzo. Rassmussen via Zoom. Absent: None

# 4. Correction or Approval of Minutes

a. City Council Meeting from 03-17-2025

Motion was made to approve the minutes.

Motion made by Alderman Koffler, Seconded by Alderwoman Buxbaum. Voting Yea: Alderman Koffler, Alderwoman Rasmussen, Alderman DiFonzo, Alderwoman Buxbaum, Alderwoman Christensen, Alderwoman Larson

b. Water and Sewer Committe Meeting 03-24-2025

Motion was made to approve the minutes.

Motion made by Alderwoman Christensen, Seconded by Alderman DiFonzo. Voting Yea: Alderman Koffler, Alderwoman Rasmussen, Alderman DiFonzo, Alderwoman Buxbaum, Alderwoman Christensen, Alderwoman Larson

# 5. Visitors

Jordan Mayer (Interstate Engineering), Jody Wells (The Roundup), Tom Kalil (City Attorney via Zoom), and Sidney High School Government class students: Kylie, Audree, Lilah, Hallie, Clementine, and Lance.

# 6. Public Hearing

a. Public Hearing Homestead Subdivision Amended Plat (tabled at 3-17-25 meeting) (<u>Tabled until</u> the 4-21-2025 Meeting)

Tabled until the 4-21-25 City Council Meeting.

# 7. Mayor Norby

Mayor Norby stated April 17th to June 17th are the open dates for filing on the November election ballot. Alderwoman Larson will file for the Ward 1 council seat she has been voted into as interim council woman. Alderman Koffler stated he does not intend to file for re-election for his Ward 1 seat.

Mayor Norby stated there is an open position on the Richland Economic Development board as Alderwoman Buxbaum has stepped down from the board.

# 8. Committee Meeting Work

a. Water and Sewer Committee: North Park Water Tower Replacement-Authorization to Advertise for Bids for Phase II

Motion was made to advertise for bids for Phase II of the North Park Water Tower Replacement.

Motion made by Alderman DiFonzo, Seconded by Alderwoman Buxbaum. Voting Yea: Alderman Koffler, Alderwoman Rasmussen, Alderman DiFonzo, Alderwoman Buxbaum, Alderwoman Christensen, Alderwoman Larson

b. Water and Sewer Committee: Meadow Village Storm Water Project and Amendment 2 Morrison Majerle to contract

Motion was made to approve the Meadow Village Storm Water Project and Amendment 2 Morrison Majerle contract.

Motion made by Alderman Koffler, Seconded by Alderwoman Buxbaum. Voting Yea: Alderman Koffler, Alderwoman Rasmussen, Alderman DiFonzo, Alderwoman Buxbaum, Alderwoman Christensen, Alderwoman Larson

# 9. Alderman Requests and Committee Reports

<u>Budget and Finance</u> – Chairman Christensen – DiFonzo, Koffler. <u>Police and Fire</u> – Chairman DiFonzo – Rasmussen, Larson. <u>Parks and Recreation</u> – Chairman Christensen – Larson, Rasmussen. <u>Water</u> and Sewer – Chairman Godfrey – Christensen, DiFonzo. <u>Street and Alley</u> – Chairman Rasmussen –



DiFonzo, Koffler. <u>Sanitation</u>— Chairman Larson — Koffler, Godfrey. <u>City Buildings & Street</u> <u>Lighting — Chairman Koffler — Godfrey, Christensen</u>

Anderson Drainage Improvements Proposal from the Water and Sewer Committee Meeting (March 24,2025) was not added to the nights agenda.

Alderwoman Buxbaum made a motion to add Anderson Drainage Improvements Proposal to the council meeting. Alderman DiFonzo seconded. Voting Yea: Alderman DiFonzo, Alderwoman Buxbaum, Alderwoman Christensen, Alderwoman Larson, Anderman Koffler, Alderwoman Rasmussen.

Motion was made to approve the Anderson Drainage Improvements Proposal.

Motion made by Alderwoman Buxbaum, Seconded by Alderman Koffler.

Voting Yea: Alderman Koffler, Alderwoman Rasmussen, Alderman DiFonzo, Alderwoman

Buxbaum, Alderwoman Larson

Voting Abstaining: Alderwoman Christensen

## 10. Unfinished Business

a. Water Tower Replacement Logo-Poll on Facebook Results

Alderwoman Buxbaum presented the council the final Facebook poll vote for the water tower design. The Facebook poll had 387 total votes, with Option 2 as the winner with 194 votes (51%). The design was shown to the council in navy blue and maroon, with and without a white pinstripe, for final design vote.

Alderman DiFonzo made a motion to approve the final Water Tower Replacement Logo Option 2 in the color maroon with a white pinstripe.

Motion made by Alderman DiFonzo, Seconded by Alderman Koffler.

Voting Yea: Alderman Koffler, Alderwoman Rasmussen, Alderman DiFonzo, Alderwoman Buxbaum, Alderwoman Christensen, Alderwoman Larson

# 11. New Business

a. Approval of Downtown Master Planning Services RFP (Interstate Engineering)

Alderwoman Buxbaum stated that one proposal for the master planning was submitted from Rachael Laqua of Interstate Engineering.

Motion was made to approve Downtown Master Planning Services RFP from Interstate Engineering.

Motion made by Alderwoman Buxbaum, Seconded by Alderman Koffler. Voting Yea: Alderman Koffler, Alderwoman Rasmussen, Alderman DiFonzo, Alderwoman Buxbaum, Alderwoman Christensen, Alderwoman Larson

# 12. City Planner

a. Amended Plat-Homestead Subdivision Phase 2 (tabled at 3-17-25 meeting) (Tabled until the 4-21-2025 Meeting)

Request to continue to table until the April 21st council meeting.

# 13. City Attorney

a. First Reading of Ordinance 601-Amending Snow Removal Code (tabled at 3-3-25 meeting)

Request was made to un-table the First Reading of Ordinance 601- Amending Snow Removal Code.

Alderman DiFonzo asked if this ordinance is modeled the same as the ordinance in Williston. City Attorney Kalil stated that it is similar to Williston's ordinance. Alderman DiFonzo asked if the City of Willistion has had any issues with the ordinance and the time limits. City Attorney Kalil stated there have been no issues that he is aware of.

Motion to un-table made by Alderman Koffler, seconded by Alderwoman Larson. Voting Yea: Alderwoman Rasmussen, Alderwoman Buxbaum, Alderman DiFonzo, Alderwoman Christensen.

Motion was made to approve the First Reading of Ordinance 601- Amending Snow Removal Code.

Motion made by Alderman DiFonzo, Seconded by Alderwoman Buxbaum.

Voting Yea: Alderman Koffler, Alderwoman Rasmussen, Alderwoman Christensen, Alderwoman Larson

b. First Reading of Ordinance 603-Amending Zoning Code (tabled at 3-17-25 meeting)

Request was made to un-table the First Reading of Ordinance 603- Amending Zoning Code.

Motion made by Alderman Koffler, Seconded by Alderwoman Christensen.

Alderwoman Christensen asked if the retaining wall height of 8 feet was correct for all zones. City Attorney Kalil stated that 8 feet is the stated height for all zones even though fence height varies by zone from 6 feet to 8 feet.

Motion was made to approve the First Reading of Ordinance 603- Amending Zoning Code.

Motion made by Alderwoman Larson, Seconded by Alderwoman Christensen. Voting Yea: Alderman Koffler, Alderwoman Rasmussen, Alderman DiFonzo, Alderwoman Buxbaum

# 14. Chief of Police

a. Eastern Montan Drug Task Force Understanding for FY 2025-2026

Request to approve Eastern Montana Drug Task Force Understanding for FY 2025-2026.

Motion made by Alderman DiFonzo, Seconded by Alderwoman Buxbaum. Voting Yea: Alderman Koffler, Alderwoman Rasmussen, Alderwoman Christensen, Alderwoman Larson

b. Police Department Monthly Report for March 2025

Chief Kraft presented the March 2025 Police Department Report.

Chief Kraft was happy to announce a new officer position has been accepted.

Chief Kraft will travel to Helena Friday for a senate confirmation hearing and appointment. Chief Kraft has been been in the governor appointed council position for the past 2 years.

# 15. Public Works Director

a. Public Works Director March 2025 Monthly Report

PWD Hintz presented the March 2025 Public Works Report.

PWD Hintz stated that work is progressing on the Water Tower Replacement Project. Temporary water lines are set up and Bac-T testing this week, digging will start on April 21st and take a month to complete.

b. Contract for B&B Builders for Facilities and Grounds Maintenance (tabled at 3-17-25 meeting)

PWD Hintz requests to continue to table the Contract for B&B Builders for Facilities and Grounds Maintenance.

- 16. Fire Marshal/Building Inspector
- 17. City Clerk/Treasurer

# 18. Consent Agenda

Motion was made to approve the claims and building permits.

Motion made by Alderman Koffler, Seconded by Alderwoman Larson. Voting Yea: Alderman Koffler, Alderwoman Rasmussen, Alderman DiFonzo, Alderwoman Buxbaum, Alderwoman Christensen, Alderwoman Larson

a. Intermediary Claims to be approved (included in total): \$4,102.06 + \$99,989.46 = \$104,091.052

#43713	SWENNY'S BODY WORKS	\$ 2,030.00
#43714	CASH \$ 1,000.00	
#43715	MDU \$17.62	
#43745	USPS \$1,054.44	

b. Building Permits to be approved: 2025-050, 2025-051, 2025-054, 2025-055, 2025-056 and RC2025-15 & RC2025-17 & RC2025-18 & RC2025-19, RC2025-21

# 19. Adjournment

Adjourned at 6:55pm.



115 2nd Street S.E., Sidney, Montana - 406-433-2809

# STAFF REPORT #MNR-25-01-Revised Homestead Subdivision Phase 2 - Minor Subdivision March 26, 2025

Owner Agent

MPEG Land Development LLC Interstate Engineering Donald Sterhan Brent Moore, AICP 606 S Grant Ave 2201 Overland Avenue Billings MT 59102 Red Lodge, MT 59068

B-1 Community Business District. **Existing Zoning:** 

**Legal Description:** Amended Plat of Lot 2, Block 2 of Phase 1 The Homestead Subdivision and portions of the SE¼, located in the SE¼ Section 29, Township 23 North, Range 59 East P.M.M., City of Sidney, Richland County, Montana. To be known as Phase 2 of The Homestead Subdivision.

# REGULATORY REQUIREMENTS

# Sidney Subdivision Regulations Chapter 5A

# 1. Information to be Considered

The governing body's decision to approve, conditionally approve, or deny a proposed preliminary plat application shall be based on review of the application, preliminary plat, applicable environmental assessment, public hearing and comment, planning board recommendation, staff report, and other information submitted or prepared in the review of the application. As per 76-3-608(10), MCA, findings of fact by the governing body concerning whether the development of the proposed subdivision meets the requirements of these regulations and the MSPA must be based on the record as a whole.

The necessary and required information for the City Council to approve, conditionally approve, or deny the preliminary plat application has been submitted for public review.

- ✓ The application was submitted for public review on February 19, 2025.
- ✓ Element Review identified deficiency which was corrected on February 26, 2025
- ✓ Sufficiency Review was completed on February 27, 2025.
- ✓ The review timeline for Minor Subdivisions is 35 working days. The review timeline for this project expires on April 17, 2025.
- ✓ The project has been scheduled for consideration and public comments at the March 17, 2025 City Council meeting.
- ✓ The developer requested time to consider the proposed mitigations related to water and sewer main extensions. The project was Tabled to the April 7, 2025 meeting and could be Tabled further if the developer does not respond to the Staff recommended modifications to the conditions.

As an Expedited Subsequent Minor Subdivision of a Tract of Record, the city may require parkland dedication if the intended use is residential, the preparation of a Summary of Probable Impacts, and hold a formal public hearing, with comment limited to compliance with Zoning, and the availability of public utilities. The application includes a summary of probable impacts addressing the review criteria found in §76-3-608 MCA.

# Proposed Mitigations:

## None

# 2. Review Criteria

The basis for the governing body's decision to approve, conditionally approve, or deny the proposed subdivision shall be whether the subdivision meets the requirements of the MSPA and these regulations. The findings of fact shall include an evaluation of the following criteria.

# a) Compliance with These Regulations

The review is based on the subdivision regulations in place at the time the application is determined to contain sufficient information for review. Subdivisions that do not comply with the following standards shall be denied, unless the requirement is made a condition of approval, or a variance is granted.

- Design Standards Compliance with provisions of the design standards in Chapter VII is required.
- ii. Special Provisions for Condominiums, Townhouses, Townhomes, and RV and Mobile Home Parks - Condominiums, townhouses, townhomes, and RV and mobile home parks shall meet the additional design standards and requirements in Chapter III and as referenced in other chapters for these types of subdivisions.
- iii. Compliance with Zoning All subdivisions must demonstrate they are designed to comply with applicable zoning.
- iv. Compliance with Other Regulations All subdivisions shall demonstrate compliance with other applicable regulations including ARMs for sanitation and water supply, floodplain regulations, etc.
- v. Conformance with Adopted Plans Subdivisions must substantially comply with adopted plans such as a growth policy, transportation plan, or community wildfire protection plan. Because plans are not regulatory, no variance for nonconformance is required, nor can denial or a condition of approval be based solely on plan conformance.
- b) Compliance with the Subdivision Review Procedure (76-3-608(3)(b)(iii), MCA) All preliminary plat applications shall be reviewed in compliance with the procedures established in these regulations and the MSPA. The record shall demonstrate the review processes and timeframes in Chapter IV were followed. All conditional approvals shall include a timeframe for final plat filing.
- c) Compliance with Survey Requirements (<u>76-3-608(3)(b)(i)</u>, <u>MCA</u>) The preliminary plat shall comply with the survey requirements in Part 4 of the MSPA.
- d) Provision of Legal and Physical Access (76-3-608(3)(d), MCA) All preliminary plat applications shall demonstrate how legal and physical access is to be provided to each parcel within the proposed subdivision and the required notation of that access on the applicable plat and any instrument of transfer concerning the parcel.
- e) Provision of Utility Easements (76-3-608(c), MCA)

The preliminary plat application shall provide easements within and to the proposed subdivision for the location and installation of any planned utilities.

- f) Impacts to Agriculture, Agriculture Water User Facilities, Local Services, the Natural Environment, Wildlife, Wildlife Habitat, and Public Health and Safety (76-3-608(3)(a). MCA) The governing body shall identify any impacts it determines to be potentially significant and adverse to agriculture, agricultural water user facilities, local services, the natural environment, wildlife, wildlife habitat, and public health and safety. The criteria for determining significant adverse impacts are the criteria adopted in the City of Sidney Growth Policy as required by 76-1-601(3)(h), MCA.
- ✓ The proposed subdivision is consistent with the Sidney Zoning Regulations.
- ✓ The proposed subdivision is consistent with the B-2 Zoning assigned to the property.
- The proposed uses of each lot in the subdivision is consistent with the B-2 Zoning assigned to the property.
  - Proposed Mitigation: If the uses of the property are residential, a parkland dedication or cash-in-lieu payment equal to 0.03 acres per dwelling unit. Commercial uses are exempt from the dedication of parkland.
- No additional public infrastructure is proposed to develop the proposed subdivision.
- ✓ The proposed subdivision does not require any variance to the adopted subdivision regulations.
- ✓ The proposed subdivision is consistent with the Sidney Growth Policy.
- ✓ The proposed subdivision is served by City water and sewer. With main extensions being advised to the northern most portion of the property. The developer has objected to the requirement to extend the water mains on 14<sup>th</sup> Ave. This will limit the location of structures on Lot 2A to comply with the requirements that buildings be located within the prescribed distance to an existing fire hydrant.
  - Proposed Mitigation: The provision of water and sewer for the lots in the subdivision including required main extensions shall be reviewed and approved by the Montana Department of Environmental Quality.
- ✓ The Sidney Public Works Department has submitted a 'Will Serve Letter' for the provision
  of water, sewer, and solid waste services. If each lot in the subdivision is connected to City
  Servies an MFE could be approved.
- ✓ Lot 2A has access to the water main in 4<sup>th</sup> Street NW. However a water main does not exist on 14<sup>th</sup> Ave and potential development on Lot 2A may be limited due to separation distance limitations from fire hydrants. requirements
  - Proposed Mitigation: The developer shall comply with all City of Sidney Department of Public Works requirements with provision of water, sewer, or solid waste services to the lots in the subdivision.
- ✓ The subdivision is subject to review as the Expedited Subsequent Minor Subdivision of a tract of record within the City of Sidney. All of the rights and exceptions associated with a Expedited Subdivision have been applied to the project.
- ✓ The preliminary plat appears to comply with the survey requirements of State Law and the Sidney Subdivision Regulations.
- ✓ Each lot has both legal and physical access to the public road infrastructure.
- ✓ All taxes due and levied must be paid in full prior to filing the Final Plat.
- ✓ Any liens and encumbrances against the property must be paid or a consent to platting certificate will need to accompany the Final Plat application.
  - Proposed Mitigation: An updated Title Commitment, not less than 30-days old, at the time of Final Plat application shall be submitted along with the Final Plat. Consent to Plats for each lien holder or owner, who is not a signatory on the Plat, must be submitted with the Final Plat.
- ✓ The preliminary plat shows the necessary public and private utility easements.
- ✓ The subdivision application includes a summary of probable impacts to the subdivision review criteria found in 76-3-608. The summary report does not include any adverse or unacceptable impacts, and is by reference incorporated into this report.

- ✓ The subdivision is exempt from the preparation of an Environmental Assessment as an Expedited Subdivision.
  - Proposed Mitigations:
    - The sewer main on 4th Street & 14th Avenue shall be required to be extended north to the furthest extents of the Homestead Subdivision alongside 14th Ave. NW.
    - Water/fire service will be extended to lot 2A and may restrict future building permits within this area depending on the type of building(s) installed. See City Code 3-1-20 paragraph C.
       Parkland dedication will need to be addressed at the time of building permit

Parkland dedication will need to be addressed at the time of building permit applications if Lot 2A is developed with residential structures. 76-3-621 spells out the parkland requirements.

# 3. Limitations

- a) Prohibition on Requiring Capital Facilities for Education (<u>76-3-510, MCA</u>) Although the governing body may require the subdivider to pay or guarantee payments for part or all of the costs of extending capital facilities related to public health and safety, the costs of constructing or extending capital facilities related to education may not be imposed on the subdivider.
- b) Restrictions on Conditional Approval or Denial for Water and Sanitation (76-3-608(6), MCA) The governing body may conditionally approve or deny a proposed subdivision as a result of the water and sanitation information provided in the preliminary plat application or on public comment related to the provided sanitation information only if the conditional approval or denial is based on existing subdivision, zoning, or other regulations that the governing body has the authority to enforce.
- c) Limitation on Waivers of the Right to Protest Improvement Districts (76-3-608(7), MCA)

  The governing body may not require as a condition of subdivision approval that a property owner waive a right to protest the creation of a special improvement district or a rural improvement district for capital improvement projects that does not identify the specific capital improvements for which protest is being waived. A waiver of a right to protest may not be valid for a time period longer than 20 years after the date the final subdivision plat is filed with the county clerk and recorder.
- d) Restriction on Encroachment onto Adjoining Private Property (76-3-608(8), MCA)

  The governing body may not approve a proposed subdivision if any of the features and improvements of the subdivision encroach onto adjoining private property in a manner that is not otherwise provided for under the Montana Sanitation in Subdivisions Act or these regulations or if the well isolation zone of any proposed well to be drilled for the proposed subdivision encroaches onto adjoining private property unless the owner of the private property authorizes the encroachment. For the purposes of this section, "well isolation zone" has the meaning provided in 76-4-102, MCA.
- e) No Denial Solely for Education or Wildland Urban Interface (<u>76-3-608(1)</u>, <u>MCA</u>) The governing body may not deny a proposed subdivision solely for impacts to educational services or designation or based solely on parcels within the subdivision having been designated as urban-wildland interface parcels under 76-13-145, MCA.
- f) No Delay or Denial Based on Failure of Public Utility or Agency to Comment (76-3-504(1)(i), MCA)
  The governing body may not delay a decision on the preliminary plat application past the review

period time limits provided in these regulations and may not deny a preliminary plat application due to failure of any public utility or agency to provide comments on the application.

None of the proposed mitigations or proposed conditions of approval violate the limitations on the review and approval, conditional or otherwise imposed by State Law or the Sidney Subdivision Regulations.

# CONCLUSION:

As discussed in this Report and the Findings of Fact, the Division is consistent with the City of Sidney Subdivision and Zoning Regulations. The necessary public infrastructure exists and is available to the resulting lot. As such, the project, subject to the imposition of conditions is approvable by the Sidney City Council.

## RECOMMENDATION:

The Subdivision Administrator recommends that the City Council Adopt this report (#MNR-25-01) as their Findings of Fact and Conclusions of Law. Further, the Subdivision Administrator recommends that the Amended Plat of Lot 2, Block 2 of Phase 1 The Homestead Subdivision and portions of the SE¼, located in the SE¼ Section 29, Township 23 North, Range 59 East P.M.M., City of Sidney, Richland County, Montana. To be known as Phase 2 of The Homestead Subdivision, be APPROVED subject to the following conditions:

- That the Final Plat substantially comply with the documents, commitments, design and layout of the preliminary plat.
- That a Subdivision Guarantee for the properties involved in the aggregation submitted for examination by the Subdivision Administrator. Said Subdivision Guarantee shall be less than 30 days old at the time of submittal.
- 3. That any lien holders' and owners sign the Final Amended Plat or provide a Consent to Plat to the filing of the Amended Plat.
- 4. That all taxes or assessments that are due and levied on the lots to be aggregated be paid prior to filing the Amended Plat.
- 5. If the uses of the property are residential, a parkland dedication or cash-in-lieu payment equal to 0.03 acres per dwelling unit. Commercial uses are exempt from the dedication of parkland.
- 6. The provision of water and sewer for the lots in the subdivision, including any required main extensions shall be reviewed and approved by the Montana Department of Environmental Quality.
- 7. Water/fire services being extended to serve all of lot 2A may result in restrictions on future building permits within this area depending on the type of building(s) installed. See City Code 3-1-20 paragraph C. This requirement/limitation shall be stated on the face of the Final Plat or in a supporting document recorded with the Final Plat.
- 8. Parkland dedication will need to be addressed at the time of building permit applications if Lot 2A is developed with residential structures. 76-3-621 spells out the parkland requirements. This requirement/limitation shall be stated on the face of the Final Plat or in a supporting document recorded with the Final Plat.
- 9. The developer shall comply with all City of Sidney Department of Public Works requirements with provision of water, sewer, or solid waste services to the lots in the subdivision.
  - a. The sewer main on 4th Street & 14<sup>th</sup> Avenue shall be required to be extended north to the furthest extents of the Homestead Subdivision alongside 14<sup>th</sup> Ave. NW.
- 10. That the Final Amended Plat is approved for filing within three (3) calendar years from the date of this approval.

Respectfully submitted,

Forrest Sanderson AICP and CFM

Contracted Planner City of Sidney



# City of Sidney, MT Zoning/Board of Adjustment Meeting April 16, 2025 8:00 AM 115 2nd Street SE | Sidney, MT 59270

The City Council meetings are open to the public attending in person, with masks encouraged when social distancing cannot be accomplished. If the public does not wish to participate in person, they are also invited to participate via a Zoom meeting. You can participate via phone:

Meeting ID: 713 080 5898 Passcode: 4332809 Call: 1-346-248-7799

Board of Adjustments: Chaiman Seitz, Meldahl, and PWD Hintz

Other's Present: FM/BI Rasmussen, Deputy Clerk/Treasurer Shanks, Jannelle Jensen, Corey Wieland, and Mayra Wieland

# 1. New Business:

# a. Corey Wieland Variance Request

Chairman Seitz opened the Zoning Board Meeting to Corey Weiland to discuss his height variance request for his shop/practice barn, BI/FM Rasmussen mentioned that the side walls will be 18feet for pole vaulting, and that he doesn't see any issues with his variance request. Corey said that his neighbor accessory building is 10 feet taller due to the crest of the hill, and most houses in his neighborhood have an accessory building on their property, and he would like to be in line with the rest of the neighborhood.

Chairman Seitz called the public hearing for the Corey Wieland Variance Request to order 8:05am. The property is located at Lot 007, Block 001 of the Anderson Subdivision.

Chairman Seitz called for any proponents, three times with the following coming forward:

Janelle Jensen came forward and mentioned that she is board member of the Hilltop Shop and also neighbor's with Corey Wieland and saw no issues or concerns with their request and would be happy to see the Weilands move forward with their project. Deputy Clerk Treasurer Shanks did state that there has been zero correspondence prior to this meeting.

Chairman Seitz called for any opponents with none coming forward.

Chairman Seitz closed the public hearing at 8:07am.

Mr. Meldahl also wanted to double check with BI/FM Rasmussen that if Corey ever decides to sell his property that there are no covenants that would tie in with pole vault, that this building could

be used for storage or other things as well, and BI/FM agreed with Mr. Meldahl that this building is not defined just for pole vault.

Mr. Meldahl stated that he saw no oppositions with this request and would like to make a formal motion to move forward with Corey Wieland Variance Request as it is listed.

Motion made by Meldahl, Seconded by Hintz.

Voting Yea: Seitz

Adjourned at 8:10am.

# DRAFT 03/27/2025

# CERTIFICATE AS TO RESOLUTION AND ADOPTING VOTE

I, the undersigned, being the duly qualified and acting recording officer of the City of Sidney, Montana (the "City"), hereby certify that the attached resolution is a true copy of Resolution No. 3946 entitled: "RESOLUTION RELATING TO FINANCING OF CERTAIN PROPOSED WATER SYSTEM IMPROVEMENTS; ESTABLISHING COMPLIANCE WITH REIMBURSEMENT BOND REGULATIONS UNDER THE INTERNAL REVENUE CODE" (the "Resolution"), on file in the original records of the City in my legal custody; that the Resolution was duly adopted by the City Council of the City at a meeting on April 7, 2025 and that the meeting was duly held by the City Council and was attended throughout by a quorum, pursuant to call and notice of such meeting given as required by law; and that the Resolution has not as of the date hereof been amended or repealed.

	; voted
against the same:	; abstained from
voting thereon:	; or were
absent:	
WITNESS my hand officially thisday of April, 2025.	
City Clerk-Tr	reasurer

# RESOLUTION NO. 3946

RESOLUTION RELATING TO FINANCING OF CERTAIN PROPOSED WATER SYSTEM IMPROVEMENTS; ESTABLISHING COMPLIANCE WITH REIMBURSEMENT BOND REGULATIONS UNDER THE INTERNAL REVENUE CODE

BE IT RESOLVED by the City Council of the City of Sidney, Montana (the "City"), as follows:

# 1. Recitals.

- (a) The United States Department of Treasury has promulgated final regulations governing the use of proceeds of tax-exempt bonds, all or a portion of which are to be used to reimburse the City for project expenditures paid by the City prior to the date of issuance of such bonds. Those regulations (Treasury Regulations, Section 1.150-2) (the "Regulations") require that the City adopt a statement of official intent to reimburse an original expenditure not later than 60 days after payment of the original expenditure. The Regulations also generally require that the bonds be issued and the reimbursement allocation be made from the proceeds of the bonds within 18 months (or three years, if the reimbursement bond issue qualifies for the "small issuer" exception from the arbitrage rebate requirement) after the later of (i) the date the expenditure is paid or (ii) the date the project is placed in service or abandoned, but (unless the issue qualifies for the "small issuer" exception from the arbitrage rebate requirement) in no event more than three years after the date the expenditure is paid. The Regulations generally permit reimbursement of capital expenditures and costs of issuance of the bonds.
- (b) The City desires to comply with requirements of the Regulations with respect to certain projects hereinafter identified.

# 2. Official Intent Declaration.

- (a) The City proposes to undertake the acquisition, construction and installation of various improvements to the City's water system, including water lires, a new water tower, restoration of the water tower site, and related improvements and costs (the "Project").
- (b) Other than (i) expenditures to be paid or reimbursed from sources other than the Bonds (as hereinafter defined), (ii) expenditures permitted to be reimbursed under the transitional provision contained in Section 1.150-2(j)(2) of the Regulations, (iii) expenditures constituting preliminary expenditures within the meaning of Section 1.150-2(f)(2) of the Regulations, or (iv) expenditures in a "de minimus" amount (as defined in Section 1.150-2(f)(1) of the Regulations), no expenditures for the Project have heretofore been paid by the City and no expenditures will be paid by the City until after the date of this Resolution.
- (b) The City reasonably expects to reimburse the expenditures made for costs of the Project out of the proceeds of debt evidenced by one or more series of bonds or notes in an estimated aggregate principal amount of approximately \$7,000,000 (collectively, the "Bonds")

after the date of payment of all or a portion of the costs of the Project. All reimbursed expenditures shall be capital expenditures, a cost of issuance of the Bonds or other expenditures eligible for reimbursement under Section 1.150-2(d)(3) of the Regulations.

- 3. <u>Budgetary Matters</u>. As of the date hereof, there are no City funds reserved, allocated on a long-term basis or otherwise set aside (or reasonably expected to be reserved, allocated on a long-term basis or otherwise set aside) to provide permanent financing for the expenditures related to the Project, other than pursuant to the issuance of the Bonds. The statement of intent contained in this resolution, therefore, is determined to be consistent with the City's budgetary and financial circumstances as they exist or are reasonably foreseeable on the date hereof.
- 4. <u>Reimbursement Allocations</u>. The City Clerk-Treasurer shall be responsible for making the "reimbursement allocations" described in the Regulations, being generally the transfer of the appropriate amount of proceeds of the Bonds to reimburse the source of temporary financing used by the City to make prior payment of the costs of the Project. Each allocation shall be evidenced by an entry on the official books and records of the City maintained for the Bonds or the Project and shall specifically identify the actual origina! expenditure being reimbursed.

Adopted this 7th day of April, 2025.

	Mayor	
attest:		

# **MASTER AGREEMENT**

by and between

CITY OF SIDNEY

and

B & B Builders Inc.

Dated as of \_\_\_\_\_\_\_, 2025

# **TABLE OF ARTICLES**

- 1 MASTER AGREEMENT TERM AND PARTY REPRESENTATIVES
- 2 WORK ORDERS
- 3 PAYMENTS
- 4 DISPUTE RESOLUTION
- 5 GENERAL PROVISIONS
- 6 OWNER
- 7 CONTRACTOR
- **8** ARCHITECT
- 9 **SUBCONTRACTORS**
- 10 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS
- 11 CHANGES IN THE WORK
- 12 TIME
- 13 PAYMENTS AND COMPLETION
- 14 PROTECTION OF PERSONS AND PROPERTY
- 15 INSURANCE AND BONDS
- 16 CORRECTION OF WORK
- 17 MISCELLANEOUS PROVISIONS
- **18 TERMINATION**
- 19 CLAIMS AND DISPUTES
- 20 SCOPE OF THIS MASTER AGREEMENT
- 21 OTHER PROVISIONS

## MASTER AGREEMENT

Notice: This Agreement contains provisions relating to indemnification, release of liability and allocation of risk. By executing this Agreement, Contractor agrees to be bound by all of those provisions.

Effective 1	Date:, 2025	
BETWEE	N:	
	<b>City of Sidney</b> , a municipal corporation of the State of Montana, whose address is 115 2 <sup>nd</sup> Street SE, Sidney, Montana, 59270,	;
	- and -	("Owner")
	<b>B &amp; B Builders Inc.</b> , whose address is 108 2 <sup>nd</sup> St NE, Sidney, Montana 59270	19
		("Contractor")

Owner and Contractor may hereinafter sometimes be referred to individually as a "Party" and collectively as the "Parties".

WHEREAS the Parties wish to provide the general terms and conditions under which Contractor will provide Services to Owner from time to time;

NOW THEREFORE the Parties agree as follows:

# ARTICLE 1 MASTER AGREEMENT TERM AND PARTY REPRESENTATIVES

§ 1.1 Owner issued a Request for Qualifications to contract with multiple construction firms under continuing services contracts as defined therein. Contractor agrees to provide professional services to Owner for Projects as determined by and selected by Owner, upon terms and conditions set forth herein and upon mutual agreement and execution of Task Orders for each Project. Each Task Order for each Project, upon execution, shall automatically be incorporated fully herein in this Master Agreement. All the terms and conditions of this Master Agreement shall fully apply to each Task Order. Each Project budget for each Task Order will vary and will be stated in each Task Order. Contractor recognizes that this Master Agreement is not exclusive, and that Owner has and/or may contract with more than one construction firm under the terms of similar continuing contracts and will equitably rotate and divide the Projects based upon Owner's current volume of work assigned, the dollar value of work assigned, the particular specialties of the contractors and their subcontractors, and the ability of the contractors to perform the services within Owner's needed schedule and other requirements.

The Scope of Work and Services are generally described in Exhibit B.

If a Task Order includes as an attachment a proposal from Contractor, only the scope of services portions of such proposal shall be part of the Task Order and any terms and conditions included in such proposal shall not be part of the Task Order or this Master Agreement and such proposal terms and conditions shall be considered null and void.

Owner retains the right to reduce the scope of any portion of the Scope of Services of any Task Order. In such event, Owner shall be entitled to proportionally reduce the Contractor's compensation.

Contractor has represented, upon which Owner has relied, that Contractor is properly qualified, licensed, and competent to perform such construction services.

Contractor recognizes that this Master Agreement is not exclusive, and that Owner has and/or may contract with more than one construction firm under the terms of similar continuing contracts, and Contractor does not object.

- § 1.2 Term. This Master Agreement shall be effective for five (5) years after the date first written above ("Date of this Master Agreement") and may be extended upon mutual written agreement of both parties for two (2) additional one (1) year terms ("Term"). This Master Agreement shall terminate at the expiration of the five (5) year initial Term and at the end of each one (1) year extension Term if not extended. In the event the Term expires and is not renewed, or is terminated as provided herein, if Contractor has not completed the services under a Work Order, then the Term will remain in effect through the date of completion of the services in the Work Order, but no new requests for proposals and no new Work Orders shall be issued.
- § 1.3 This Master Agreement shall apply to all Work Orders agreed to by the parties within the term of this Master Agreement until completion of the Work Order. In the event of a conflict between terms and conditions of this Master Agreement and a Work Order, the terms of the Work Order shall take precedence for the Work provided pursuant to the Work Order.
- § 1.4 The Owner identifies the following representative authorized to act on the Owner's behalf with respect to this Master Agreement:

Jeff Hintz (406) 433-2809

- § 1.4.1 In each Work Order, the Owner will identify a representative authorized to act on the Owner's behalf with respect to the Work Order.
- § 1.5 The Contractor identifies the following representative authorized to act on the Contractor's behalf with respect to this Master Agreement:

Sean Suisse, President (406) 482-4401

§ 1.5.1 In each Work Order, the Contractor will identify a representative authorized to act on behalf of the Contractor with respect to the Work Order.

# ARTICLE 2 WORK ORDERS

- § 2.1 The Owner is not required to issue any Work Orders under this Master Agreement.
- § 2.2 The Contractor may decline to accept any Work Order issued by the Owner.
- § 2.3 The Contractor shall execute the Work set forth in each agreed upon Work Order, attached as **Exhibit C**, or such other document as the Owner and Contractor may mutually agree upon. Each Work Order shall state the name, location, and detailed description of the Project; identify the Architect; state the Contract Time; state the Contract Sum; describe the Work; and enumerate the Contract Documents.
- § 2.4 The Owner shall make the Contract Documents available to the Contractor prior to execution of the Work Order, and thereafter, upon request. The Owner may charge the Contractor for the reasonable cost to reproduce the Contract Documents provided to the Contractor.

# ARTICLE 3 PAYMENTS

# § 3.1 Contract Sum and Progress Payments

- § 3.1.1 Each Work Order shall include a Contract Sum. The Owner shall pay the Contractor the Contract Sum in current funds in accordance with each individual Contract. Where the Contract Sum is based on the Cost of the Work under Section 3.3 or 3.4 of the Work Order, the Cost of the Work is defined in **Exhibit A**, Determination of the Cost of the Work.
- § 3.1.2 Applications for Payment will be submitted individually for each Contract. Travel and per diem charges shall not exceed the limits as set forth in Montana Code Annotated.
- § 3.1.3 Based upon Applications for Payment for individual Contracts submitted to the Architect by the Contractor, and Certificates for Payment issued by the Architect, the Owner shall make progress payments on account of the Contract Sum to the Contractor as provided below and elsewhere in the Contract Documents.
- § 3.1.4 The period covered by each Application for Payment shall be one calendar month ending on the last day of the month, or as set forth in the Work Order.
- § 3.1.5 Provided that an Application for Payment is received by the Architect not later than the 25th day of the month, the Owner shall make payment of the certified amount to the Contractor not later than the 20th day of the next month. If an Application for Payment is received by the Architect after the date fixed above, payment shall be made by the Owner not later than twenty-

five (25) days after the Architect receives the Application for Payment.

§ 3.1.6 For each progress payment made prior to Substantial Completion of the Work, the Owner may withhold retainage from the payment otherwise due as follows:

5% retainage shall be withheld. Upon Substantial Completion of the entire Work of each Work Order, all retainage shall be released except 150% of the value of punch list Work, incomplete Work, and defective Work shall be withheld.

§ 3.1.7 Payments due and unpaid under the Contract shall bear interest from the date payment is due at the rate stated below, or in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located.

# § 3.2 Final Payment

- § 3.2.1 Contractor shall submit its final pay request within 45 days of Final Completion of each Work Order. Final payment for individual Contracts, constituting the entire unpaid balance of the Contract Sum, less 150% of the value of punch list Work, shall be made by the Owner to the Contractor when:
  - 1. the Contractor has fully performed the Work except for the Contractor's responsibility to correct Work as provided in Section 16.2, and to satisfy other requirements, if any, which extend beyond final payment;
  - 2. the Contractor has submitted a final accounting for the Cost of the Work, where payment is on the basis of the Cost of the Work with or without a Guaranteed Maximum Price:
  - **3.** the Contractor has delivered to the Owner a Contractor's Final Affidavit pursuant to Montana Law and a final Certificate for Payment has been issued by the Architect in accordance with Section 13.7.1.
  - **4.** Contractor has delivered to Owner a certification that the Work does not contain any asbestos; and
  - **5.** Contractor has delivered to Owner a certification from the Building Department that the building permit has been finalized and closed out.
- § 3.2.2 The Owner's final payment to the Contractor shall be made no later than 30 days after the issuance of the Architect's final Certificate for Payment, but in no event until all the conditions of 3.2.1 have been met.

# ARTICLE 4 DISPUTE RESOLUTION

# § 4.1 Binding Dispute Resolution

For any claim subject to, but not resolved by, mediation pursuant to Section 19.5, the method of binding dispute resolution shall be as follows:

	Arbitration pursuant to Section 19.6 of this Master Agreement
[X]	Litigation in a court of competent jurisdiction
[ ]	Other: (Specify)

If the Owner and Contractor do not select a method of binding dispute resolution, or do not subsequently agree in writing to a binding dispute resolution method other than litigation, claims will be resolved in a court of competent jurisdiction.

## ARTICLE 5 GENERAL PROVISIONS

# § 5.1 The Work

The term "Work" means the construction and services required by the Contract Documents enumerated in a Work Order, whether completed or partially completed, and includes all other labor, materials, equipment and services provided or to be provided by the Contractor to fulfill the Contractor's obligations under the Work Order and related Contract. The Work may constitute the whole or a part of the Project identified in a particular Work Order.

# § 5.2 The Contract Documents

The Contract Documents are enumerated in each Work Order and consist of this Master Agreement; the Work Order executed by the Owner and Contractor (including, if applicable, Supplementary and other Conditions applicable to the Work Order); all Drawings, Specifications, and Addenda issued in connection with the Work Order; other documents listed in the Work Order; and Modifications issued after execution of the Work Order. A Modification is (I) a written amendment to the Contract signed by both parties, (2) a Change Order, (3) a Construction Change Directive or (4) a written order for a minor change in the Work issued by the Architect. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all; performance by the Contractor shall be required to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results.

# § 5.3 The Contract

The Contract Documents for each Work Order form a separate Contract for construction of the Work ("The Contract"). The Contract represents the entire and integrated agreement between the parties hereto for construction of the Work and supersedes prior negotiations, representations or agreements, either written or oral. A Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind between any persons or entities other than the Owner and the Contractor.

# § 5.4 Instruments of Service

Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Architect and the Architect's consultants under their respective professional services agreements. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, digital models, and other similar materials.

# § 5.5 Ownership and Use of Drawings, Specifications and Other Instruments of Service

§ 5.5.1 Drawings, specifications, Instruments of Service and other documents including those in electronic form, prepared by the Architect or Contractor and furnished for the Project are the

property of Owner. The Owner shall retain all common law, statutory and other reserved rights in their Instruments of Service, including copyrights. The Contractor, Subcontractors, Subsubcontractors, and suppliers shall not own or claim a copyright in the Instruments of Service. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with the Project is not to be construed as publication in derogation of the Owner's reserved rights.

§ 5.5.2 The Contractor, Subcontractors, Sub-subcontractors and suppliers are authorized to use and reproduce the Instruments of Service provided to them, subject to the protocols established pursuant to Sections 5.6 and 5.7, solely and exclusively for execution of the Work. All copies made under this authorization shall bear the copyright notice, if any, shown on the Instruments of Service. The Contractor, Subcontractors, Sub-subcontractors, and suppliers may not use the Instruments of Service on other projects or for additions to a Project outside the scope of a Contract without the specific written consent of the Owner.

# § 5.6 Digital Data Use and Transmission

The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form.

# § 5.7 Building Information Models Use and Reliance

Any use of, or reliance on, all or a portion of a building information model without agreement to protocols governing the use of, and reliance on, the information contained in the model and without having those protocols set forth in Building Information Modeling and Digital Data Exhibit, and the requisite Project Building Information Modeling Protocol Form, shall be at the using or relying party's sole risk and without liability to the other party and its contractors or consultants, the authors of, or contributors to, the building information model, and each of their agents and employees.

# § 5.8 Severability

The invalidity of any provision of the Contract Documents shall not invalidate the Contract or its remaining provisions. If it is determined that any provision of the Contract Documents violates any law, or is otherwise invalid or unenforceable, then that provision shall be revised to the extent necessary to make that provision legal and enforceable. In such case the Contract Documents shall be construed, to the fullest extent permitted by law, to give effect to the parties' intentions and purposes in executing the Contract.

# **§ 5.9 Notice**

§ **5.9.1** Except as otherwise provided in Section 5.9.2, where the Contract Documents require one party to notify or give notice to the other party, such notice shall be provided in writing to the designated representative of the party to whom the notice is addressed and shall be deemed to have been duly served if delivered in person, by mail, by courier, or by electronic transmission in accordance with Building Information Modeling and Digital Data Exhibit, if completed, or as otherwise set forth below:

Electronic mail may be used by the parties using the email addresses provided by each party in this Master Agreement or in each Work Order.

§ 5.9.2 Notice of Claims shall be provided in writing and shall be deemed to have been duly served only if delivered to the designated representative of the party to whom the notice is addressed by certified or registered mail, or by courier providing proof of delivery.

# § 5.10 Relationship of the Parties

Contractor accepts the relationship of trust and confidence established by this Master Agreement and covenants with the Owner to cooperate with the Architect and exercise the Contractor's skill and judgment in furthering the interests of the Owner; to furnish efficient business administration and supervision; to furnish at all times an adequate supply of workers and materials; and to perform the Work in an expeditious and economical manner consistent with the Owner's interests. The Owner agrees to famish and approve, in a timely manner, information required by the Contractor and to make payments to the Contractor in accordance with the requirements of the Contract Documents.

# ARTICLE 6 OWNER

# § 6.1 Information and Services Required of the Owner

- § 6.1.1 The Owner shall furnish all necessary surveys and a legal description of sites referenced in a Work Order, if required.
- § **6.1.2** The Contractor shall be entitled to rely on the accuracy of information furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work.
- § 6.1.3 Except for permits and fees that are the responsibility of the Contractor under the Contract Documents, including those required under Section 7.6.1, the Owner shall secure and pay for other necessary approvals, easements, assessments and charges required for the construction, use or occupancy of permanent structures or for permanent changes in existing facilities.

# § 6.2 Owner's Right to Stop the Work

If the Contractor fails to correct Work which is not in accordance with the requirements of the Contract Documents, or repeatedly fails to carry out the Work in accordance with the Contract Documents, the Owner may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order is eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity.

# § 6.3 Owner's Right to Carry Out the Work

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents, and fails within a ten-day period after receipt of notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to any other remedies the Owner may have, correct such default or neglect. Such action by the Owner and amounts charged to the Contractor are both subject to prior approval of the Architect and the Architect may, pursuant to Section 13.4.3, withhold or nullify a Certificate for Payment in whole or in part, to the extent reasonably necessary to

reimburse the Owner for the reasonable cost of correcting such deficiencies, including the Owner's expenses and compensation for the Architect's additional services made necessary by such default, neglect, or failure. If the Contractor disagrees with the actions of the Owner or the Architect, or the amounts claimed as costs to the Owner, the Contractor may file a Claim pursuant to Article 19.

# ARTICLE 7 CONTRACTOR

# § 7.1 Review of Contract Documents and Field Conditions by Contractor

- § 7.1.1 Execution of a Work Order by the Contractor is a representation that the Contractor has visited the relevant site, become generally familiar with local conditions under which the Work is to be performed and correlated personal observations with requirements of the Contract Documents.
- § 7.1.2 Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 6.1.1, shall take field measurements of any existing conditions related to that portion of the Work and shall observe any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents; however, the Contractor shall promptly report to the Architect any errors, inconsistencies, or omissions discovered by or made known to the Contractor as a request for information in such form as the Architect may require. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional unless otherwise specifically provided in the Contract Documents.
- § 7.1.3 The Contractor is not required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Contractor shall promptly report to the Architect any nonconformity discovered by or made known to the Contractor as a request for information in such form as the Architect may require.

# § 7.2 Supervision and Construction Procedures

- § 7.2.1 The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for and have control over construction means, methods, techniques, sequences and procedures, and for coordinating all portions of the Work under a Contract, unless the Contract Documents give other specific instructions concerning these matters.
- § 7.2.2 The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons or entities performing portions of the Work.

# § 7.3 Labor and Materials

§ 7.3.1 Unless otherwise provided in the Contract Documents, the Contractor shall provide and

pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.

- § 7.3.2 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not skilled in tasks assigned to them.
- § 7.3.3 The Contractor may make a substitution only with the consent of the Owner, after evaluation by the Architect and in accordance with a Modification.

# § 7.4 Warranty

The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. For a period of one (1) year after the date of Substantial Completion (and longer if extended warranties are provided in the Contract Documents for certain materials, equipment, and installations), the Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. Work, materials, or equipment not conforming to these requirements may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation or normal wear and tear under normal usage. All other warranties required by the Contract Documents shall be issued in the name of the Owner, or shall be transferable to the Owner, and shall commence in accordance with Section 13.6.3.

# § 7.5 Taxes

The Contractor shall pay sales, consumer, use and other similar taxes that are legally enacted when bids are received or negotiations concluded for an individual Contract, whether or not yet effective or merely scheduled to go into effect.

# § 7.6 Permits, Fees, Notices, and Compliance with Laws

- § 7.6.1 Contractor shall secure and pay for the building permit as well as other permits, fees, licenses and inspections by government agencies necessary for proper execution and completion of the Work that are customarily secured after execution of a Contract and legally required at the time bids are received or negotiations concluded. Such permit fees are included in the Contract Sum.
- § 7.6.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities applicable to performance of the Work. If the Contractor performs Work knowing it to be contrary to applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction.

# § 7.7 Allowances

The Contractor shall include in the Contract Sum for each Work Order all allowances stated in the Contract Documents for that Work Order. The Owner shall select materials and equipment under allowances with reasonable promptness. Allowance amounts shall include the costs to the Contractor of materials and equipment delivered at the site and all required taxes, less applicable trade discounts. Contractor's costs for unloading and handling at the site, labor, installation, overhead, profit, and other expenses contemplated for stated allowance amounts shall be included in the Contract Sum but not in the allowance.

# § 7.8 Contractor's Construction Schedules

§ 7.8.1 The Contractor, promptly after executing a Work Order, shall submit for the Owner's and Architect's information a Contractor's construction schedule for the Work described in that Work Order. The schedule shall not exceed time limits current under the Contract Documents, shall be revised at appropriate intervals as required by the conditions of the Work and Project, shall be related to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practicable execution of the Work.

§ 7.8.2 The Contractor shall perform the Work in general accordance with the most recent schedule submitted to the Owner and Architect.

# § 7.9 Submittals

§ 7.9.1 The Contractor shall review for compliance with the Contract Documents and submit to the Architect Shop Drawings, Product Data, Samples and similar submittals required by the Contract Documents in coordination with the Contractor's construction schedule and in such sequence as to allow the Architect reasonable time for review. By submitting Shop Drawings, Product Data, Samples and similar submittals, the Contractor represents to the Owner and Architect that the Contractor has (1) reviewed and approved them; (2) determined and verified materials, field measurements, and field construction criteria related thereto, or will do so; and (3) checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents. The Work shall be in accordance with approved submittals.

§ 7.9.2 Shop Drawings, Product Data, Samples and similar submittals are not Contract Documents.

§ 7.9.3 The Contractor shall not be required to provide professional services that constitute the practice of architecture or engineering unless such services are specifically required by the Contract Documents or unless the Contractor needs to provide such services in order to carry out the Contractor's own responsibilities. If professional design services or certifications by a design professional are specifically required, the Owner and the Architect will specify the performance and design criteria that such services must satisfy. The Contractor shall cause such services or certifications to be provided by an appropriately licensed design professional. If no criteria are specified, the design shall comply with applicable codes and ordinances. Each Party shall be entitled to rely upon the information provided by the other Party. The Architect will review and approve or take other appropriate action on submittals for the limited purpose of checking for conformance with information provided and the design concept expressed in the Contract

Documents. The Architect's review of Shop Drawings, Product Data, Samples, and similar submittals shall be for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. In performing such review, the Architect will approve, or take other appropriate action upon, the Contractor's Shop Drawings, Product Data, Samples, and similar submittals.

# § 7.10 Use of Site

The Contractor shall confine operations at the site(s) to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, lawful orders of public authorities, and the Contract Documents and shall not unreasonably encumber the site(s) with materials or equipment.

# § 7.11 Cutting and Patching

The Contractor shall be responsible for cutting, fitting or patching required to complete the Work or to make its parts fit together properly.

# § 7.12 Cleaning Up

The Contractor shall keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under a Contract. At completion of the Work, the Contractor shall remove waste materials, rubbish, the Contractor's tools, construction equipment, machinery and surplus material from and about the Project.

# § 7.13 Access to Work

The Contractor shall provide the Owner and Architect access to the Work in preparation and progress wherever located.

# § 7.14 Royalties, Patents and Copyrights

The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and Architect harmless from loss on account thereof, but shall not be responsible for defense or loss when a particular design, process, or product of a particular manufacturer or manufacturers is required by the Contract Documents or where the copyright violations are contained in Drawings, Specifications or other documents prepared by the Owner or Architect. However, if an infringement of a copyright or patent is discovered by, or made known to, the Contractor, the Contractor shall be responsible for the loss unless the information is promptly furnished to the Architect.

# § 7.15 Indemnification

§ 7.15.1 Contractor shall indemnify and hold harmless the Owner from any and all liabilities, damages, losses, and costs, including, but not limited to, reasonable attorneys' fees, to the extent caused by the negligence, recklessness, or intentional wrongful conduct of the Contractor, its subcontractors of every tier, or its agents or employees in connection with providing the Work and services called for in this Master Agreement. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity which would otherwise exist as to a party or person described in this Section 7.15.1.

§ 7.15.2 In claims against any person or entity indemnified under this Section 7.15 by an employee of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, the indemnification obligation under Section 7.15.1 shall not be limited by a limitation on amount or type of damages, compensation or benefits payable by or for the Contractor or Subcontractor under workers' compensation acts, disability benefit acts or other employee benefit acts.

# **ARTICLE 8 ARCHITECT**

- § 8.1 The Owner shall retain an Architect to perform the services enumerated in this Article 8 and as described elsewhere in this Master Agreement. If an Architect is not required by law, or otherwise not engaged on the Project, the Owner shall perform such services.
- § 8.2 The Architect listed on each Work Order will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative during construction, until the date the Architect issues the final Certificate for Payment for the Contract. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents, unless otherwise modified in writing in accordance with other provisions of the Contract. However, any approvals, certificates, or decisions of the Architect are subject to the approval of Owner. At Owner's discretion, the duties of the Architect described herein may be performed by the Owner or the Owner's representative.
- § 8.3 Duties, responsibilities, and limitations of authority of the Architect as set forth in the Contract Documents shall not be restricted, modified, or extended without written consent of the Owner, Contractor, and Architect. Consent shall not be unreasonably withheld.
- § 8.4 The Architect will visit the site at intervals appropriate to the stage of the construction to become generally familiar with the progress and quality of the portion of the Work completed, and to determine in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. The Architect will not have control over, charge of, or responsibility for, the construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, since these are solely the Contractor's rights and responsibilities under the Contract Documents.
- § 8.5 On the basis of the site visits, the Architect will keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work. The Architect will not be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect will not have control over or charge of and will not be responsible for acts or omissions of the Contractor, Subcontractors, or their agents or employees, or any other persons or entities performing portions of the Work.

- § **8.6** Based on the Architect's evaluations of the Work and of the Contractor's Applications for Payment, the Architect will review and celtify the amounts due the Contractor and will issue Certificates for Payment in such amounts.
- § **8.7** The Architect has authority to reject Work that does not conform to the Contract Documents and to require inspection or testing of the Work.
- § 8.8 The Architect will review and approve or take other appropriate action upon the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents.
- § 8.9 The Architect will interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect will make initial decisions on all claims, disputes and other matters in question between the Owner and Contractor but will not be liable for results of any interpretations or decisions rendered in good faith.
- § 8.10 The Architect's decisions on matters relating to aesthetic effect will be final if consistent with the intent expressed in the Contract Documents.

# **ARTICLE 9 SUBCONTRACTORS**

- § 9.1 A Subcontractor is a person or entity who has a direct contract with the Contractor to perform a portion of the Work at a Project site.
- § 9.2 Unless otherwise stated in the Contract Documents, the Contractor, as soon as practicable after execution of a Work Order, shall notify the Owner and Architect of the Subcontractors or suppliers proposed for each of the principal portions of the Work. The Contractor shall not contract with any Subcontractor or supplier to whom the Owner or Architect has made reasonable written objection within ten days after receipt of the Contractor's list of Subcontractors and suppliers. If the proposed but rejected Subcontractor was reasonably capable of performing the Work, the Contract Sum and Contract Time shall be increased or decreased by the difference, if any, occasioned by such change, and an appropriate Change Order shall be issued before commencement of the substitute Subcontractor's Work. The Contractor shall not be required to contract with anyone to whom the Contractor has made reasonable objection.
- § 9.3 Contracts between the Contractor and Subcontractors shall (1) require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by the terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the Subcontractor's Work, which the Contractor, by the Contract Documents, assumes toward the Owner and Architect, and (2) allow the Subcontractor the benefit of all rights, remedies and redress against the Contractor that the Contractor, by the Contract Documents, has against the Owner.

### ARTICLE 10 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS

- § 10.1 The term "Separate Contractor(s)" shall mean other contractors retained by the Owner under separate agreements. The Owner reserves the right to perform construction or operations related to a Project with the Owner's own forces, and with Separate Contractors retained under Conditions of the Contract substantially similar to those of this Contract, including those provisions of the Conditions of the Contract related to insurance and waiver of subrogation.
- § 10.2 The Contractor shall afford the Owner and the Owner's separate contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Contractor's activities with theirs as required by the Contract Documents.
- § 10.3 The Owner shall be reimbursed by the Contractor for costs incurred by the Owner which are payable to a separate contractor because of delays, improperly timed activities or defective construction of the Contractor. The Owner shall be responsible to the Contractor for costs incurred by the Contractor because of delays, improperly timed activities, damage to the Work, or defective construction of a separate contractor.

# ARTICLE 11 CHANGES IN THE WORK

- § 11.1 By appropriate Modification, changes in the Work may be accomplished after execution of a Work Order. The Owner, without invalidating the Contract, may order changes in the Work within the general scope of the Contract consisting of additions, deletions, or other revisions, with the Contract Sum and Contract Time being adjusted accordingly. Such changes in the Work shall be authorized by written Change Order signed by the Owner, Contractor, and Architect, or by written Construction Change Directive signed by the Owner and Architect. Upon issuance of the Change Order or Construction Change Directive, the Contractor shall proceed promptly with such changes in the Work, unless otherwise provided in the Change Order or Construction Change Directive.
- § 11.2 Adjustments in the Contract Sum and Contract Time resulting from a change in the Work shall be determined by mutual agreement of the parties or, in the case of a Construction Change Directive signed only by the Owner and Architect, by the Contractor's cost of labor, material, equipment, and reasonable overhead and profit as listed in Exhibit A, unless the parties agree on another method for determining the cost or credit. Pending final determination of the total cost of a Construction Change Directive, the Contractor may request payment for Work completed pursuant to the Construction Change Directive. The Architect will make an interim determination of the amount of payment due for purposes of certifying the Contractor's monthly Application for Payment. When the Owner and Contractor agree on adjustments to the Contract Sum and Contract Time arising from a Construction Change Directive, the Architect will prepare a Change Order.
- § 11.3 Subject to the Owner's written approval, the Architect will have authority to order minor changes in the Work not involving adjustment in the Contract Sum or extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes shall be effected by written order and shall be binding on the Owner and Contractor. The Contractor shall

carry out such written orders promptly. If the Contractor believes that the proposed minor change in the Work will affect the Contract Sum or Contract Time, the Contractor shall notify the Architect and shall not proceed to implement the change in the Work.

- § 11.4 If concealed or unknown physical conditions are encountered at the site that differ materially from those indicated in the Contract Documents or from those conditions ordinarily found to exist, the Contract Sum and Contract Time shall be equitably adjusted as mutually agreed between the Owner and Contractor; provided that the Contractor provides notice to the Owner and Architect promptly and before conditions are disturbed.
- § 11.5 For Change Orders or Construction Change Directive Work, the amount for overhead and for profit added or deducted shall be a total of ten percent (10%) of the Cost of the Work that is added or deducted, respectively ("Contractor's Fee"). Cost of the Work is defined in Exhibit "A". Such Contractor's Fee includes all profit, overhead, general conditions costs, and insurance (but not additional bond costs, if applicable), except if a Change Order extends the Date of Substantial Completion, then Contractor's reasonable extended daily general conditions costs shall be added to the cost of the Change Order per §12.5 of the Master Agreement. Likewise, subcontractors shall not add general conditions costs, except that if the change includes an increase in the Contract Time, then extended daily general conditions costs may be added in accordance with §12.5 of the Master Agreement. However, fee shall not be reduced for deductive Change Orders arising as a result of the Sales Tax Savings Program of Article 21.12 of the Master Agreement, except fee on the amount of the sales tax saved shall be deducted.
  - 1. For Change Order Work accomplished by the Contractor's own forces, acting in the role of a subcontractor, overhead and profit shall be a maximum of 10% of the Cost of the Work.
  - **2.** For Change Order Work accomplished by Subcontractors, Subcontractor's overhead and profit shall be a maximum of 10% of the Cost of the Work.
  - **3.** For each Subcontractor, for Work performed by that Subcontractor's subsubcontractor, no overhead charges, but a 5% fee for overhead and profit on the amount due the sub-subcontractor shall be paid the Subcontractor.
  - **4.** For each Sub-subcontractor, for Work performed by that Sub-subcontractor, overhead and profit shall be a maximum of 10% of the Cost of the Work.
  - **5.** No further tiering of sub-subcontractors will be allowed mark up for fees, overhead, or profit.

#### **ARTICLE 12 TIME**

- § 12.1 Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Work Order, the Contractor confirms that the Contract Time is a reasonable period for performing the Work.
- § 12.2 Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work.
- § 12.3 The term "day" as used in the Contract Documents shall mean calendar day unless

otherwise specifically defined.

§ 12.4 The date of Substantial Completion is the date certified by the Architect in accordance with Section 13.6.3.

§ 12.5 If the Contractor is delayed at any time in the commencement or progress of the Work of any Work Order by changes ordered in the Work, by fire, acts of God, unusual, extraordinary adverse weather events not reasonably anticipatable, unavoidable casualties or any causes beyond the Contractor's control or responsibility, or by other causes which the Architect determines may justify delay, then the Contract Time shall be extended by Change Order for such reasonable time as the Architect may determine, subject to the provisions of Article 21; however, the Contract Sum shall not be increased, except for a per diem payment as described below. It is the express and bargained for intent of the parties that the risk of any monetary damages caused by any delays described in this Section or any other delays from any other cause are accepted and assumed entirely by Contractor, and in no event shall any claim relating thereto for an increase in the Contract Sum be made or recognized, except for the per diem payment as described below. Contractor's sole remedy for any delay, impact, disruption, or interruption caused by any of the reasons listed in this Section shall be an equitable extension of time to perform the Work for each day of such delay that impacts the critical path schedule of the Project that extends the Date of Substantial Completion, and a per diem payment not to exceed the Contractor's reasonable extended daily general conditions costs per day for each day of such delay that impacts the critical path schedule of the Project that extends the Date of Substantial Completion.

Contractor shall not make any other claim nor seek any other damages of any kind against Owner or Architect for any delays, impacts, disruptions, or interruptions of any kind. If a Contractor caused delay runs concurrent with such delays, Contractor shall not be entitled to any extension of time or per diem payment for the concurrent period of delay. Delays caused by labor disputes, delivery delays, and dispute resolution proceedings are considered within the Contractor's control and shall not be grounds for a delay claim. This Section does preclude recovery of damages for delay by Contractor under any other provisions of the Contract Documents.

Extensions of time will be granted only if the item, task, or other phase of construction delayed is critical to the Work and so indicated in the Contractor's Schedule.

Extensions of time due to unusual, extraordinary adverse weather events not reasonably anticipatable will be granted only because of such inclement weather occurring on a normal working day and preventing the execution of the major or critical item of construction ordinarily performed at the time. In addition, extensions of time for such unusual extraordinary adverse weather events not reasonably anticipatable will be considered only of such inclement weather exceeds by twice that normally recorded by the National Weather Bureau for the same month and location on a 20-year average basis.

#### ARTICLE 13 PAYMENTS AND COMPLETION

#### § 13.1 Schedule of Values

- § 13.1.1 Where the Contract is based on a Stipulated Sum or the Cost of the Work with a Guaranteed Maximum Price pursuant to the Work Order, the Contractor shall submit a schedule of values to the Architect before the first Application for Payment, allocating the entire Stipulated Sum or Guaranteed Maximum Price to the various portions of the Work. The schedule of values shall be prepared in the form, and supported by the data to substantiate its accuracy required by the Architect. This schedule of values shall be used as a basis for reviewing the Contractor's Applications for Payment.
- § 13.1.2 The allocation of the Stipulated Sum or Guaranteed Maximum Price under this Section 13.1 shall not constitute a separate stipulated sum or guaranteed maximum price for each individual line item in the schedule of values.

#### § 13.2 Control Estimate

§ 13.2.1 Where the Contract Sum is the Cost of the Work, plus the Contractor's Fee without a' Guaranteed Maximum Price pursuant to the Work Order, the Contractor shall prepare and submit to the Owner a Control Estimate within 14 days of executing the Work Order. The Control Estimate shall include the estimated Cost of the Work plus the Contractor's Fee.

#### § 13.2.2 The Control Estimate shall include:

- 1. the Contract Documents enumerated in the Work order, including all Modifications thereto:
- 2. a list of the assumptions made by the Contractor in the preparation of the Control Estimate to supplement the information provided by the Owner and contained in the Contract Documents:
- **3.** a statement of the estimated Cost of the Work organized by trade categories or systems and the Contractor's Fee;
- 4. a project schedule upon which the Control Estimate is based, indicating proposed Subcontractors, activity sequences and durations, milestone dates for receipt and approval of pertinent information, schedule of shop drawings and samples, procurement and delivery of materials or equipment, the Owner's occupancy requirements, and the date of Substantial Completion; and
- 5. a list of any contingency amounts included in the Control Estimate for further development of design and construction.
- § 13.2.3 When the Control Estimate is acceptable to the Owner and Architect, the Owner shall acknowledge it in writing. The Owner's acceptance of the Control Estimate does not imply that the Control Estimate constitutes a Guaranteed Maximum Price.
- § 13.2.4 The Contractor shall develop and implement a detailed system of cost control that will provide the Owner and Architect with timely information as to the anticipated total Cost of the Work. The cost control system shall compare the Control Estimate with the actual cost for activities in progress and estimates for uncompleted tasks and proposed changes. This information shall be reported to the Owner, in writing, no later than the Contractor's first Application for Payment and shall be revised and submitted with each Application for Payment.

§ 13.2.5 The Owner shall authorize preparation of revisions to the Contract Documents that incorporate the agreed-upon assumptions contained in the Control Estimate. The Owner shall promptly furnish such revised Contract Documents to the Contractor. The Contractor shall notify the Owner and Architect of any inconsistencies between the Control Estimate and the revised Contract Documents.

#### § 13.3 Applications for Payment

- § 13.3.1 Applications for Payment will be submitted individually for each Contract as set forth in this section 13.3.
- § 13.3.2 At least ten days before the date established for each progress payment, the Contractor shall submit to the Architect an itemized Application for Payment prepared in accordance with the schedule of values, if required under Section 13.1, for completed portions of the Work. The application shall be notarized, if required; be supported by all data substantiating the Contractor's right to payment that the Owner or Architect require; shall reflect retainage if provided for in the Contract Documents; and include any revised cost control information required by Section 13.2.4. Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not intend to pay a Subcontractor or supplier, unless such Work has been performed by others whom the Contractor intends to pay.
- § 13.3.3 With each Application for Payment where the Contract Sum is based upon the Cost of the Work with a Guaranteed Maximum Price, the Contractor shall submit payrolls, petty cash accounts, receipted invoices or invoices with check vouchers attached, and any other evidence required by the Owner to demonstrate that cash disbursements already made by the Contractor on account of the Cost of the Work equal or exceed progress payments already received by the Contractor plus payrolls for the period covered by the present Application for Payment, less that portion of the progress payments attributable to the Contractor's Fee.
- § 13.3.4 Payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment stored, and protected from damage, off the site at a location agreed upon in writing.
- § 13.3.5 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information and belief, be free and clear of liens, claims, security interests or other encumbrances adverse to the Owner's interests.

#### § 13.4 Certificates for Payment

§ 13.4.1 The Architect will, within seven days after receipt of the Contractor's Application for Payment, either issue to the Owner a Certificate for Payment, with a copy to the Contractor, for such amount as the Architect determines is properly due, or notify the Contractor and Owner of the Architect's reasons for withholding certification in whole or in part as provided in Section

- 13.4.3. The Architect's Certificate of Payment is subject to the approval of the Owner.
- § 13.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner, based on the Architect's evaluations of the Work and the data in the Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated, the quality of the Work is in accordance with the Contract Documents, and the Contractor is entitled to payment in the amount certified. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to correction of minor deviations from the Contract Documents prior to completion and to specific qualifications expressed by the Architect. However, the issuance of a Certificate for Payment will not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work; (2) reviewed construction means, methods, techniques, sequences or procedures; (3) reviewed copies of requisitions received from Subcontractors and suppliers and other data requested by the Owner to substantiate the Contractor's right to payment; or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.
- § 13.4.3 The Architect may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the Architect's opinion the representations to the Owner required by Section 13.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Section 13.4.1. If the Contractor and the Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which the Architect is able to make such representations to the Owner. The Architect may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Architect's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 7.2.2, because of
  - 1. defective Work not remedied;
  - 2. third-party claims filed or reasonable evidence indicating probable filing of such claims unless security acceptable to the Owner is provided by the Contractor;
  - 3. failure of the Contractor to make payments properly to Subcontractors or suppliers for labor, materials or equipment;
  - **4.** reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum:
  - **5.** damage to the Owner or a Separate Contractor;
  - 6. reasonable evidence that the Work will not be completed within the Contract Time and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay; or
  - 7. repeated failure to carry out the Work in accordance with the Contract Documents.
- § 13.4.5 The Architect may not withhold a Certificate for Payment in whole or in part, and the Owner shall not withhold payments to the Contractor, pertaining to one Contract to offset amounts in dispute under a separate Contract.

§ 13.4.6 When either party disputes the Architect's decision regarding a Certificate for Payment under Section 13.4.3, in whole or in part, that party may submit a Claim in accordance with Article 19.

#### § 13.5 Progress Payments

- § 13.5.1 The Contractor shall pay each Subcontractor, no later than seven days after receipt of payment from the Owner, the amount to which the Subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to sub-subcontractors in a similar manner. As a condition to each progress payment the Contractor shall have furnished Owner with a partial lien waiver and release signed by Contractor, conditioned upon payment for all Work performed that is included in the current Application for Payment in the form attached to the Contract or if not attached, in a form approved by Owner, and shall have furnished Owner with partial unconditional lien waivers and releases signed by all subcontractors, suppliers, persons or entities furnishing any labor or material, equipment, services, and materials for the Project and for all Work performed by same that is included in the respective prior Application for Payment. In addition, as a condition to each progress payment, in the event that any liens have been filed against the Project, the Contractor shall have either recorded a satisfaction of such lien or transferred the lien to a bond satisfactory to Owner within ten (10) days of notice from Owner.
- § 13.5.2 Neither the Owner nor Architect shall have an obligation to pay or see to the payment of money to a Subcontractor or supplier except as may otherwise be required by law.
- § 13.5.3 A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Contract Documents.
- § 13.5.4 Provided the Owner has fulfilled its payment obligations under the Contract Documents, the Contractor shall defend and indemnify the Owner from all loss, liability, damage or expense, including reasonable attorney's fees and litigation expenses, arising out of any lien claim or other claim for payment by any Subcontractor or supplier of any tier. Upon receipt of notice of a lien claim or other claim for payment, the Owner shall notify the Contractor. If approved by the applicable court, when required, the Contractor may substitute a surety bond for the property against which the lien or other claim for payment has been asserted.

#### § 13.6 Substantial Completion

- § 13.6.1 Substantial Completion is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work for its intended use.
- § 13.6.2 When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Architect a comprehensive list of items to be completed or corrected prior to final payment. Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.

- § 13.6.3 Upon receipt of the Contractor's list, the Architect will make an inspection to determine whether the Work or designated portion thereof is substantially complete. When the Architect determines that the Work or designated portion thereof is substantially complete, the Architect will issue a Certificate of Substantial Completion which shall establish the date of Substantial Completion; establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance; and fix the time within which the Contractor shall finish all items on the list accompanying the Certificate. Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.
- § 13.6.4 The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in the Certificate. Upon such acceptance and consent of surety, if any, the Owner shall make payment of retainage applying to the Work or designated portion thereof. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents.

#### § 13.7 Final Completion and Final Payment

- § 13.7.1 Upon receipt of the Contractor's notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection and, when the Architect finds the Work acceptable under the Contract Documents and the Contract fully performed, the Architect will promptly issue a final Certificate for Payment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architect's on-site visits and inspections, the Work has been completed in accordance with the Contract Documents and that the entire balance found to be due the Contractor and noted in the final Certificate is due and payable. The Architect's final Certificate for Payment will constitute a further representation that conditions stated in Section 13.7.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.
- § 13.7.2 Final payment shall not become due until the Contractor has delivered to the Owner a complete release of all liens arising out of the Contract or receipts in full covering all labor, materials and equipment for which a lien could be filed, or a bond satisfactory to the Owner to indemnify the Owner against such lien. If such lien remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging such lien, including costs and reasonable attorneys' fees.
- § 13.7.3 The making of final payment shall constitute a waiver of claims by the Owner except those arising from
  - 1. liens, claims, security interests or encumbrances arising out of the Contract and unsettled;
  - 2. failure of the Work to comply with the requirements of the Contract Documents;
  - 3. terms of special warranties required by the Contract Documents; or
  - **4.** audits performed by the Owner, if permitted by the Contract Documents, after final payment

§ 13.7.4 Acceptance of final payment by the Contractor, a Subcontractor or supplier shall constitute a waiver of claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of the final Application for Payment.

#### ARTICLE 14 PROTECTION OF PERSONS AND PROPERTY

#### § 14.1 Safety Precautions and Programs

The Contractor shall be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of a Contract. The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury or loss to

- 1. employees on the Work and other persons who may be affected thereby;
- 2. the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody, or control of the Contractor or the Contractor's Subcontractors or Sub-subcontractors; and
- **3.** other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation, or replacement in the course of construction.

The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules, and regulations, and lawful orders of public authorities bearing on safety of persons and property and their protection from damage, injury, or loss. The Contractor shall promptly remedy damage and loss to property caused in whole or in part by the Contractor, a Subcontractor, a sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Sections 14.1.2 and 14.1.3, except for damage or loss attributable to acts or omissions of the Owner or Architect or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 7.15.

#### § 14.2 Hazardous Materials

§ 14.2.1 The Contractor is responsible for compliance with the requirements of the Contract Documents regarding hazardous materials or substances. If the Contractor encounters a hazardous material or substance not addressed in the Contract Documents, and if reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered on the site by the Contractor, the Contractor shall, upon recognizing the condition, immediately stop Work in the affected area and notify the Owner and Architect of the condition. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Contractor. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum shall be increased in the amount of the Contractor's reasonable additional costs of shutdown, delay, and start-up.

#### ARTICLE 15 INSURANCE AND BONDS

#### § 15.1 Contractor's Insurance

The Contractor shall purchase and maintain insurance coverage required in this Article 15 and as otherwise required for a specific project as set forth in a Work Order.

- § 15.1.1 The Contractor shall purchase and maintain insurance of the types and limits of liability, containing the endorsements, and subject to the terms and conditions, as described in this Section 15.1 or elsewhere in the Contract Documents. The Contractor shall purchase and maintain the insurance required by this Master Agreement from an insurance company or insurance companies lawfully authorized to issue insurance in the jurisdiction where the Project is located. The Contractor shall maintain the required insurance until the expiration of the period for correction of Work as set forth in Section 16.4, unless a different duration is stated below:
- § 15.1.2 Commercial General Liability insurance for the Project written on an occurrence form with policy limits of not less than One Million (\$1,000,000) each occurrence, Two Million (\$2,000,000) general aggregate, and Two Million (\$2,000,000) aggregate for products-completed operations hazard, providing coverage for claims including
  - 1. damages because of bodily injury, sickness or disease, including occupational sickness or disease, and death of any person;
  - 2. personal and advertising injury;
  - **3.** damages because of physical damage to or destruction of tangible property, including the loss of use of such property;
  - 4. bodily injury or property damage arising out of completed operations; and
  - 5. the Contractor's indemnity obligations under Section 7.15.
- § 15.1.3 Automobile Liability covering vehicles owned by the Contractor and non-owned vehicles used by the Contractor, with policy limits of not less than One Million (\$1,000,000) per accident, for bodily injury, death of any person, and property damage arising out of the ownership, maintenance, and use of those motor vehicles along with any other statutorily required automobile coverage.
- § 15.1.4 The Contractor may achieve the required limits and coverage for Commercial General Liability and Automobile Liability through a combination of primary and excess or umbrella liability insurance, provided such primary and excess or umbrella insurance policies result in the same or greater coverage as those required under Section 15.1.2 and 15.1.3, and in no event shall any excess or umbrella liability insurance provide narrower coverage than the primary policy. The excess policy shall not require the exhaustion of the underlying limits only through the actual payment by the underlying insurers.
- § **15.1.5** Workers' Compensation at statutory limits.
- § **15.1.6** Employers' Liability with policy limits not less than One Million (\$1,000,000) each accident, One Million (\$1,000,000) each employee, and One Million (\$1,000,000) policy limit.
- § 15.1.7 If the Contractor is required to furnish professional services as part of the Work, the Contractor shall procure Professional Liability insurance covering performance of the professional services, with policy limits of not less than One Million (\$1,000,000) per claim and One Million (\$1,000,000) in the aggregate.

- § 15.1.8 If the Work involves the transport, dissemination, use, or release of pollutants, the Contractor shall procure Pollution Liability insurance, with policy limits of not less than One Million (\$1,000,000) per claim and One Million (\$1,000,000) in the aggregate.
- § **15.1.9** Coverage under Sections 15.1.7 and 15.1.8 may be procured through a Combined Professional Liability and Pollution Liability insurance policy, with combined policy limits of not less than One Million (\$1,000,000) per claim and One Million (\$1,000,000) in the aggregate.
- § 15.1.10 The Contractor shall provide certificates of insurance acceptable to the Owner evidencing compliance with the requirements in this Section 15.1 at the following times: (1) prior to commencement of the Work; (2) upon renewal or replacement of each required policy of insurance; and (3) upon the Owner's written request. An additional certificate evidencing continuation of liability coverage, including coverage for completed operations, shall be submitted with the final Application for Payment and thereafter upon renewal or replacement of such coverage until the expiration of the period required by Section 15.1.1. The certificates will show the Owner as an additional insured on the Contractor's Commercial General Liability and excess or umbrella liability policy.
- § 15.1.11 The Contractor shall disclose to the Owner any deductible or self-insured retentions applicable to any insurance required to be provided by the Contractor.
- § 15.1.12 To the fullest extent permitted by law, the Contractor shall cause the commercial liability coverage required by this Section 15.1 to include (1) the Owner, the Architect, and the Architect's Consultants as additional insureds for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's operations; and (2) the Owner as an additional insured for claims caused in whole or in part by the Contractor's negligent acts or omissions for which loss occurs during completed operations. The additional insured coverage shall be primary and non-contributory to any of the Owner's general liability insurance policies and shall apply to both ongoing and completed operations. To the extent commercially available, the additional insured coverage shall be no less than that provided by Insurance Services Office, Inc. (ISO) forms CG 20 10 07 04, CG 20 37 07 04, and, with respect to the Architect and the Architect's Consultants, CG 20 32 07 04. Certificates of Insurance acceptable to the Owner shall be filed with the Owner prior to commencement of the Work. Each policy shall contain a provision that the policy will not be canceled or allowed to expire until at least 30 days' prior written notice has been given to the Owner.
- § 15.1.13 Within three (3) business days of the date the Contractor becomes aware of an impending or actual cancellation or expiration of any insurance required by this Section 15.1, the Contractor shall provide notice to the Owner of such impending or actual cancellation or expiration. Upon receipt of notice from the Contractor, the Owner shall, unless the lapse in coverage arises from an act or omission of the Owner, have the right to stop the Work until the lapse in coverage has been cured by the procurement of replacement coverage by the Contractor. The furnishing of notice by the Contractor shall not relieve the Contractor of any contractual obligation to provide any required coverage.

#### § 15.1.14 Other Insurance Provided by the Contractor

**Coverage** Limits

Excess Liability, Umbrella form Three Million (\$3,000,000)

#### § 15.2 Owner's Insurance

#### 15.2.1 Owner's Liability Insurance

The Owner shall be responsible for purchasing and maintaining the Owner's usual liability insurance.

#### § 15.2.2 Property Insurance Provided by the Contractor

- § 15.2.2.1 For each Work Order, the Contractor shall purchase and maintain, from an insurance company or insurance companies lawfully authorized to issue insurance in the jurisdiction where the Project is located, property insurance written on a builder's risk "all-risks" completed value or equivalent policy form and sufficient to cover the total value of the entire Project on a replacement cost basis. The property insurance coverage shall be no less than the amount of the initial Contract Sum, plus the value of subsequent Modifications and labor performed or materials or equipment supplied by others. The property insurance shall be maintained until Substantial Completion and thereafter as provided in Section 15.2.2, unless otherwise provided in the Contract Documents or otherwise agreed in writing by the parties to this Master Agreement. This insurance shall include the interests of the Owner, Contractor, Subcontractors, and Sub-subcontractors in the Project as insureds. This insurance shall include the interests of mortgagees as loss payees.
- § 15.2.2.2 Unless the parties agree otherwise, upon Substantial Completion, the Contractor shall continue the insurance required by Section 15.2.2.1 or, if necessary, replace the insurance policy required under Section 15.2.2.1 with property insurance written for the total value of the Project that shall remain in effect until expiration of the period for correction of the Work set forth in Section 16.4.
- § 15.2.2.3 If the insurance required by this Section 15.2.2 is subject to deductibles or self-insured retentions, the Owner shall be responsible for all loss not covered because of such deductibles or retentions, unless such costs are caused by the acts, omissions, or negligence of Contractor and in such event, Contractor shall pay such costs not covered because of the deductibles.
- § 15.2.2.4 For each Work Order, if the Work involves remodeling an existing structure or constructing an addition to an existing structure, the Owner shall purchase and maintain, until the expiration of the period for correction of Work as set forth in Section 16.4, "all-risks" property insurance, on a replacement cost basis, protecting the existing structure against direct physical loss or damage, notwithstanding the undertaking of the Work. The Owner shall be responsible for all co-insurance penalties.
- § 15.2.2.5 Prior to commencement of the Work in each Work Order, the Contractor shall secure the insurance, and provide evidence of the coverage, required under this Section 15.2.2 and, upon the Owner's request, provide a copy of the property insurance policy or policies required by this Section 15.2.2. The copy of the policy or policies provided shall contain all applicable conditions, definitions, exclusions, and endorsements.

§ 15.2.2.6 Within three (3) business days of the date the Contractor becomes aware of an impending or actual cancellation or expiration of any insurance required by this Section 15.2.2, the Contractor shall provide notice to the Owner of such impending or actual cancellation or expiration.

#### § 15.2.2.7 Waiver of Subrogation

§ 15.2.2.7.1 The Owner and Contractor waive all rights against (1) each other and any of their subcontractors, sub-subcontractors, agents, and employees, each of the other; (2) the Architect and Architect's consultants; and (3) Separate Contractors, if any, and any of their subcontractors, sub-subcontractors, agents, and employees, for damages caused by fire, or other causes of loss, to the extent those losses are covered by and paid by property insurance required by this Master Agreement or other property insurance applicable to the Project where the loss occurred, except such rights as they have to proceeds of such insurance. The Owner or Contractor, as appropriate, shall require similar written waivers in favor of the individuals and entities identified above from the Architect, Architect's consultants, Separate Contractors, subcontractors, and subsubcontractors. The policies of insurance purchased and maintained by each person or entity agreeing to waive claims pursuant to this Section 15.2.2.7 shall not prohibit this waiver of subrogation. This waiver of subrogation shall be effective as to a person or entity (1) even though that person or entity would otherwise have a duty of indemnification, contractual or otherwise, (2) even though that person or entity did not pay the insurance premium directly or indirectly, or (3) whether or not the person or entity had an insurable interest in the damaged property.

§ 15.2.2.7.2 If during a Project construction period the Owner insures properties, real or personal or both, at or adjacent to the site by property insurance under policies separate from those insuring the Project, or if after final payment property insurance is to be provided on the completed Project through a policy or policies other than those insuring the Project during the construction period, to the extent permissible by such policies, the Owner waives all rights in accordance with the terms of Section 15.2.2.7.1 for damages caused by fire or other causes of loss covered by this separate property insurance.

§ 15.2.2.8 A loss insured under the Owner's property insurance shall be adjusted by the Owner as fiduciary and made payable to the Owner as fiduciary for the insureds, as their interests may appear, subject to requirements of any applicable mortgagee clause. The Owner shall pay the Architect and Contractor their just shares of insurance proceeds received by the Owner, and by appropriate agreements, written where legally required for validity, the Architect and Contractor shall make payments to their consultants and Subcontractors in similar manner.

### § 15.2.3 Other Insurance Provided by the Owner

**Coverage** Limits

Discretionary with Owner

#### § 15.3 Performance Bond and Payment Bond

Contractor shall furnish unconditional performance and payment bonds in an amount equal to 100% of the Contract Sum for each Work Order covering faithful performance of the Agreement

and payment of all obligations arising thereunder. The cost of the bonds is included in the Contract Sum. The bonds shall comply with the requirements of Montana Code Annotated; however, "Conditional" Payment Bonds shall **not** be acceptable. Proper Power of Attorney shall accompany said bonds. Said bonds shall be delivered to the Owner. Contractor shall record the bonds in the public records of the County where the Project is located. No Work shall commence on each Project until the Notice of Commencement and copies of the Payment Bonds are recorded and certified copies thereof are posted at the Project site.

- § 15.3.1 The Owner shall have the right to require the Contractor to furnish bonds covering faithful performance of the Contract and payment of obligations arising thereunder as stipulated in the Contract Documents on the date of execution of the Contract.
- § 15.3.2 Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished.

#### ARTICLE 16 CORRECTION OF WORK

- § 16.1 The Contractor shall promptly correct Work rejected by the Architect or failing to conform to the requirements of the Contract Documents, whether discovered before or after Substantial Completion and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for the Architect's services and expenses made necessary thereby, shall be at the Contractor's expense.
- § 16.2 In addition to the Contractor's obligations under Section 7.4, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 13.6.3, or by terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of notice from the Owner to do so unless the Owner has previously given the Contractor a written acceptance of such condition. The Owner shall give such notice promptly after discovery of the condition. During the one-year period for correction of Work, if the Owner fails to notify the Contractor and give the Contractor an opportunity to make the correction, the Owner waives the rights to require correction by the Contractor and to make a claim for breach of warranty.
- § 16.3 If the Contractor fails to correct nonconforming Work within a reasonable time, the Owner may correct it in accordance with Section 6.3.
- § 16.4 The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work.
- § 16.5 The one-year period for correction of Work shall not be extended by corrective Work performed by the Contractor pursuant to this Article 16.

#### **ARTICLE 17 MISCELLANEOUS PROVISIONS**

#### § 17.1 Assignment of Contract

Neither party to a Contract shall assign the Contract without written consent of the other, except that the Owner may, without consent of the Contractor, assign the Contract to a lender providing construction financing for the Project if the lender assumes the Owner's rights and obligations under the Contract Documents. The Contractor shall execute all consents reasonably required to facilitate such assignment.

#### § 17.2 Governing Law

The Work Order shall be governed by the law of the place where the Project is located, excluding that jurisdiction's choice of law rules. If the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 19.6. Venue for any dispute shall be exclusively in the State Courts in the County where the Project is located.

#### § 17.3 Tests and Inspections

Tests, inspections, and approvals of portions of the Work required by the Contract Documents or by applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities shall be made at an appropriate time. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections, and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections, and approvals. The Contractor shall give the Architect timely notice of when and where tests and inspections are to be made so that the Architect may be present for such procedures. The Owner shall bear costs of tests, inspections, or approvals that do not become requirements until after bids are received or negotiations concluded. The Owner shall directly arrange and pay for tests, inspections, or approvals where building codes or applicable laws or regulations require.

§ 17.4 Neither the Owner's nor the Contractor's representatives, for this Master Agreement or for individual Work Orders, identified in accordance with sections 1.4 and 1.5, shall be changed without ten days' prior notice to the other party.

#### ARTICLE 18 TERMINATION

#### § 18.1 Termination of a Contract

A Contract may be terminated in accordance with this Article 18. Termination of a Contract under this Article 18 shall not be deemed a termination of any other Contract created pursuant to this Master Agreement.

#### § 18.1.1 Termination by the Contractor

If the Architect fails to certify payment as provided in Section 13.4.1 for a period of 30 days through no fault of the Contractor, or if the Owner fails to make payment as provided in Section 3.1.5 for a period of 30 days, the Contractor may, upon seven additional days' notice to the Owner and the Architect, terminate the Contract and recover from the Owner payment for Work

executed. Contractor shall not be entitled to any payment, damages, fees, costs, or profits for Work not yet performed.

#### § 18.1.2 Termination by the Owner for Cause

- § 18.1.2.1 The Owner may terminate a Contract if the Contractor
  - 1. repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
  - **2.** fails to make payment to Subcontractors for materials or labor in accordance with the respective agreements between the Contractor and the Subcontractors;
  - 3. repeatedly disregards applicable laws, statutes, ordinances, codes, rules and regulations or lawful orders of a public authority; or
  - **4.** otherwise is guilty of substantial breach of a provision of the Contract Documents.
- § 18.1.2.2 When any of the reasons described in Section 18.1.2. I exists, the Owner, may, without prejudice to any other remedy the Owner may have and after giving the Contractor seven days' notice, terminate a Contract and take possession of the site and of all materials, equipment, tools, and construction equipment and machinery thereon owned by the Contractor and may finish the Work by whatever reasonable method the Owner may deem expedient. Upon request of the Contractor, the Owner shall furnish to the Contractor a detailed accounting of the costs incurred by the Owner in finishing the Work.
- § 18.1.2.3 When the Owner terminates a Contract for one of the reasons stated in Section 18.1.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished.
- § 18.1.2.4 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. The amount to be paid to the Contractor or Owner, as the case may be, shall be certified by the Architect, upon application, and this obligation for payment shall survive termination of the Contract.
- § 18.1.2.5 In the event the Owner terminates this Contract for cause, but a Court later determines that the Owner did not have cause, then such termination shall be automatically deemed a termination for convenience pursuant to §18.1.3 below.

#### §18.1.3 Termination by the Owner for Convenience

The Owner may, at any time, terminate a Contract for the Owner's convenience and without cause. As the Contractor's sole remedy hereunder, Owner shall pay the Contractor for the Work completed as of the effective date of the termination of this Agreement. Contractor shall not be entitled to any payment, damages, fees, costs, or profits for Work not yet performed.

§ 18.2 In the event of any termination, the Contractor shall upon Owner's request, assign any and all subcontracts and purchase orders to Owner. Contractor shall have clauses in all its subcontracts and purchase orders allowing such assignment to Owner. As a precondition to any payment due Contractor as a result of any termination of this Agreement, Contractor shall execute any and all documents necessary to assign all rights and benefits of such subcontracts and purchase orders to Owner.

#### **ARTICLE 19 CLAIMS AND DISPUTES**

§ 19.1 Claims, disputes, and other matters in question arising out of or relating to a Contract executed pursuant to this Master Agreement, including those alleging an error or omission by the Architect but excluding those arising under Section 14.2, shall be referred initially to the Architect for decision. Such matters, except those waived as provided for in Section 19.11 and Sections 13.7.3 and 13.7.4, shall, after initial decision by the Architect or 30 days after submission of the matter to the Architect, be subject to mediation as a condition precedent to binding dispute resolution. Venue for mediation shall be exclusively in the State Courts of the County where the Project is located.

#### § 19.2 Notice of Claims

- § 19.2.1 Claims by either the Owner or Contractor, where the condition giving rise to the Claim is first discovered prior to expiration of the period for correction of the Work set forth in Section 16.2, shall be initiated by notice to the Architect within 21 days after occurrence of the event giving rise to such Claim or within 21 days after the claimant first recognizes the condition giving rise to the Claim, whichever is later.
- § 19.2.2 Claims by either the Owner or Contractor, where the condition giving rise to the Claim is first discovered after expiration of the period for correction of the Work set forth in Section I6.2, shall be initiated by notice to the other party.

#### § 19.3 Time Limits on Claims

The Owner and Contractor shall commence all claims and causes of action against the other and arising out of or related to the Contract in accordance with the requirements of the final dispute resolution method selected in this Master Agreement, whether in contract, tort, breach of warranty, or otherwise, within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Contractor waive all claims and causes of action not commenced in accordance with this Section 19.3. Venue for any dispute shall be exclusively in the State Courts where the Project is located.

- § 19.4 If a claim, dispute or other matter in question relates to or is the subject of a mechanic's lien, the party asserting such matter may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.
- § 19.5 The parties shall endeavor to resolve their disputes by mediation which, unless the parties mutually agree otherwise, shall be in accordance with Monana Statutes.

#### § 19.6 Continuing Contract Performance

Pending final resolution of a Claim, except as otherwise agreed in writing, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents.

#### § 19.7 Waiver of Claims for Consequential Damages

The Contractor and Owner waive claims against each other for consequential damages arising out of or relating to any Contracts formed pursuant to this Master Agreement. This mutual waiver includes

- 1. damages incurred by the Owner for rental expenses, for losses of use, income, profit, financing, business and reputation, and for loss of management or employee productivity or of the services of such persons; and
- 2. damages incurred by the Contractor for losses of use, principal office expenses including the compensation of personnel stationed there, for losses of financing, business and reputation, lost bonding capacity, loss of future work, loss of productivity, lost rentals, and for loss of profit except anticipated profit arising directly from the Work properly performed.

This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination in accordance with Article 18. Nothing contained in this Section 19.11 shall be deemed to preclude an award of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents. The Parties agree that costs to repair, correct, or remedy defective Work are not considered consequential damages.

#### ARTICLE 20 SCOPE OF THIS MASTER AGREEMENT

- § 20.1 This Master Agreement represents the entire and integrated Master Agreement between the Owner and the Contractor and supersedes all prior negotiations, representations or agreements, either written or oral. This Master Agreement may be amended only by written instrument signed by both Owner and Contractor.
- § 20.2 This Master Agreement is comprised of the following documents listed below:
  - 1. Standard Form Master Agreement Between Owner and Contractor;
  - **2.** Exhibit A, Determination of the Cost of the Work.
  - **3.** Other documents:

**Exhibit B** - Scope of Work

Exhibit C - form of Work Order.

#### **ARTICLE 21 OTHER PROVISIONS**

- § 21.1 Owner shall be entitled to make any decision or approval required by this Agreement to be made by the Architect. Any decision of Owner shall supersede any decision of the Architect. Any time Contractor shall be required to notify or report to Architect such notice or report shall also be made to Owner.
- § 21.2 Bonds. Contractor shall furnish unconditional performance and payment bonds in an amount equal to 100% of the Contract Sum of each Work Order covering faithful performance of

the Agreement and payment of all obligations arising thereunder. The cost of the bonds is included in the Contract Sum. The bonds shall comply with the requirements of Montana Statutes; however, "Conditional" Payment Bonds shall **not** be acceptable. Proper Power of Attorney shall accompany said bonds. Said bonds shall be delivered to the Owner. Contractor shall record the bonds in the public records of the County where the Project is located. No Work shall commence on the Project until the Notice of Commencement and copies of the Payment Bonds are recorded and certified copies thereof are posted at the Project site.

- § 21.3 There are no intended or unintended third-party beneficiaries of this Agreement, and no parties other than the Owner and the Contractor shall have the right to enforce this Agreement.
- § 21.4 Discrimination Prohibited. The Contractor and the Contractor's Subcontractors shall not discriminate against any employee or applicant for employment because of race, religion, color, sex, or national origin. The Contractor shall take affirmative action to ensure that qualified applicants are employed if work is available, and that employees are treated during employment without regard to their race, religion, color, sex, or national origin. Contractor agrees to post in places available to all employees and applicants for employment, notices setting forth the policies of nondiscrimination.
- § 21.5 Conflicts. This Master Agreement shall have precedence in the event of any conflicts between this Agreement and any of the Drawings, Specifications, Contractor's Proposal or Quotation Form, attachments, or other documents incorporated by reference to this Master Agreement.
- § 21.6 Use of Contingency. If the Contract Sum breakdown for a Work Order includes a line item for contingency, such amount shall be for the Contractor's use to cover those costs considered reimbursable as the Cost of the Work but not included in a Change Order, with all unspent contingency sums accruing to the benefit of the Owner. Examples of acceptable use of contingency include: scope gaps, omitted items, trade contractors which fail to sign acceptable subcontracts, and unforeseen field conditions for which a change order is not authorized hereunder. Contingency funds shall not be used for expenses related to correcting defective work, legal expenses, or subcontractor defaults. At Final Completion, the Contract Sum shall be reduced by Change Order by the amount of the unspent contingency sum. Expenditure of contingency funds by the Contractor shall be done only with written approval of the Owner.
- § 21.7 Contractor agrees to remove from the Project any employee, Subcontractor, or Subcontractor employee that commits any breach of the Contract Documents or any breach of the Owner's written rules and regulations regarding jobsite conduct.
- § 21.8 Contractor shall require all construction personnel to maintain a neat general appearance at all times. Shirts, trousers, and proper shoes are required apparel. The display of vulgar words, signs, or figures is prohibited. Sandals and flip-flops are prohibited on the Project site.
- § 21.9 Construction personnel shall not use profanity, illegal drugs, or alcohol on the Project site.

- § 21.10 The use of radios, tape players, cd players, boom boxes, sound producing devices, and the like are prohibited on the Project site.
- § 21.11 Smoking and all tobacco products are prohibited on the Project site, and prohibited anywhere on Owner's property. Tobacco is defined as tobacco products including, but not limited to, cigars, cigarettes, e-cigarettes, pipes, chewing tobacco and snuff. Failure to abide by this policy may result in civil penalties levied under Chapter 386, Florida Statutes and/or contract enforcement remedies.
- § 21.12 Sales Tax Savings Program. The Owner is a not-for-profit, tax-exempt organization. The Contractor shall work with and assist the Owner to prevent payment of taxes not due.

Notwithstanding anything herein to the contrary, because Owner is exempt from sales tax and wishes to generate sales tax savings for the Project. Owner reserves the right to make direct purchases of various construction materials and equipment included in the Work. Before purchasing materials or equipment for the Project or a combination of like items that exceed \$5,000 in cost, Contractor shall identify in writing to Owner all materials and equipment or a combination of like items to be included in the Work that will generate sales tax savings if purchased direct by Owner. Contractor shall comply with the written procedures for Owner Purchased materials provided by Owner to Contractor and shall incorporate a similar provision to this provision in all of its subcontracts requiring the Subcontractors' compliance with said procedures. Owner shall execute direct purchase orders with vendors for such Owner Purchased materials. The purchase orders shall contain Owner's Consumer Certificate of Exemption number. Owner shall assume all risk of loss and theft for the Owner Purchased Materials upon their delivery to the Project site and until they are incorporated into the real property. Vendors shall invoice Owner directly for the Owner Purchased Materials. Owner shall pay vendors directly for the Owner Purchased Materials. Contractor represents and warrants that it will use its best efforts to cooperate with Owner in implementing this sales tax savings program in order maximize cost savings for the Project. Upon the execution of a purchase order, Owner and Contractor shall execute a Change Order decreasing the Contract Sum by the total cost, including the saved sales tax for the matelials or equipment purchased directly by Owner under said purchase order. With respect to all direct purchases by Owner, Contractor shall remain responsible for coordinating, installing, inspecting, storing, safekeeping, handling, warranting, installation, and quality control for all direct purchases. Notwithstanding anything herein to the contrary, Contractor expressly acknowledges and agrees that any materials or equipment directly purchased by Owner pursuant to this Paragraph shall be included within and covered to the same extent as all other warranties and performance guarantees provided by Contractor pursuant to the terms of the Contract Documents, including bonds. Owner assigns to Contractor any and all warranties and rights Owner may have from any manufacturer or supplier of any such direct purchases by Owner. The Parties understand that the above Owner direct purchase procedures are generally not available to purchase fabricated materials from a vendor that, in addition to fabricating them, also installs those same fabricated materials into the Project.

The Owner agrees to indemnify and hold harmless the Contractor, its Subcontractors, vendors and suppliers from and against the payment of or liability for any sales or use taxes arising out of or resulting from the Owner purchasing materials for the Project under the Owner's sales tax

exempt status, and for which the Owner or the Contractor, Subcontractors, vendors or suppliers should become liable, but only upon the condition that Contractor has properly complied with the Owner's written sales tax program procedures.

- § 21.13 Contractor shall permit Owner to have reasonable access to Contractor's records, account books, bills, invoices, payrolls, daily logs, and other records related to the Project. Contractor shall preserve such records for five (5) years after Substantial Completion of the Project.
- § 21.14 The Contractor shall ascertain, coordinate, and minimize interruptions to Owner's ongoing operations that exist on or near the Project sites. To avoid impacts to Owner's operations, Contractor has anticipated the need to use non-standard work hours and has included same in the Contract Sum. No claims for extras or requests for Change Orders based upon premium or overtime work due to non-standard work hours as defined by this Section shall be submitted to Owner.
- § 21.15 Sovereign Immunity. Owner's limits of liability are set forth in the Montana Statutes, and nothing herein shall be construed to extend the liabilities of Owner beyond that provided in the Montana Statutes. Nothing herein is intended as a waiver of Owner's sovereign immunity under the Montana Statutes. Nothing hereby shall inure to the benefit of any third party for any purpose, including but not limited to anything which might allow claims otherwise barred by sovereign immunity or operation of law. Furthermore, all of Owner's obligations under this Agreement are limited to the payment of no more than the per person amount limitation and the aggregate contained in the Montana Statutes, even if the sovereign immunity limitations of that statute are not otherwise applicable to the matters as set forth herein.

In no event shall Owner be liable to Contractor for indirect, special, or consequential damages, including, but not limited to, loss of revenue, loss of profit, cost of capital, or loss of opportunity regardless of whether such liability arises out of contract, tort (including negligence), strict liability, or otherwise. Owner shall not assume any liability for the acts, omissions, or negligence of Contractor its agents, servants, employees, or subcontractors. In all instances, Contractor shall be responsible for any injury or property damage resulting from any activities conducted by Contractor.

- § 21.16 Contractor and the Contractor's Subcontractors shall, in all solicitations or advertisements for employees placed by them or on their behalf, state that all qualified applicants will receive consideration for employment without regard to race, religion, color, sex or national origin.
- § 21.17 Prohibition against Contingent Fees. Contractor warrants that it has not employed or retained any company or person, other than a bona fide employee working solely for the Contractor to solicit or secure this Agreement and that it has not paid or agreed to pay any person, company, corporation, individual, or firm, other than a bona fide employee working solely for Contractor any fee, commission, percentage, gift or other consideration contingent upon or resulting from the award or making of this Agreement.

§ 21.18 Contractor shall be responsible for maintaining in good condition all cultivated grass plots, trees, shrubs, and landscaping on the Project site. If damaged, Contractor shall restore same to its original condition after completion of the Work.

#### § 21.19 Public Records.

- a. To the extent Contractor is acting on behalf of Owner, Contractor shall:
- i. Keep and maintain public records required by Owner to perform the services under this Agreement.
- ii. Upon request from Owner's custodian of public records, provide Owner with a copy of the requested records or allow the records to be inspected or copied within a reasonable time at a cost that does not exceed the costs provided by law.
- iii. Ensure that public records that are exempt or confidential and exempt from public records disclosure requirements are not disclosed except as authorized by law for the duration of the Agreement term and following completion of the Agreement if the Contractor does not transfer the records to Owner.
- iv. Upon completion of the Agreement, transfer, at no cost, to Owner all public records in possession of Contractor or keep and maintain public records required by Owner to perform the service. If the Contractor transfers all public records to Owner upon completion of the Agreement, the Contractor shall destroy any duplicate public records that are exempt or confidential and exempt from public records disclosure requirements. If the Contractor keeps and maintains public records upon completion of the Agreement, the Contractor shall meet all applicable requirements for retaining public records. All records stored electronically must be provided to Owner, upon request from Owner's custodian of public records, in a format that is compatible with the information technology systems of Owner.
- b. If the Contractor fails to provide the public records to Owner within a reasonable time the Contractor may be subject to penalties under Montana law. Further, Owner may exercise any remedies at law or in equity, including, without limitation, the right to (i) impose sanctions and assess financial consequences, (ii) withhold and/or reduce payment, and (iii) terminate this Agreement in accordance with the terms hereof.
  - Contractor shall defend, at its own cost, indemnify, and hold harmless Owner, their officers, directors, and employees from and against all claims, damages, losses, and expenses, (including but not limited to fees and charges of attorneys or other professionals and court and arbitration or other dispute resolution costs) arising out of or resulting from Contractor's failure to comply with the terms of this Section.
- § 21.20 Contractor shall not use any funds received pursuant to this Agreement for lobbying the Montana Legislature, the judicial branch, or any state agency.
- § 21.21 Contractor represents that it is not on the State's discriminatory vender list and that for services related to this Agreement, Contractor shall not transact business with any entity that has been placed on the State's discriminatory vendor list.
- § 21.22 The obligations of Owner under this Agreement are subject to availability of funds

lawfully appropriated for its purpose by the Owner's Board of Trustees, or other specified funding source for this Agreement.

- § 21.23 Contractor shall coordinate and conduct with its subcontractors and Owner a Project Warranty Inspection at the Project site on a mutually convenient date within the 14 day period before the expiration of the one (1) year warranty period.
- § 21.24 Contractor shall provide the builder's risk insurance for all Work Orders.
- § 21.25 Contractor shall provide a harassment-free workplace, with any allegation of harassment given priority attention and action by management. Contractor shall insert a similar provision in accordance with this Section, in all subcontracts for all Projects.
- § 21.26 Contractor shall maintain a redlined set of drawings at the job site to show any deviations made from the Contract drawings, and which reflect the "As-Built" conditions. These drawings shall clearly identify all dimensions established in the field and all deviations to the drawings, as approved by the Owner representative. All red-line drawings shall be turned-over and provided to Owner within two (2) weeks of Final Completion.
- § 21.27 PROGRESS REPORTING. Contractor shall provide to Owner with each pay request, a written progress report containing a narrative description of the Work completed, whether the Work is proceeding according to the Project schedule, a description of any executed Change Orders, problems encountered, problem resolutions, schedule of the Work to be performed during the next reporting period.
- § 21.28 Non-Exclusivity. Contractor agrees that this Master Agreement is not exclusive. This Master Agreement does not guarantee Contractor any right to enter into with Owner any Task Order and does not guarantee any work, revenue or level of revenue will be provided to Contractor. Only if Owner and Contractor execute a Task Order and Contractor properly performs the Work will Contractor be entitled to any compensation. Contractor does not have exclusivity to perform construction services for Owner. Owner has and may enter into other continuing services contracts and other contracts with other construction firms to provide construction services to Owner.

This Master Agreement entered into as of the day and year first written above.

OWNER (Signature)	CONTRACTOR (Signature)				
(Printed name and title)	(Printed name and title)				

#### **EXHIBIT A**

# DETERMINATION OF THE COST OF THE WORK FOR WORK ORDERS BASED UPON COST OF THE WORK, CHANGE ORDER WORK, AND CONSTRUCTION CHANGE DIRECTIVE WORK

#### **COST OF THE WORK**

#### § 1.1 Costs to Be Reimbursed

- § 1.1.1 The term Cost of the Work shall mean the actual costs necessarily incurred by the Contractor in the proper performance of the Work. Such costs shall be at rates not higher than the standard paid at the place of the Project except with prior consent of the Owner. The Cost of the Work shall include only the items set forth in Sections 1.1 through 1.7.
- § 1.1.2 Where any cost is subject to the Owner's prior approval, the Contractor shall obtain this approval prior to incurring the cost. The parties shall endeavor to identify any such costs prior to executing Guaranteed Maximum Price Amendment.

#### § 1.2 Labor Costs

- § 1.2.1 Wages of construction workers directly employed by the Contractor to perform the construction of the Work at the site or, with the Owner's prior approval, at off-site workshops.
- § 1.2.2 Wages or salaries of the Contractor's supervisory and administrative personnel when stationed at the site with the Owner's prior approval.

None.

- § 1.2.3 Wages and salaries of the Contractor's supervisory or administrative personnel (including Contractor's Project Manager) engaged at factories, workshops or on the road, in expediting the production or transportation of materials or equipment required for the Work, but only for that portion of their time required for the Work. Prior to commencing the Work, the Contractor shall submit to the Owner for approval, a list of supervisory and field office personnel, their duties on the Work and their respective pay rates, and anticipated hours for the Project that will be assigned as a Cost of the Work. Time and wages beyond 40 hours per week for salaried personnel will not be reimbursed.
- § 1.2.4 Costs paid or incurred by the Contractor for taxes, insurance, contributions, assessments and benefits required by law or collective bargaining agreements and, for personnel not covered by such agreements, customary benefits such as sick leave, medical and health benefits, holidays, vacations and pensions, provided such costs are based on wages and salaries included in the Cost of the Work under Sections 1.2.1 through 1.2.3.
- § 1.2.5 Relocation and temporary out of town living costs of personnel, but only if approved by Owner in writing.

#### § 1.3 Subcontract Costs

Payments made by the Contractor to Subcontractors in accordance with the requirements of the subcontracts.

#### § 1.4 Costs of Materials and Equipment Incorporated in the Completed Construction

- § 1.4.1 Costs, including transportation and storage, of materials and equipment incorporated or to be incorporated in the completed construction.
- § 1.4.2 Costs of materials described in the preceding Section 1.4.1 in excess of those actually installed to allow for reasonable waste and spoilage. Unused excess materials, if any, shall become the Owner's property at the completion of the Work or, at the Owner's option, shall be sold by the Contractor. Any amounts realized from such sales shall be credited to the Owner as a deduction from the Cost of the Work.

#### § 1.5 Costs of Other Materials and Equipment, Temporary Facilities and Related Items

- § 1.5.1 Costs of transportation, storage, installation, maintenance, dismantling and removal of materials, supplies, temporary facilities, machinery, equipment and hand tools not customarily owned by construction workers that are provided by the Contractor at the site and fully consumed in the performance of the Work. Costs of materials, supplies, temporary facilities, machinery, equipment and tools that are not fully consumed shall be based on the cost or value of the item at the time it is first used on the Project site less the value of the item when it is no longer used at the Project site. Costs for items not fully consumed by the Contractor shall mean fair market value.
- § 1.5.2 Rental charges for temporary facilities, machinery, equipment and hand tools not customarily owned by construction workers that are provided by the Contractor at the site and costs of transportation, installation, minor repairs, dismantling and removal. Rates of contractor-owned equipment and quantities of equipment shall be subject to the Owner's prior approval. The total rental cost of any piece of equipment for the duration of the Project that the Contractor, any subcontractor, or any related party, person, or entity owns shall not exceed 75% of the fair market value of that equipment at the commencement of the rental period. Where a rental agreement contains an option to purchase and this option is exercised, the equity accrued shall be credited to the Owner against the total rental cost of that equipment on the Work. Prior to beginning the Work, the Contractor shall submit a list to the Program Manager and Owner of rental equipment owned by the Contractor or a related party of the Contractor indicating the fair market value at commencement of the Work and the proposed rental rates. The Contractor shall attach to each monthly Application for Payment an itemized list of rental equipment owned by the Contractor or a related party and the applicable rates for such equipment used on the Project during that period.
- § 1.5.3 Costs of removal of debris from the site of the Work and its proper and legal disposal.
- § 1.5.4 Costs of document reproductions, facsimile transmissions and long-distance telephone calls, postage and parcel delivery charges, telephone service at the site and reasonable petty cash expenses of the site office. There shall be no markup on these expenses.

- § 1.5.5 Costs of materials and equipment suitably stored off the site at a mutually acceptable location, subject to the Owner's prior approval.
- § 1.5.6 Job site vehicle shall be limited to one per site or as agreed to in writing by the Owner. Rental rates for the job site vehicle shall include gas, maintenance, repairs, and taxes and licenses and shall not be billed separately. This single job site vehicle is in addition to the supervisors' vehicle allowances.

#### § 1.6 Miscellaneous Costs

- § 1.6.1 Premiums for that portion of insurance and bonds required by the Contract Documents that can be directly attributed to this Contract. Self-insurance for either full or partial amounts of the coverages required by the Contract Documents, with the Owner's prior approval. If the Owner optionally implements and funds directly an Owner Controlled Insurance Program (OCIP) for the Project, no insurance costs other than the insurance costs required in the OCIP contract provisions to be paid by Contractor shall be included in the Cost of the Work. However, the cost of the Performance and Payment Bonds required for the Contractor by this Agreement shall be included in the Cost of the Work.
- § 1.6.2 Sales, use or similar taxes imposed by a governmental authority that are related to the Work and for which the Contractor is liable.
- § 1.6.3 Fees and assessments for the building permit and for other permits, licenses and inspections for which the Contractor is required by the Contract Documents to pay.
- § 1.6.4 Fees of laboratories for tests required by the Contract Documents, except those to be provided by the Owner and those related to defective or nonconforming Work for which reimbursement is excluded by the Contract Documents, and which do not fall within the scope of Section 1.7.3.
- § 1.6.5 Royalties and license fees paid for the use of a particular design, process or product required by the Contract Documents.
- § 1.6.6 Small tools shall be treated as a Cost item and are defined as those tools costing five hundred dollars (\$500.00) or less. A record showing the disposition of these tools is to be on file at the Contractor's office on the project site. Ownership of small tools not consumed during construction shall remain with the Owner upon completion of the Project.
- § 1.6.7 Deposits lost for causes other than the Contractor's negligence or failure to fulfill a specific responsibility in the Contract Documents.
- § 1.6.8 In lieu or renting certain items of equipment, machinery and tools, valued at more than five hundred dollars (\$500.00) from the Contractor or other third parties, the Owner reserves the right to have those items purchased and maintained as a Cost of the Work. A record showing the disposition of these items is to be on file at the Contractor's office on the project site. Ownership of the items not consumed during construction shall remain with the Owner upon completion of the Project.

#### § 1.7 Other Costs and Emergencies

- § 1.7.1 Other costs incurred in the performance of the Work if, and to the extent, approved in advance in writing by the Owner.
- § 1.7.2 Costs incurred in taking action to prevent threatened damage, injury or loss in case of an emergency affecting the safety of persons and property.
- § 1.7.3 Costs of repairing or correcting damaged or nonconforming Work executed by the Contractor, Subcontractors or suppliers, provided that such damaged or nonconforming Work was not caused by negligence or failure to fulfill a specific responsibility of the Contractor and only to the extent that the cost of repair or correction is not recovered by the Contractor from insurance, sureties, Subcontractors, suppliers, or others.
- § 1.7.4 The costs described in Sections 1.1 through 1.7 shall be included in the Cost of the Work, unless such costs are excluded by the provisions of Section 1.8.

#### § 1.8 Costs Not To Be Reimbursed

- § **1.8.1** The Cost of the Work shall not include the items listed below:
  - 1. Salaries and other compensation of the Contractor's personnel stationed at the Contractor's principal office or offices other than the site office, except as specifically provided in Section 1.2;
  - **2.** Expenses of the Contractor's principal office and offices other than the site office;
  - **3.** Overhead and general expenses, except as may be expressly included in Sections 1.1 to 1.7;
  - **4.** The Contractor's capital expenses, including interest on the Contractor's capital employed for the Work;
  - **5.** Except as provided in Section 1.7.3 of this Agreement, costs due to the negligence or failure of the Contractor, Subcontractors and suppliers or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable to fulfill a specific responsibility of the Contract;
  - **6.** Any cost not specifically and expressly described in Sections 1.1 to 1.7;
  - 7. Costs, other than costs included in Change Orders approved by the Owner, that would cause the Guaranteed Maximum Price to be exceeded; and
  - **8.** Costs for services incurred during the Preconstruction Phase, unless provided for in a separate Change Order.
  - **9.** Payments to Contractor's employees over and above their regular pay (bonuses, incentive pay, profit sharing, severance pay, etc.), including salary increases not identified in the Contractor's bid proposal.
  - 10. Technology, Data Processing, Project Specific Web Sites, or Project Management System Cost incurred by the Contractor in preparing the Project Schedule, Payroll, Accounting, Project Cost Reports or Project Status Reports and any other reports necessary to the progress of the Work.
  - 11. Any fees paid to Contractor organizations (AGC, ABC, etc.).
  - 12. Contractor's business license.
  - **13.** Recruitment or training costs of personnel.

- 14. Overtime expense of any salaried personnel.
- **15.** Except as provided in Section 1.7, any cost not specifically and expressly described in Section 1.
- **16.** Bond costs for Contractor's suppliers and subcontractors.
- **17.** Legal fees, attorney's fees, litigation costs, mediation costs, dispute resolution costs, settlements, and arbitration costs.
- 18. Travel expenses.
- 19. Relocation and temporary out of town living costs of personnel.
- **20.** Audit or audit response expenses.

#### § 1.9 Discounts, Rebates and Refunds

- § 1.9.1 Cash discounts obtained on payments made by the Contractor shall accrue to the Owner. Trade discounts, commissions, volume discounts, rebates, refunds and amounts received from sales of surplus materials and equipment shall accrue to the Owner, and the Contractor shall make provisions so that they can be obtained.
- § 1.9.2 Amounts that accrue to the Owner in accordance with the provisions of Section 1.9.1 shall be credited to the Owner as a deduction from the Cost of the Work.

# EXHIBIT "B" SCOPE OF WORK AND SERVICES

Contractor shall provide and or perform and be fully responsible for the various professional construction Work and construction management services required for each assigned project either by self-performance or through subcontractors. The various professional construction management services, depending on the scope, composition and complexity of each project may include, but not necessarily be limited to:

Construction, repair, remodeling, renovation, rehabilitation, alteration, upgrade, maintenance services, site work and new construction related to buildings and facilities located throughout City of Sidney, Montana generally with limited design requirements and valued generally less than \$1,000,000, although on occasion could be higher. Because any Work under this Master Agreement will be on an as needed basis, exact project specifications are generally not available. The Work shall be performed in accordance with industry standards and in compliance with federal, state, and local published rules, regulations, standards and/or statutes.

### EXHIBIT C - WORK ORDER

for use with Master Agreement Between Owner and Contractor

WORK ORDER number	made as of the	day of	, 20
<b>BETWEEN</b> the Owner:			
City of Sidney 115 2 <sup>nd</sup> Street SE Sidney, Montana, 59270 (406) 433-2809			
and the Contractor:			
<b>B &amp; B Builders Inc.</b> 108 2 <sup>nd</sup> St NE Sidney, Montana, 59270 (406) 482-4401			
License no		PROJECT:	
The Architect for the Project: (Name, legal status, address, and	nd other information)		
THE CONTRACT This Work Order, together w Master Agreement between Ow form the Contract.			
The Owner and Contractor agree	e as follows.		

#### TABLE OF ARTICLES

- 1. THE WORK OF THIS WORK ORDER
- 2. DATE OF COMMENCEMENT AND SUBSTANTIAL COMPLETION
- 3. CONTRACT SUM
- 4. PAYMENTS
- 5. INSURANCE AND BONDS
- 6. PARTY REPRESENTATIVES
- 7. ENUMERATION OF CONTRACT DOCUMENTS

#### ARTICLE 1 THE WORK OF THIS WORK ORDER

The Contractor shall execute and pay for the Work described in the Contract Documents enumerated in Article 7 of this Work Order, and any modifications issued after execution of this Work Order, except as specifically indicated in the Contract Documents to be the responsibility of others.

#### ARTICLE 2 DATE OF COMMENCEMENT AND SUBSTANTIAL COMPLETION

§ 2.1 The date of commencement of the Work shall be:

(Check one of the following boxes.)

[]	The date of this Work Order.
[]	A date set forth in a notice to proceed issued by the Owner.
[]	Established as follows: (Insert a date or a means to determine the date of commencement of the Work.)
	The Date of Commencement of the Work for the Project shall be the date that the

If a date of commencement of the Work is not selected, then the date of commencement shall be the date of this Work Order.

§ 2.2 The Contract Time shall be measured from the date of commencement.

#### § 2.3 Substantial Completion

§ 2.3.1 Subject to adjustments of the Contract Time as provided in the Contract Documents, the Contractor shall achieve Substantial Completion of the entire Work:

(Check the appropriate box and complete the necessary information.)

Building Permit is issued for the Work.

[]	Not later than (_) calendar days from the date of commencement of the Work.
[]	By the following date:
portions of the	ct to adjustments of the Contract Time as provided in the Contract Documents, if e Work are to be completed prior to Substantial Completion of the entire Work, the all achieve Substantial Completion of such portions by the following dates:
Portio	on of Work Substantial Completion Date
	Contractor fails to achieve Substantial Completion as provided in this Section 2.3, nages, if any, shall be assessed as set forth in Section 3.6.
§ <b>3.1</b> The Ow performance of	CONTRACT SUM  ner shall pay the Contractor the Contract Sum in current funds for the Contractor's of the Contract. The Contract Sum shall be one of the following:   propriate box.)
[]	Stipulated Sum, in accordance with Section 3.2 below
[]	Cost of the Work plus the Contractor's Fee, in accordance with Section 3.3 below
[]	Cost of the Work plus the Contractor's Fee with a Guaranteed Maximum Price, in accordance with Section 3.4 below
[]	Other, in accordance with Section 3.5 below
(Based on the	selection above, complete Section 3.2, 3.3, 3.4 or 3.5 below.)
in the Contr	ted Sum  sipulated Sum shall be (\$), subject to additions and deductions as provided act Documents. See Exhibit "B" for Contract Sum breakdown. Contractor's to the scope of Work are attached as Exhibit "C". The Work shall be performed in

For Change Orders or Construction Change Directive Work, the amount for overhead and for profit added or deducted shall be a total of ten percent (10%) of the Cost of the Work that is added or deducted, respectively ("Contractor's Fee"). Cost of the Work is defined in Exhibit "A". Such Contractor's Fee includes all profit, overhead, general conditions costs, and insurance (but not additional bond costs, if applicable), except if a Change Order extends the Date of Substantial Completion, then Contractor's reasonable extended daily general conditions costs shall be added to the cost of the Change Order per §12.5 of the Master Agreement. Likewise, subcontractors shall not add general conditions costs, except that if the change includes an increase in the Contract Time, then extended daily general conditions costs may be added in accordance with §12.5 of the Master Agreement. However, fee shall not be reduced for deductive Change Orders arising as a result of the Sales Tax Savings Program of Article 21.12 of

accordance with the schedule attached as Exhibit "D".

the Master Agreement, except fee on the amount of the sales tax saved shall be deducted.

- 1. For Change Order Work accomplished by the Contractor's own forces, acting in the role of a subcontractor, overhead and profit shall be a maximum of 10% of the Cost of the Work.
- 2. For Change Order Work accomplished by Subcontractors, Subcontractor's overhead and profit shall be a maximum of 10% of the Cost of the Work.
- 3. For each Subcontractor, for Work performed by that Subcontractor's subsubcontractor, no overhead charges, but a 5% fee for overhead and profit on the amount due the sub-subcontractor shall be paid the Subcontractor.
- 4. For each Sub-subcontractor, for Work performed by that Sub-subcontractor, overhead and profit shall be a maximum of 10% of the Cost of the Work.
- 5. No further tiering of sub-subcontractors will be allowed mark up for fees, overhead, or profit.

# § 3.2.2 The Stipulated Sum is based upon the following alternates, if any, which are described in the Contract Documents and are hereby accepted by the Owner:

(State the numbers or other identification of accepted alternates. If the Owner is permitted to accept other alternates subsequent to the execution of this Work Order, attach a schedule of such other alternates showing the change in the Stipulated Sum for each and the deadline by which the alternate must be accepted.)

Exhibit "B"

#### § **3.2.3** Unit prices, if any:

(Identify the item and state the unit price and any applicable quantity limitations.)

Item Units and Limitations Price per Unit (\$0.00) Exhibit "B"

§ **3.2.4** Allowances, if any, included in the Stipulated Sum: *(Identify each allowance.)* 

Item Price Exhibit "B"

#### § 3.3 Cost of the Work plus Contractor's Fee

- § **3.3.1** The Cost of the Work is as defined in Exhibit A, Determination of the Cost of the Work.
- § **3.3.1.1** The following costs are subject to the Owner's prior approval:

#### § **3.3.2** The Contractor's Fee:

(State a lump sum, percentage of Cost of the Work or other provision for determining the Contractor's Fee and the method of adjustment to the fee for changes in the Work.)

8	3.	4	Cost of	f the	Work	plus	Contractor's	Fee	with a	Guaranteed	Maximum	Price

- § **3.4.1** The Cost of the Work is as defined in Exhibit A, Determination of the Cost of the Work.
- § **3.4.1.1** The following costs are subject to the Owner's prior approval:

#### § **3.4.2** The Contractor's Fee:

(State a lump sum, percentage of Cost of the Work or other provision for determining the Contractor's Fee and the method of adjustment to the fee for changes in the Work.)

#### § 3.4.3 Guaranteed Maximum Price

§ **3.4.3.2** The Guaranteed Maximum Price is based on the following alternates, if any, which are described in the Contract Documents and are hereby accepted by the Owner:

(State the numbers or other identification of accepted alternates. If the bidding or proposal documents permit the Owner to accept other alternates subsequent to the execution of this Work Order, attach a schedule of such other alternates showing the amount for each and the date when that amount expires.)

#### § **3.4.3.3** Unit Prices, if any:

(*Identify the item and state the unit price and any applicable quantity limitations.*)

Item Units and Limitations Price Per Unit (\$0.00)

§ **3.4.3.4** Allowances, if any, included in the Guaranteed Maximum Price: *(Identify each allowance.)* 

**Item** Price

§ 3.4.3.5 Assumptions, if any, on which the Guaranteed Maximum Price is based:

- § 3.4.3.6 To the extent that the Contract Documents are anticipated to require further development, the Guaranteed Maximum Price includes the costs attributable to such further development consistent with the Contract Documents and reasonably inferable therefrom. Such further development does not include changes in scope, systems, kinds and quality of materials, finishes or equipment, all of which, if required, shall be incorporated by Change Order.
- § 3.4.3.7 The Owner shall authorize preparation of revisions to the Contract Documents that incorporate the agreed-upon assumptions contained in Section 3.4.3.5. The Owner shall promptly furnish such revised Contract Documents to the Contractor. The Contractor shall notify the Owner and Architect of any inconsistencies between the agreed-upon assumptions contained in Section 3.4.3.5 and the revised Contract Documents.

#### **§ 3.5 Other**

§ **3.5.1** The Contract Sum shall be determined in accordance with the following: (*Insert a description of how the Contract Sum will be determined.*)

§ **3.6** Liquidated damages, if any: (*Insert terms and conditions for liquidated damages, if any.*)

**Liquidated Damages.** If the Contractor fails to substantially complete the Work before the agreed upon Date of Substantial Completion set forth in this Agreement (including any extension granted by the Owner in writing), then the Contractor agrees to pay the Owner, as liquidated damages, a sum equal to the (Contract Sum\* 10%)/365 per calendar day until Substantial Completion of the Work is actually achieved. Such amount is agreed upon as a reasonable measure of the damages that the Owner will sustain from the Contractor's failure to timely complete the Work. Owner and the Contractor recognize the delays, expense and difficulties involved in providing in an arbitration or judicial proceeding the actual loss suffered by the Owner if the Work is not completed on time and accordingly, instead of requiring such proofs, they agree upon the above-stated amount as liquidated damages for delay (but not as a penalty). The foregoing liquidated damages shall be in lieu of all other monetary remedies that the Owner shall have in the event of the Contractor's delay in completing the Project.

#### **ARTICLE 4 PAYMENTS**

§ 4.1 Payments shall be in accordance with Article 3 of the Master Agreement, except as indicated below:

(Indicate all payment terms that differ from those set forth in the Master Agreement, such as period covered by each Application for Payment or date upon which each Application/or Payment is due.)

See Master Agreement.

§ 4.2 Retainage will be withheld in accordance with Article 3 of the Master Agreement, except as indicated below: (Indicate all retainage terms that differ from those set forth in the Master

Agreement, such as retainage amount, items not subject to retainage, terms for reduction, or limitation of retainage.)

See Master Agreement.

#### ARTICLE 5 INSURANCE AND BONDS

§ 5.1 Insurance and Bonds shall be in accordance with Article 15 of the Master Agreement.

#### ARTICLE 6 PARTY REPRESENTATIVES

§ **6.1** The Owner identifies the following representative in accordance with Section 1.4.1 of the Master Agreement:

Jeff Hintz (406) 433-2809

§ **6.2** The Contractor identifies the following representative in accordance with Section 1.5.1 of the Master Agreement:

(List name, address, and other information.)

#### ARTICLE 7 ENUMERATION OF CONTRACT DOCUMENTS

- § 7.1 The Contract Documents are defined in Section 5.2 of the Master Agreement and, except for Modifications issued after execution of this Work Order, are enumerated in the sections below.
- § 7.1.1 This Work Order
- § 7.1.2 The Master Agreement
- § 7.1.3 The Supplementary and other Conditions of the Contract:

**Document** Title Date Pages See Exhibit E, if any.

§ 7.1.4 The Specifications:

(Either list the Specifications here or refer to an exhibit attached to this Work Order.)

Section Title Date Pages
See Exhibit E

§ 7.1.5 The Drawings: (Either list the Drawings here	or refer to an exhibit	t attached to this Work Order.)	
<b>Number</b> See Exhibit E	Title	Date	
§ 7.1.6 The Addenda, if any:			
<b>Number</b> See Exhibit E	Date	Pages	
Portions of Addenda relating unless the bidding requiremen		nents are not part of the Contract this Article 5.	Documents
§ 7.1.7 Additional documents, (List here any additional docu		of the Contract Documents: led to form part of the Contract Do	ocuments.)
This Work Order entered into	as of the day and yea	ır first written above.	
OWNER (Signature)		CONTRACTOR (Signature)	
(Printed name and title)		(Printed name and title)	

## Ordinance No. 601

# AN ORDINANCE OF THE CITY COUNCIL OF THE CITY OF SIDNEY, MONTANA, AMENDING TITLE 9, CHAPTER 7, REMOVING OF SNOW AND/OR ICE, OF THE CITY CODE OF THE CITY OF SIDNEY, MONTANA.

WHEREAS, the City Council of the City of Sidney desires to amend Title 9, Chapter 7, Removing of Snow and/or Ice, to update the snow removal code.

NOW, THEREFORE, BE IT ORDAINED AS FOLLOWS by the City Council of the City of Sidney that Chapter 7 of Title 9 of the City Code of the City of Sidney is hereby amended as follows:

## **CHAPTER 7**

## REMOVAL OF SNOW AND/OR ICE

#### **SECTION:**

## 9-7-1: Duty To Remove Snow and Ice from Sidewalks:

- (a) It shall be the duty of the owner or tenant of any privately held real property in the following zoning districts the city of Sidney to remove snow, ice and slush from sidewalks abutting said real property within forty-eight (48) hours after a snowfall: R-1, R-2, R-3, C-1, C-2, C-3.
- (b) It shall be the duty of the owner or tenant of any privately held real property in any other zoning district to remove snow, ice and slush from sidewalks abutting said real property within twenty-four (24) hours after a snowfall.
- (c) It shall be the duty of the owner of a lot or parcel of land where the sidewalk runs to an alley or other public wat to clear snow and ice from said surface for one-half of the distance across the alley or public way or to the gutter line of the street.

## 9-7-2: Unlawful Dumping of Snow onto Public Rights of Way:

(a) It shall be unlawful and it is hereby declared a nuisance for any private owner or tenant of real property within the City to dump or deposit snow, ice and slush from said real property onto public streets, avenues, alleys or rights of ways of the City of Sidney; provided, that owners and tenants of real property in the "central business district" of the City of Sidney where no landscaped boulevard exists adjacent to said real property may deposit snow, slush and ice from sidewalks adjacent to their property onto public streets or avenues of the City of Sidney. For the purposes of this section, the "central business district" is defined as: [INSERT description]

(b) Persons cleaning or causing to be cleaned any parking lot, service station, yard or court shall remove snow to a suitable disposal point not on any public rights of way, at his/her own expense.

## 9-7-3: Parking of Motorized vehicles During Removal of Snow and Ice.

Snow emergency routes. The following streets in the City of Sidney are declared snow emergency routes as follows:

(a) Snow emergency routes are those streets normally traversing the entire City or a major part of the City of Sidney and are essential to the rapid movement of emergency vehicles and normally carry the heaviest traffic volumes. The following Streets or portions of streets in the City are hereby designated as snow emergency routes:

East Main Street, from 10th Avenue to Central Avenue. West Main Street, from Central Avenue to 11th Avenue. 5th Street S.E., from 9th Avenue S.E. to Central Avenue. 5th Street S.W. from Central Avenue to 14th Avenue S.W. 14th Avenue S.W., from 5th Street S.W. to West Holly Street. 12th Avenue S.W., from 5th Street S.W. to West Holly Street. 4th Avenue S.E., from 14th Street S.E. to East Main Street. 4th Avenue N.E., from East Main Street to East Holly Street. 12th Avenue N.W., from West Holly Street to 4th Street N.W. 9th Avenue S.W., from 5th Street S.W. to 14th Street S.W. 22nd Avenue N.W., from West Holly to Bitterroot Drive.

The City of Sidney, City Council by resolution may designate and amend streets or portions of streets which shall constitute snow emergency routes within the City.

(b) Snow emergency defined. When winter weather conditions exist that make it necessary for motorized vehicle travel to be expedited such that parking on snow emergency routes be prohibited for snow plowing and/or hauling, an emergency conditions exists. Whenever the Director of Public Works or the City Street Superintendent determines that three inches (3") or more of snow, sleet, or freezing rain has accumulated on the designated snow emergency routes, and that a public service announcement to the local media and/or social media platform has been issued stating that snow route plowing will begin, a parking prohibition on said routes shall be become necessary.

A parking prohibition under this section shall remain in effect until such time as the City plows have completed plowing and/or hauling snow on a particular street designated as a snow emergency route. It is City policy to start plowing snow emergency routes within 12 hours of a snow event ending that creates the emergency situation. While the parking prohibition is in effect, no person shall park, or allow to remain parked, any vehicle or trailer on any portion of the designated snow emergency route.

Nothing herein contained will be construed to permit parking at any time or place where it is forbidden by any other ordinance provision.

- (c) Signs to mark the route. Snow emergency routes shall be posted with special signs with wording "EMERGENCY SNOW ROUTE NO PARKING IF OVER THREE INCHES."
- (d) Impounding vehicles. Any vehicle stopped on any snow emergency route in violation of the provisions of this section or on any other street of the City in violation of this section providing for the completion of plowing of snow or ice may be impounded by members of the Sidney Police Department and held in accordance with City ordinance or state law.
- (e) Evidence with respect to vehicles parked or left in violation of section. In any prosecution with regard to a vehicle parked or left in place in violation of any provision of this section, proof that the particular vehicle described in the complaint was parked or left in violation of a provision of the section, together with proof that the defendant named in the complaint was at the time the registered owner of such vehicle, shall constitute prima facie evidence that the defendant was the person who parked or left the vehicle in violation of this section.
- (f) Effective date. This section shall be in full force and effect from and after the passage, approval, and publication of the listed streets when signed as snow emergency routes in accordance with this section.
- (g) Violations Traffic citation complaint. Whenever any motor vehicle is found stopped or parked contrary to the provisions of this section, the owner of such vehicle may be charged with a violation of this section. Any person violating this section is subject to a fine of \$100 per day of violation.

#### 9-7-4: Violations.

- (a) Any person, whether owner or occupant, who refuses or fails to keep sidewalks upon the property free from snow and ice shall be guilty of an offense and shall be subject to a penalty as follows:
- (b) Upon failure of the property owner to remove snow and ice, the city may take action to remove snow and ice and assess the property, as provided in MCA7-14-4106 and further described below.
- 1. If snow and ice are not removed from sidewalks within the time and in the manner provided by the ordinances of the municipality, the snow and ice may be removed by or under the direction of the director of public works director or street commissioner and the necessary expense thereof shall be chargeable against the abutting property.

Annually, on or before May first, the public works director shall make and file in the office of the city clerk a list showing separately the amount chargeable and assessed against each

lot and tract and stating the name of the owner of each such lot or tract so far as known to the director of public works.

- 2. The city clerk shall give notice of the hearing and confirmation of the report of snow and ice removal and of the assessment therefore at the first regular June meeting of the governing body. Such notice shall notify all persons objecting to the report and assessment to appear and present their objections. The notice shall be published once each week for two consecutive weeks in the official municipal newspaper and the last publication shall not be less than eight days before the date set for the hearing.
- 3. At the meeting of the governing body in June or at such later meeting as the hearing and confirmation of such assessment may be adjourned to, the governing body shall consider and hear any objection to the snow and ice removal assessment, or to any pert thereof, and after revising or correcting the assessment, if revision or correction is necessary, it shall approve and confirm the same. The city clerk shall attach to the assessment list the city clerk's certificate that the list is correct as confirmed by the governing body and shall file the same in the city clerk's office and shall certify the assessment in the manner provided by law for the assessment of property taxes.
- 4. From the time any assessment list provided for in this chapter is approved by the governing body, the assessment, with interest and penalties thereon, shall be and remain a permanent lien upon the property upon which the assessment is levied until the assessment is paid in full, and it shall have precedence over all other liens except general taxes. Such lien shall not be divested b any judicial sale, and no mistake in the description of the property assessed nor in the name of the owner thereof shall defeat such lien if the property assessed can be identified by the description in the assessment list.

#### 9-7-5: Penalties.

- (a) Any person who interferes with or hinders the removal of said snow and ice by the city, shall be guilty of a misdemeanor.
- (b) Any person who dumps or deposits snow, ice and slush from any real property onto the public streets, avenues, alleys or rights of ways of the City of Sidney shall be guilty of a misdemeanor.

This ordinance shall be in full force and effect upon its final passage and approval.

(ALTERNATE VERBAGE FOR SECTION 9-7-3)

## 9-7-3: Parking of Motorized vehicles During Removal of Snow and Ice.

(h) Snow emergency routes. The following streets in the City of Sidney are declared as primary posted snow emergency routes as follows:

12th Avenue S.W./N.W., from 5th Street S.W. to West Holly Street.

12th Avenue N.W., From West Holly Street to 4th Street N.W.
5th Street S.W. from Central Avenue to 14th Avenue S.W.
9th Avenue S.W., from 5th Street S.W. to 14th Street S.W.
22nd Avenue N.W., from West Holly to Bitterroot Drive.
Lincoln Avenue N.W./S.W., from West Holly to Central Avenue.
East Main Street, from 10th Avenue to Central Avenue.
West Main Street, from Central Avenue to Lincoln Avenue

(i) Snow emergency routes are those streets normally traversing the entire City or a major part of the City of Sidney and are essential to the rapid movement of emergency vehicles and normally carry the heaviest traffic volumes. The following Streets or portions of streets in the City are hereby designated as secondary, non-posted snow emergency routes.

5th Street S.E., from 9th Avenue S.E. to Central Avenue. West Main Street, from Lincoln Avenue to 11th Avenue. 14th Avenue S.W., from 5th Street S.W. to West Holly Street. 4th Avenue S.E., from 14th Street S.E. to East Main Street. 4th Avenue N.E., from East Main Street to East Holly Street. 14th Street S.W., from 16th Avenue to Central Avenue.

The City of Sidney, City Council by resolution may designate and amend streets or portions of streets which shall constitute snow emergency routes within the City.

1 <sup>st</sup> reading:								
2 <sup>nd</sup> reading:								
Commissioner	moved	the	adoption	of tl	ne for	egoin	g C	Ordinance.
The Motion was seconded by Commissioner						·	On	roll call
vote of the Commissioners vote "AYE":								
					, a	nd th	ne	following
Commissioners vote "NAY":								Absent
and not voting:								<u></u> .

WHEREUPON, the Motion was passed	and the Ordinance declared adopted this	day of
	Mayor	
ATTEST:		
City Clerk/Treasurer		

#### Ordinance No. 603

AN ORDINANCE OF THE CITY COUNCIL OF THE CITY OF SIDNEY, MONTANA, AMENDING TITLE 11, CHAPTER 18, RETAINING WALLS/FENCING, OF THE CITY CODE OF THE CITY OF SIDNEY, MONTANA.

WHEREAS, the City Council of the City of Sidney desires to amend Title 11, Chapter 18, Retaining Walls/Fencing,

NOW, THEREFORE, BE IT ORDAINED AS FOLLOWS by the City Council of the City of Sidney that Chapter 18 of Title 11 of the City Code of the City of Sidney is hereby amended as follows:

## §11.18.28 Retaining Walls/Fencing

## 11.18.28 A Retaining Walls

Retaining walls shall not exceed the height of eight (8) feet. Fences in Residential Zones shall not exceed six (6) feet in height above natural grade. More than one (1) retaining wall in a terraced arrangement shall be permitted. Retaining walls lower than three (3) feet designed and constructed to retain earth are exempt from other standards of this Ordinance that regulate walls. Walls taller than three (3) feet are required to be permitted and must include a stamp from a licensed Montana engineer.

#### 11.18.28 B Fences by District

Fences in Residential Zones (R- Zones) shall not exceed six (6) feet in height above natural grade. The use of barbed wire or electric fences in residential zones is prohibited.

Fences in Business Zones (numbered B and C Zones) shall not exceed six (6) feet in height above natural. In the B-1 Zone fences shall not exceed eight (8) feet in height. The use of barbed wire or electric fences in Business Zones is prohibited.

Fences in Manufacturing Zones (CLM and MI) shall not exceed eight (8) feet in height above natural grade. The use of barbed wire is allowed in Manufacturing Zones. The use of electric fences is prohibited in Manufacturing Zones.

#### 11.18.28 C Additional Fencing Requirements

Any fences constructed may extend beyond the property line onto the City right-of-way, provided that such fence ends at least five feet (5') from the established curb line to allow room for sidewalk construction. In the event no curb line has been established at the time of fence construction, the director of Public Works may establish a projected curb line for the purpose of determining placement of sidewalks in the future.

Notwithstanding any other part of this Section, a chain link fence may be constructed to a height of forty-two inches (42") within ten feet (10') from the intersection of a street or alley or within twenty-five feet (25') of the main street, provided that no materials are inserted into such chain link fence that would cause a decrease or obstruction to the vision of any operator of a vehicle.

Barbed wire and electrical fences may be permitted as an accessory use for safety reasons only upon the specified approval of the Building Inspector.

No yard fence shall be constructed of barbed wire, chicken wire, electrical wire, snow fence or any other wire unless such wire fencing is specifically designed and made for yard fencing and approved by the Building Inspector. Chain link fencing materials are acceptable. No steel "T" or "U" fence posts may be used.

Any wood fence constructed from salvaged material must be painted so that all materials are of the same color within 30-days of installation.

In no case shall such fences, walls or hedges be so placed and/or maintained that they impede vision for a driver of a vehicle approaching a street, alley or driveway intersection with another vehicle-traveled area; visibility triangles at all intersections shall be maintained unobstructed by fences.

1 <sup>st</sup> reading:	
2 <sup>nd</sup> reading:	
Commissioner	moved the adoption of the foregoing Ordinance.
The Motion was seconded by Commissioner	On roll call
vote of the Commissioners vote "AYE":	
	, and the following
Commissioners vote "NAY":	Absent
and not voting:	
WHEREUPON, the Motion was passed and	the Ordinance declared adopted this day of
, 2025.	

	Mayor	
ATTEST:		
City Clerk/Treasurer		



# **Compliance Officer Report**

Properties	Noticed	Abated	Outstanding
Mowing (9-5-1 and 9-5-2)			0
Other			0
Totals		0	0 0

Vehicles	Noticed	Abated	Outstandi	Outstanding	
Trailers (10-2-15)		12	12	0	
On Street Excess of 5 Days (61-8-56)		5	5	0	
Semi's (10-2-18)				0	
Other				0	
Totals		17	17	0	

Nuisance Properties	# of Properties
On Nuisance Committee Agenda	
Approval to be noticed by Nuisance Committee for Abatement	
Dispatched for Abatement	
Totals	

# Meadows Subdivision Stormwater Improvements

Building Resilient Infrastructure and Communities (BRIC) Grant





# FEMA Cancels BRIC program

# WHAT WE KNOW

(Information provided by Montana DES)

- All future and pending BRIC applications are cancelled
- Fully obligated projects (including Meadows) are subject to review
- FEMA will review Period of Performance and project status
- No schedule extensions to period of performance will be granted
- FEMA is required to manually review all details of reimbursement (will take longer)

  Morrison
  Majorle

# Meadows BRIC Grant Details - Fully Obligated



BRIC Award: \$1,192,506

State Resiliency Dollars: \$511,074

Management Costs Award: \$85,179

Total Award: \$1,788,759

Period of Performance (3 years): From 6/20/24 To 6/19/27

Project/Grant Status:

Reimbursement Request 1
Q3 2024 - \$72k
(grant writing and preliminary design)

Submitted Oct 2024
Payment NOT received

Reimbursement Request 2
Q4 2024 - \$5k
(final design)

Submitted Jan 2025
Payment NOT received



Reimbursement Request 3
Q1 2025 - \$60k
(final design)

NOT submitted yet

# **Next Steps**

 Prepare memo for FEMA Review outlining schedule (within Period of Performance) and project status.



## Schedule Option 1

Permitting: April to October 2025

Advertise to Bid: May 2025

<u>Construction of Pond Improvements</u>: July – October 2025 (City/County)

Construction of Storm Drain: October 2025 - April 2026

Expedited schedule shows commitment to finishing project before Period of Performance ends (June 2027)

Financial risk due to unknown reimbursement schedule

## Schedule Option 2

Item b.

Permitting: April to October 2025

<u>Construction of Pond Improvements</u>: July – October 2025 (City/County)

Advertise to Bid: Fall 2025

Construction of Storm Drain: October 2026 – April 2027

Project still finished before Period of Performance ends (June 2027)

Allows time for more information to be released to have better understanding of reimbursement



## March 2025 SVFD Run Report

2025-031	#1	False Alarm	3/3/2025	Fire	City	1	hrs
2025-032	#2	Airport Standby	3/6/2025	Fire	County	1	hrs
2025-033	#3	Co2 alarm	3/7/2025	Not Paged	County	1	hrs
2025-034	#4	Fire Alarm	3/8/2025	Not Paged	city	1	hrs
2025-035	#5	co alarm	3/8/2025	Fire	City	1	hrs
2025-036	#6	Fire Alarm	3/9/2025	Fire	city	1	hrs
2025-037	#7	EMS Assist	3/11/2025	Medical	city	1	hrs
2025-038	#8	EMS Assist	3/15/2025	city	City	1	hrs
2025-039	#9	Haybale	3/16/2025	fire	county	2	hrs
2025-040	#10	Smell of Smoke	3/21/2025	Not Paged	city	1	hrs
2025-041	#11	EMS Assist	3/24/2025	Medical	City	1	hrs
2025-042	#12	EMS Assist	3/25/2025	Medical	City	1	hrs
2025-043	#13	Grass Fire assist	3/25/2025	fire	county	6	hrs
2025-044	#14	False alarm	3/27/2025	Fire	City	1	hrs
2025-045	#14	EMS Assist	3/29/2025	Medical	city	1	hrs

Item a.

## **Water/Sewer Bank Transfer**

Quarter: January 1st, 2025 to March 31th, 2025

Black Mountai	n Casl	h Balances	Bank Account Bala	nces	
Water (5210)	\$	6,651,263.72	Water/Sewer	\$	12,085,450.99
Water Impact (5211)	\$	307,891.38	Repurchase	\$	6,195,775.82
Sewer (5310)	\$	4,345,302.68	Checking	\$	250,184.76
Sewer Impact (5310)	\$	155,831.53			
·	\$	11,460,289.31			

## Transfer Amount: \$ (625,161.68)

(Positive number is transfer from Repurchase to Water/Sewer) (Negative number is transfer from Water/Sewer to Repurchase)

Completed By: Karmen John iers	Date: 4/10/25
Approved: Bream Shanks	Date: 4/16/85
Approved:	Date:

CITY OF SIDNEY
Cash Report by Fund/Account
For the Accounting Period: 3/25

Page: 1 of 1 Report ID: L160

Funds 5210-5311

04/03/25 16:43:06

Fund/Account	Beginning Balance	Received	Transfers In	Disbursed	Transfers Out	Ending Balance
5210 Water Utility		*0	<b>2</b> 9	•5	•	
101000 Cash - Operating	4,956,084.81	222,007.92	34.18	00.00	130,418.33	5,047,708.58
101235 UNRESTRICTED CASH ACCOUNT	80,736,41	00.00	00.00	0.00	00.00	80,736.41
101240 UNRESTRICTED CASH ACCOUNT	1,047,208.12	00.00	00.00	00.00	00.00	1,047,208.12
102000 Cash - Restricted	-5,437.39	00.00	00.00	00.00	00.00	-5,437.39
102200 Cash-Restricted for Bond	192,316.00	00.00	00.00	00.00	00.00	192,316.00
102230 Cash-Reserve for Rural	288,507.00	00.00	00.00	00.0	00.00	288,507.00
103000 Petty Cash	225.00	00.00	00.00	00.0	00.00	225.00
Total Fund	6,559,639.95	222,007.92	34.18		130,418.33	6,651,263.72
5211 WATER IMPACT FEES						
101000 Cash - Operating	305,891.38	2,000.00	00.00	00.0	00.00	307,891.38
5310 Sewer Utility						
101000 Cash - Operating	2,379,228.69	210,254.24	2,995.53	00.00	73,284.35	2,519,194.11
101240 UNRESTRICTED CASH ACCOUNT	1,066,236.41	00.00	00.00	00.00	00.00	1,066,236.41
102200 Cash-Restricted for Bond	797,668.00	00.00	00.00	00.00	00.00	797,668.00
102240 Cash-Replacement &	-1,162.50	00.00	00.00	00.0	00.00	-1,162.50
102250 Cash-Capital Equipment	-36,633.34	00.0	00.00	00.0	00.00	-36,633.34
Total Fund	4,205,337.26	210,254.24	2,995.53		73,284.35	4,345,302.68
5311 SEWER IMPACT FEES	152.831.53	3.000.00	00.00	00.00	0.00	155,831.53
	1100	E				
Totals	11,223,700.12	437,262:16	3,029.71	00.00	203,702.68	11,460,289.31

\*\*\* Transfers In and Transfers Out columns should match, with the following exceptions:

1) Cancelled electronic checks increase the Transfers In column. Disbursed column will be overstated by the same amount and will not balance to the Redeemed Checks List.

2) Payroll Journal Vouchers including local deductions with receipt accounting will reduce the Transfers Out column by the total amount of these checks.

Miles City, Montena \$9307/0250

>002441 8746419 0001 93518 10Z 30

CITY OF SIDNEY WATER/SEWER-CITY TREASURER 115 2ND ST SE SIDNEY MT 59270-4103

CITY OF SIDNEY Page 1 of 6 Account Number: XXXXXX5410 Item a.

**Managing Your Accounts** 

Branch Name Stockman Bank

301 West Holly Street Mailing Address Sidney MT 59270

406-433-8600 Phone Number

www.stockmanbank.com Online Access

## Summary of Accounts

**Ending Balance Account Number Account Type** \$12,085,450.99 XXXXXX5410 CITY/COUNTY/STATE



Statement Ending 03/31/2025

CITY OF SIDNEY

Page 1 of 18



Statement Ending 03/31/2025

CITY OF SIDNEY

## Summary of Accounts

**Ending Balance Account Number** \$6,195,775.82 Account Type XXXXXXX0173 CITY/COUNTY/STATE

## CITY/COUNTY/STATE - XXXXXXX0173

## **Summary of Accounts**

**Ending Balance** Account Number \$250,184.76 **Account Type** XXXXXXX0486 CITY/COUNTY/STATE

## CITY/COUNTY/STATE - XXXXXXX0486

**Account Summary** 

Page: 1 of 5

04/14/25 CITY OF SIDNEY 10:48:05 Report ID: L100 Journal Voucher Details For the Accounting Period: 3/25

					Description			Debit	Credit	User ID
Doc #	Line #	Fund Org	Account	Object	Fund Account	Type	Date	Amount	Amount	Proj
PR 250300	3/25						04/01/	25		bree
	1	1000	101000		Employer Contributions				12,650.1	.0
	2	1000	101000		Payroll Expenditure				118,320.2	13
	3	1000	410130	100	Payroll Expenditure			1,500.00		
	4	1000	410130	142	Employer Contributions			9.24		
	5	1000	410130	143	Employer Contributions			114.77		
	6	1000	410210	100	Payroll Expenditure			2,050.00		
	7	1000	410210	142	Employer Contributions			12.60		
	8	1000	410210	143	Employer Contributions			134.00		
	9	1000	410540	100	Payroll Expenditure			2,090.12		
	10	1000	410540	141	Employer Contributions			3.17		
	11	1000	410540	142	Employer Contributions			12.82		
	12	1000	410540	143	Employer Contributions			152.93		
	13	1000	410550	100	Payroll Expenditure			2,090.10		
	14	1000	410550	141	Employer Contributions			3.13		
	15	1000	410550	142	Employer Contributions			39.67		
	16	1000	410550	143	Employer Contributions			152.93		
	17	1000	420100	100	Payroll Expenditure			92,031.62		
	18	1000	420100	141	Employer Contributions			138.06		
	19	1000	420100	142	Employer Contributions			2,691.06		
	20	1000	420100	143	Employer Contributions			6,860.62		
	21	1000	420180	100	Payroll Expenditure			1,948.78		
	22	1000	420180	141	Employer Contributions			2.92		
	23	1000	420180	142	Employer Contributions			11.97		
	24	1000	420180	143	Employer Contributions			136.59		
	25	1000	420400	100	Payroll Expenditure			3,352.14		
	26	1000	420400	141	Employer Contributions			5.03		
	27	1000	420400	142	Employer Contributions			512.61		
	28	1000	420400	143	Employer Contributions			234.40		
	29	1000	420531	100	Payroll Expenditure			5,300.91		
	30	1000	420531	141	Employer Contributions			7.96		
	31	1000	420531	142	Employer Contributions			32.57		
	32	1000	420531	143	Employer Contributions			370.99		
	33	1000	460430	100	Payroll Expenditure			7,956.56		
	34	1000	460430	141	Employer Contributions			11.92		
	35	1000	460430	142	Employer Contributions			423.20		
	36	1000	460430	143	Employer Contributions			574.94		
	37	2370	101000		Employer Contributions				13,013.7	'3
	38	2370	410130	144				34.02		
	39	2370	410540	144	Employer Contributions			189.56		
	40	2370	410550	144	Employer Contributions			189.57		
	41	2370	420100	144	Employer Contributions			10,917.32		
	42	2370	420180	144	Employer Contributions			176.75		
	43	2370	420400	144	Employer Contributions			304.04		
	44	2370	420531	144	Employer Contributions			480.80		
	45	2370	460430	144				721.67		
	46	2371	101000		Employer Contributions			1 04:	23,539.4	:U
	47	2371	410210	146	Employer Contributions			1,244.18		
	48	2371	410540	146	Employer Contributions			545.82		
	49	2371	410550	146	Employer Contributions			545.75		

Page: 2 of 5

Report ID: L100

04/14/25 CITY OF SIDNEY

10:48:05 Journal Voucher Details

For the Accounting Period: 3/25

Debit. Credit User ID/ Description Line # Fund Org Account Object Doc # Fund Account Type Date Amount Amount Proj 50 2371 420100 146 Employer Contributions 14,399.09 51 2371 420180 146 Employer Contributions 975.60 2371 420400 992.27 52 146 Employer Contributions 53 2371 420531 146 Employer Contributions 1,967.84 54 2371 430200 146 Employer Contributions 2.55 55 2371 460430 146 Employer Contributions 2,866.30 2565 101000 Employer Contributions 6,549.85 56 57 2565 101000 Payroll Expenditure 14,668.54 58 2565 430200 100 Payroll Expenditure 14,668.54 59 2565 430200 141 Employer Contributions 22 01 2565 779.59 60 430200 142 Employer Contributions 61 2565 430200 143 Employer Contributions 1,075.69 62 2565 430200 Employer Contributions 1,330.41 144 63 2565 430200 Employer Contributions 3,342.15 146 64 2566 101000 Employer Contributions 2,486.79 65 2566 101000 Payroll Expenditure 5,057.81 66 2566 430251 100 Payroll Expenditure 5,057.81 67 2566 430251 141 Employer Contributions 7.59 68 2566 430251 142 Employer Contributions 268 39 69 2566 369.80 430251 143 Employer Contributions 70 2566 430251 144 Employer Contributions 458.74 71 2566 430251 146 Employer Contributions 1,382.27 72 5210 101000 Employer Contributions 10,267.57 73 5210 101000 Payroll Expenditure 21,779.41 74 5210 430500 100 Payroll Expenditure 21,779.41 75 5210 430500 141 Employer Contributions 32.67 76 5210 430500 142 Employer Contributions 881.35 77 5210 430500 143 Employer Contributions 1,599.43 78 5210 1,975.40 430500 144 Employer Contributions 79 5210 430500 146 Employer Contributions 5,778.72 5310 Employer Contributions 10.246.08 80 101000 5310 Payroll Expenditure 23,308.69 81 101000 82 5310 430600 100 Payroll Expenditure 23,308.69 83 5310 430600 141 Employer Contributions 34.96 84 5310 430600 142 Employer Contributions 773.82 5310 430600 Employer Contributions 1,715.51 85 143 5310 2,114.10 86 430600 Employer Contributions 87 5310 430600 146 Employer Contributions 5,607.69 88 5410 101000 Employer Contributions 13,200.26 89 5410 101000 Payroll Expenditure 28,184.17 90 5410 430830 100 Payroll Expenditure 28.184.17 91 5410 430830 141 Employer Contributions 42 27 92 5410 430830 142 Employer Contributions 1,425.05 2,076.17 93 5410 430830 143 Employer Contributions 94 5410 430830 144 Employer Contributions 2,556.31 95 5410 430830 146 Employer Contributions 7,100.46 96 5710 101000 Employer Contributions 2,930.49 97 5710 101000 Payroll Expenditure 6,867.79 6,867.79 98 5710 430252 100 Payroll Expenditure 5710 430252 141 Employer Contributions 10.29

Page: 3 of 5

Report ID: L100

04/14/25 CITY OF SIDNEY

10:48:05 Journal Voucher Details

For the Accounting Period: 3/25

					Description			Debit	Credit User II
Doc #	Line #	Fund Org	Account	Object	Fund Account	Type	Date	Amount	Amount Proj
	100	5710	430252	142	Employer Contributions			271.76	
	101	5710	430252	143	Employer Contributions			506.12	
	102	5710	430252	144	Employer Contributions			622.94	
	103	5710	430252	146	Employer Contributions			1,519.38	
	104	7910	101000		Direct Deposit Clearing				126,558.55
	105	7910	101000		Electronic Check				96,072.62
	106	7910	101000		Employee Checks			218,186.64	
	107	7910	101000		Employer Contributions			94,884.27	
	108	7910	201000		Check for tax/benefit plan				84,534.17
	109	7910	201000		Employee Checks				21,601.84
	110	7910	212200		Electronic Check			17,081.98	
	111	7910	212200		Employee Deduction				6,567.20
	112	7910	212200		Employer Contributions				10,514.78
	113	7910	212501		Electronic Check			32,149.78	
	114	7910	212501		Employee Deduction				16,074.89
	115	7910	212501		Employer Contributions				16,074.89
	116	7910	212502		Electronic Check			21,622.89	
	117	7910	212502		Employee Deduction				10,066.04
	118	7910	212502		Employer Contributions				11,556.85
	119	7910	212503		Electronic Check			910.50	
	120	7910	212503		Employer Contributions				321.98
	121	7910	212504		Check for tax/benefit plan			23,253.45	
	122	7910	212504		Employer Contributions				8,145.70
	123	7910	212505		Electronic Check			16,745.41	
	124	7910	212505		Employee Deduction			•	16,745.41
	125	7910	212506		Electronic Check			6,963.00	
	126	7910	212506		Employee Deduction				6,963.00
	127	7910	212510		Check for tax/benefit plan			60,551.10	•
	128	7910	212510		Electronic Check			599.06	
	129	7910	212510		Employee Deduction				12,880.09
	130	7910	212510		Employer Contributions				48,270.07
	131	7910	212515		Check for tax/benefit plan			729.62	,
	132	7910	212515		Employee Deduction				729.62
JB 28					1 1211		04/01/	25	UB
	1	5210	122000		Billing - UB		. , . ,	140,249.91	
	2	5210	313021		Billing - UB				707.34
	3	5210	343021		Billing - UB				139,542.57
	4	5310	122000		Billing - UB			151,055.50	
	5	5310	343031		Billing - UB			131,033.30	151,055.50
JB 28	43 3/25	3310	313031		21111119 02		04/01/	25	UB
20	1 1	5210	101000		Receipts - ACH   UB		01/01/	29,272.56	0.5
	2	5210	122000		Receipts - ACH   UB			25,272.50	29,272.56
	3	5310	101000		Receipts - ACH   UB			31,540.41	25,272.50
	4	5310	122000		Receipts - ACH   UB			31,340.41	31,540.41
JB 28	44 3/25	2210	122000		Receipes - Acn   UB		04/01/	25	31,540.41 UB
20	1	5210	101000		Batch Payment		01/01/	3,310.92	U.B
	2	5210	122000		Batch Payment  Batch Payment			3,310.92	3,310.92
	3							2 072 65	3,310.92
		5310	101000		Batch Payment			3,873.65	2 072 65
	4	5310	122000		Batch Payment				3,873.65

04/14/25 CITY OF SIDNEY Page: 4 of 5
10:48:05 Journal Voucher Details Report ID: L100
For the Accounting Period: 3/25

Description Debit Credit User ID/ Line # Fund Org Account Object Fund Account Doc # Type Date Amount Amount Proj UB 2845 3/25 04/01/25 UB 1 5210 101000 Batch Payment 1,597.93 2 5210 122000 Batch Payment 1,597.93 5310 101000 1,737.16 Batch Payment 5310 122000 Batch Payment 1,737.16 UB 2846 3/25 04/01/25 UB 1 5210 101000 Batch Payment 1,275.27 2 5210 122000 Batch Payment 1,275.27 101000 3 5310 Batch Payment 1,678.29 5310 122000 4 Batch Payment 1,678.29 UB 04/01/25 UB 2847 3/25 1 5210 101000 Batch Payment 5,005.42 2 5210 122000 Batch Payment 5,005.42 3 5310 101000 Batch Payment 5,653.96 4 5310 122000 Batch Payment 5,653.96 UB 2848 3/25 04/01/25 UB 1 5210 101000 Batch Payment 2,268.54 2 5210 122000 Batch Payment 2,268.54 3 5310 101000 Batch Payment 2,019.56 5310 122000 4 Batch Payment 2.019 56 2849 3/25 IJB 04/01/25 UB 1 5210 101000 Batch Payment 7,137.40 122000 2 5210 Batch Payment 7,137.40 5310 101000 Batch Payment 7,557.65 4 5310 122000 Batch Payment 7,557.65 2850 3/25 04/01/25 UB 5210 101000 3,226.39 Batch Payment 5210 122000 Batch Payment 3,226.39 5310 101000 3,608.53 Batch Payment 5310 122000 Batch Payment 3,608.53 2851 3/25 04/01/25 UB IJB 101000 Batch Payment 1 5210 1,134.72 122000 2 5210 Batch Payment 1,134.72 3 5310 101000 Batch Payment 1,487.19 4 5310 122000 Batch Payment 1,487.19 UB 2852 3/25 04/01/25 UB 1 5210 101000 Batch Payment 954.95 122000 5210 Batch Payment 954.95 101000 3 5310 Batch Payment 1,035.43 4 5310 122000 Batch Payment 1,035.43 IJB 2853 3/25 04/01/25 IJB 101000 1 5210 Batch Payment 1,933.05 5210 122000 Batch Payment 2 1,933.05 5310 101000 2,072.95 3 Batch Payment 5310 122000 4 Batch Payment 2,072.95 IJB 2854 3/25 04/01/25 UB 1 5210 101000 Batch Payment 1,350.56 2 5210 122000 Batch Payment 1,350.56 5310 101000 Batch Payment 1,123.32 5310 122000 Batch Payment 1,123.32

Page: 5 of 5

04/14/25 CITY OF SIDNEY 10:48:05 Report ID: L100 Journal Voucher Details

For the Accounting Period: 3/25

oc #	‡	Line #	Fund Org	Account.	Object		Description Fund Account	Type	Date	Debit Amount	Credit   Amount	Jser II Proj
В	2855	3/25							04/01/	25		UB
		1	5210	101000		Batch	Payment			359.75		
		2	5210	122000			Payment				359.7	5
		3	5310	101000			Payment			519.58		
		4	5310	122000		Batch	Payment				519.5	
В	2856	3/25							04/01/			UB
		1	5210	101000			Payment			1,453.62		_
		2	5210	122000			Payment				1,453.6	2
		3	5310	101000			Payment			1,420.28	1 400 0	2
	0057	4	5310	122000		Batch	Payment		04/01/	25	1,420.2	
В	2857	3/25	E010	101000		D. t. ole	D		04/01/			UB
		1	5210	101000			Payment			770.65	550	_
		2	5210	122000			Payment				770.6	5
		3	5310	101000			Payment			815.68	015 6	2
	0050	4	5310	122000		Batch	Payment		04/01/	25	815.68	
В	2858	3/25	E010	101000		D. t. d.	D		04/01/			UB
		1	5210	101000			Payment			1,085.66	1 005 6	_
		2	5210	122000			Payment			1 126 41	1,085.6	0
		3 4	5310 5310	101000			Payment			1,136.41	1 126 4	1
	2050		5310	122000		Batch	Payment		04/01/	25	1,136.4	
В	2859	3/25	5210	101000		Datab	Danmanh		04/01/	323.94		UB
		1 2		101000			Payment			323.94	202.0	4
			5210	122000			Payment			200 50	323.9	+
		3 4	5310 5310	101000 122000			Payment			389.59	389.5	n
В	2060	3/25	3310	122000		Batti	Payment		04/01/	25	309.3.	UB
)	2000	1	5210	101000		Patah	Payment		04/01/	439.72		UB
		2	5210	122000			Payment			439.72	439.7	2
		3	5310	101000			Payment			499.68	439.7.	۷.
		4	5310	122000			Payment			499.00	499.68	Q
3	2861	3/25	3310	122000		Daccii	raymene		04/01/	25	400.00	UB
,	2001	1	5210	101000		Ratch	Payment		04/01/	674.36		OD
		2	5210	122000			Payment			074.50	674.3	5
		3	5310	101000			Payment			674.92	074.5	5
		4	5310	122000			Payment			074.52	674.9	2
3	2862	3/25	3310	122000		200011	2 0/10010		04/01/	25	0,113.	UB
	2002	1	5210	101000		Batch	Payment		01/01/	2,437.03		02
		2	5210	122000			Payment			2,137.03	2,437.0	3
		3	5310	101000			Payment			2,584.02	_,	-
		4	5310	122000			Payment			_,	2,584.0	2
В	2863	3/25					<u>-</u>	TRANSFER	04/01/	25	,	UB
		1	5210	122000		Adi-UE	3 Auto Distribute			2,995.53		
		2	5210	101000			3 Auto Distribute			,	2,995.5	3
		3	5310	101000			Auto Distribute			2,995.53	·	
		4	5310	122000			Auto Distribute				2,995.5	3
3	2864	3/25							04/01/	25		UB
		1	5210	122000		Adjust	ment - UB			414.08		
		2	5210	343021			ment - UB				414.0	В
		3	5310	122000			ment - UB			38.84		
		4	5310	343031			ment - UB				38.8	4

Grand Total 1,241,938.70 1,241,938.70

## Treasurer's Report March 2025

	Fund Name	Expended YTD	% Expended	Revenued YTD	% Revenued	Difference Rev vs Exp	Cash Balance	Notes
und	General	\$2,023,499.83	51%	\$2,172,049.36	55%	\$148,549.53	\$1,461,172.90	
1000	Playgrounds & Parks	\$0.00	0%	\$1,623.22	100%	\$1,623.22	\$31,242.27	
2060	Ballparks & Ballfields	\$0.00	0%	\$1,236.59	100%	\$1,236.59	\$23,800.75	
2061	Tennis Courts	\$39,520.00	17%	\$8,111.33	10%	-\$31,408.67	\$116,599.56	
2062	Bike Path Enhancement	\$1,500.00	2%	\$4,652.89	32%	\$3,152.89	\$86,412.23	
2063	TBID	\$63,521.27	21%	\$80,480.74	27%	\$16,959.47	\$29,968.60	
2101		\$0.00	0%	\$10,860.87	62%	\$10,860.87	\$15,970.63	
2170	Airport	\$51,586.50	100%	\$26,009.57	67%	-\$25,576.93	-\$9,746.64	
2190	Comprehensive Liability	\$0.00	0%	\$2,637.59	187%	\$2,637.59	\$26,506.22	
2220	Library Levy		3%	\$5,809.91	75%	\$4,096.41	\$47,349.54	
2260	Emergency Disaster	\$1,713.50 \$825.00	3%	\$18,200.42	59%	\$17,375.42	\$17,375.10	
2350	Local Govt Study Comm		45%	\$104,342.25	65%	-\$28,243.29	\$109,196.22	
2370	PERS	\$132,585.54	53%	\$171,074.51	59%	-\$50,132.18	\$80,664.74	
2371	Group Health	\$221,206.69	0%	\$273.40	194%	\$273.40	\$2,844.57	
2372	Permissive Health Levy	\$0.00		\$2,615.26	18%	-\$7,431.75	\$32,683.02	
2390	Drug Forfeiture	\$10,047.01	40%		104%	\$17,053.26	\$316,617.50	
2399	Impact Fees	\$0.00	0%	\$17,053.26	73%	\$21,902.52	\$410,740.93	
2425	Street Lighting	\$97,047.24	50%	\$118,949.76		\$250.94	\$4,829.95	
2550	Dutch Elm Tree Removal	\$0.00	0%	\$250.94	100%	\$248,618.46	\$405,330.94	
2565	Street Maintenance	\$361,800.27	67%	\$610,418.73	131%		\$23,133.04	
2566	Snow Removal	\$99,237.13	49%	\$6,340.88	3%	-\$92,896.25	\$100,953.72	
2584	Mowing	\$2,238.53	4%	\$10,668.58	30%	\$8,430.05	\$29,628.52	
2598	MVS Park Maintenance	\$1,650.00	14%	\$3,367.30	127%	\$1,717.30	\$11,322.66	-
2810	Police Reserve Training	\$5,268.14	26%	\$939.83	6%	-\$4,328.31	\$504,985.64	
2820	Gas Apportionment Tax	\$900,469.25	81%	\$252,653.65	81%	-\$647,815.60	\$28.72	
2821	New Fuel Tax	\$0.00	0%	\$0.00	0%	\$0.00		
2861	Main Street MT Grant	\$49,548.77	99%	\$49,548.77	99%	\$0.00	\$0.00	
2869	Nuisance	\$24,043.25	48%	\$1,897.50	3%	-\$22,145.75	-\$22,145.75	
2890	Oil/Gas Severance	\$42,043.23	6%	\$545,569.67	101%	\$503,526.44	\$831,280.23	
2990	HB 645- ARPA	\$138,196.76	71%	\$10,310.78	100%	-\$127,885.98	\$60,255.67	
	Revolving Fund	\$0.00	0%	\$3,354.90	100%	\$3,354.90	\$64,571.93	
3400	SID 100	\$0.00	0%	\$0.00	18%	\$0.00	\$28,715.09	
3600	SID 101A	\$0.00	0%	\$0.00	0%	\$0.00	\$48,667.45	
3601	SID 101A	\$0.00	0%	\$0.00	0%	\$0.00	\$8,418.79	
3602	SID 102 SID 103	\$0.00	0%	\$0.00	0%	\$0.00	\$4,750.00	
3603		\$25,996.45	49%	\$35,017.66	63%	\$9,021.21	\$22,905.25	
3604	SID 104	\$5,540.93	4%	\$6,584.96	100%	\$1,044.03	\$110,529.94	
4010	City Hall CIP	\$70,099.05	36%	\$7,946.48	15%	-\$62,152.57	\$82,847.43	
4011	Pool CIP		14%	\$4,784.99	16%	-\$10,715.01	\$76,596.99	
4015	Parks CIP	\$15,500.00	0%	\$822.05	2%	\$822.05	\$15,822.05	
4016	Parks Facility CIP	\$0.00		\$7,806.61	100%	10,472.17	\$131,975.65	
4020	Police CIP	\$18,278.78	20,0		18%	-\$891.48	\$50,592.96	
4025	Police Investigative CIP	\$3,713.00	12%	\$2,821.52	3%	-\$182,569.80	-\$86,075.63	
4030	Street Equipment	\$187,858.00	64%	\$5,288.20	100%	\$6,002.95	\$115,539.28	
4031	Street Construction	\$0.00	0%	\$6,002.95	49%	\$47,841.00	\$920,799.23	
4040	Fire Equipment	\$0.00	0%	\$47,841.00		\$4,690.89	\$90,285.94	
4060	Bike Path Enhancement	\$0.00	0%	\$4,690.89	100%	\$1,274.35	\$24,527.42	
4070	Downtown Enhancement	\$0.00	0%	\$1,274.35	100%		\$1,174.96	
4075	Curb & Sidewalk	\$9,425.00	91%	\$0.00	0%	-\$9,425.00	\$6,651,263.72	
5210	Water Utility	\$1,306,844.03	13%	\$1,991,015.12	18%	\$684,171.09	\$307,891.38	
5211	Water Impact Fees	\$761.00	0%	\$21,801.11	138%	\$0.00		
5310	Sewer Utiltiy	\$1,812,675.53	23%	\$1,940,813.65	29%	\$128,138.12	\$4,345,302.68	
5311	Sewer Impact Fees	\$761.00	1%	\$13,900.70	138%	\$13,139.70	\$155,831.53	
5410	Solid Waste	\$1,002,754.60	74%	\$676,728.06	50%	-\$326,026.54	\$237,771.93	
5710	Sweeping Operating	\$422,479.53	74%	\$230,820.20	66%	-\$191,659.33	\$353,172.37	
7060	Playgrounds & Parks	\$0.00	0%	\$0.00	0%	\$0.00	\$0.00	
7120	Fire Disability	\$42,500.00	47%	\$57,239.30	70%	\$14,739.30	\$25,672.40	
	Grant-Richland County	\$0.00	0%	\$896.42	0%	\$896.42	\$17,969.24	
7970	Grant-Richard County	40.00						
	Totals	\$9,192,734.81	30%	\$9,305,398.68	32%	\$112,663.87	\$18,556,521.46	

04/14/25 10:54:53

Page: 1 of 3 Report ID: B100F

CITY OF SIDNEY
Statement of Expenditure - Budget vs. Actual Report
For the Accounting Period: 3 / 25

Fund	Committed Current Month	Committed YTD	Original Appropriation	Current Appropriation	Available Appropriation	Comm.
	193,673.84	2,023,499.83	3,952,428.00	3,952,428.00	1,928,928.17	51 %
2060 PLAYGROUNDS & PARKS	0.00	0.00	30,000.00	30,000.00	30,000.00	%
2061 BALLPARKS & BALLFIELDS	0.00	0.00	19,000.00	19,000.00	19,000.00	%
2062 TENNIS COURTS	0.00	39,520.00	230,000.00	230,000.00	190,480.00	17%
2063 BIKE PATH	0.00	1,500.00	97,400.30	97,400.00	95,900.00	2%
2101 TBID	0.00	63,521.27	300,000.00	300,000.00	236,478.73	8 21%
2170 Airport	0.00	0.00	19,958.00	19,958.00	19,958.00	0%
2190 Comprehensive Liability	0.00	51,586.50	51,587.00	51,587.00		0.50 100%
2220 Library Levy	0.00	0.00	25,000.00	25,000.00	25,000.00	0 0%
2260 Emergency Disaster	1,713.50	1,713.50	49,135.00	49,135.00	47,421.50	0 3%
2350 Local Govt Study Commission	0.00	825.00	31,000.00	31,000.00	30,175.00	0 3%
2370 P.E.R.S Employer Contribution	13,013.73	132,585.54	295,016.00	295,016.00	162,430.46	6 45%
2371 Employer Contribution Group Health	23,539.40	221,206.69	416,729.00	416,729.00	195,522.31	1 53%
2372 Permissive Health LEvy	0.00	0.00	2,700.00	2,700.00	2,700.00	0 0%
2390 Drug Forfeiture	80.06	10,047.01	25,000.00	25,000.00	14,952.99	9 40%
2399 Impact Fees	0.00	0.00	310,990.00	310,990.00	310,990.00	0 0%
2425 Street Lighting	11,624.85	97,047.24	195,500.00	195,500.00	98,452.76	6 50%
2550 Tree Removal – Dutch Elm Disease	0.00	0.00	4,600.00	4,600.00	4,600.00	0 0%
2565 City Wide Street Maintenance	32,400.73	361,800.27	539,261.00	539,261.00	177,460.73	3 67%
2566 SNOW REMOVAL	13,954.47	99,237.13	203,553.00	203,553.00	104,315.87	7 49%
2584 Mowing	0.00	2,238.53	60,000.00	60,000.00	57,761.47	17 4%
2598 MVS Park Maintenance #98	0.00	1,650.00	12,000.00	12,000.00	10,350.00	0 14%
2810 Police Reserve Training	0.00	5,268.14	20,000.00	20,000.00	0 14,731.86	36 26%
2820 Gas Apportionment Tax	0.00	900,469.25	1,117,628.00	1,117,628.00	0 217,158.75	75 81%
2861 MAIN STREET MT GRANT	0.00	49,548.77	50,000.00	50,000.00	0 451.23	23 99%

04/14/25 10:54:53

CITY OF SIDNEY
Statement of Expenditure - Budget vs. Actual Report
For the Accounting Period: 3 / 25

Page: 2 of 3 Report ID: B100F

	Committed	Committed	Original	Current		¢/p
Fund	Current Month	YTD	Appropriation	Appropriation	- 1 - 2	0
2869 Nuisance	0.00	24,043.25	50,000.00	50,000.00	25,956.75	48%
2890 Oil/Gas Severance	0.00	42,043.23	656,100.00	656,100.00	614,056.77	<i>⇔</i>
2990 ARFA	0.00	138,196.76	193,333.00	193,333.00	55,136.24	71%
3600 SID 100 SMV Paving	0.00	0.00	28,715.00	28,715.00	28,715.00	0%
3601 SID 101A	0.00	0.00	48,667.00	48,667.00	48,667.00	%
3604 SID #104	0.00	25,996.45	53,000.00	53,000.00	27,003.55	49%
4010 City Hall CIP	2,000.00	5,540.93	124,412.00	124,412.00	118,871.07	42
4011 POOL CIP	0.00	70,099.05	197,000.00	197,000.00	126,900.95	36%
4015 Parks CIP	0.00	15,500.00	114,500.00	114,500.00	99,000.00	14%
4016 PARKS FACILITY CIP	0.00	0.00	15,000.00	15,000.00	15,000.00	%0
4020 Police CIP	0.00	18,278.78	90,000.00	90,000.00	71,721.22	20%
4025 Police Investigative CIP	0.00	3,713.00	30,000.00	30,000.00	26,287.00	12%
4030 Cap Proj-Street Equipment	0.00	187,858.00	291,700.00	291,700.00	103,842.00	64%
4031 Cap Proj-Street Construction	0.00	0.00	83,000.00	83,000.00	83,000.00	0%
4040 Capital Projects - Fire Equipment	0.00	0.00	40,000.00	40,000.00	40,000.00	%
4060 Enhancement Project-CTEP-Bike Path	0.00	0.00	65,000.00	65,000.00	65,000.00	0%
4070 Downtown Enhancement Capital	0.00	0.00	12,500.00	12,500.00	12,500.00	%
4075 Curb & Sidewalk	0.00	9,425.00	10,324.00	10,324.00	899.00	91%
5210 Water Utility	137,744.63	1,306,844.03	10,357,084.00	10,357,084.00	9,050,239.97	ب س 4
5211 WATER IMPACT FEES	0.00	761.00	297,225.00	297,225.00	296,464.00	0 %
5310 Sewer Utility	82,907.39	1,812,675.53	7,764,738.00	7,764,738.00	5,952,062.47	23 %
5311 SEWER IMPACT FEES	0.00	761.00	146,315.00	146,315.00	145,554.00	1%
5410 Solid Waste	73,173.91	1,002,754.60	1,363,493.00	1,363,493.00	360,738.40	74%
5710 Sweeping Operating	23,836.43	422,479.53	567,403.00	567,403.00	144,923.47	74%
7120 Fire Disability	0.00	42,500.00	90,000.00	90,000.00	47,500.00	47%

Item c.

04/14/25 10:54:53

CITY OF SIDNEY
Statement of Expenditure - Budget vs. Actual Report
For the Accounting Period: 3 / 25

Page: 3 of 3 Report ID: B100F

					3	all all
	Committed	Committed	Original	Current	Available	) o¥
Fund	Current Month	YTD	Appropriation	Appropriation	Appropriation	Comm.
7970 Grant-Richland County	0.00	0.00	5,223.00	5,223.00	5,223.00 0%	0 0%
Grand Total:	609,662.94	9,192,734.81	30,753,217.00	9,192,734.81 30,753,217.00 30,753,217.00	21,560,482.19 30%	9 30%

04/14/25 13:08:23

CITY OF SIDNEY
Statement of Revenue Budget vs Actuals
For the Accounting Period: 3 / 25

Page: 1 of 2 Report ID: B110F

Franci	Received Current Month	Received YTD I	Estimated Revenue	Revenue % To Be Received Received
1000 General	310,720.66	2,172,049.36	3,952,951.00	1,780,901.64 55 %
2060 PLAYGROUNDS & PARKS	0.00	1,623.22	1,623.00	-0.22 100 %
2061 BALLPARKS & BALLFIELDS	0.00	1,236.59	1,237.00	0.41 100 %
2062 TENNIS COURTS	0.00	8,111.33	83,111.00	74,999.67 10 %
2063 BIKE PATH	0.00	4,652.89	14,563.00	9,910.11 32 %
2101 TBID	0.00	80,480.74	301,180.00	220,699.26 27 %
2170 Airport	48.11	10,860.87	17,613.00	6,752.13 62 %
2190 Comprehensive Liability	144.45	26,009.57	38,791.00	12,781.43 67 %
2220 Library Levy	16.86	2,637.59	1,408.00	-1,229.59 187 %
2260 Emergency Disaster	17.82	5,809.91	7,780.00	1,970.09 75 %
2350 Local Govt Study Commission	74.28	18,200.42	31,002.00	12,801.58 59 %
2370 P.E.R.S Employer Contribution	498.84	104,342.25	160,156.00	55,813.75 65 %
2371 Employer Contribution Group Health	866.57	171,074.51	288,673.00	117,598.49 59 %
2372 Permissive Health LEvy	1.16	273.40	141.00	-132.40 194 %
2390 Drug Forfeiture	42.50	2,615.26	14,280.00	11,664.74 18 %
2399 Impact Fees	578.53	17,053.26	16,417.00	-636.26 104 %
2425 Street Lighting	738.84	118,949.76	163,501.00	44,551.24 73 %
2550 Tree Removal - Dutch Elm Disease	0.00	250.94	251.00	0.06 100 %
2565 City Wide Street Maintenance	3,272.18	610,418.73	465,702.00	-144,716.73 131 %
2566 SNOW REMOVAL	0.00	6,340.88	220,441.00	214,100.12 3 %
2584 Mowing	0.00	10,668.58	35,412.00	24,743.42 30 %
2598 MVS Park Maintenance #98	18.15	3,367.30	2,650.00	-717.30 127 %
2810 Police Reserve Training	0,00	939.83	16,940.00	16,000.17 6 %
2820 Gas Apportionment Tax	20,170.09	252,653.65	313,639.00	60,985.35 81 %
2861 MAIN STREET MT GRANT	0.00	49,548.77	50,000.00	451.23 99 %
2869 Nuisance	10.00	1,897.50	65,000.00	63,102.50 3 %

04/14/25 13:08:23

CITY OF SIDNEY
Statement of Revenue Budget vs Actuals
For the Accounting Period: 3 / 25

Page: 2 of 2 Report ID: B110F

3.00 667,214.94 50 0.00 116,499.80 66 0.00 25,050.70 70 0.00 -896.42			c	7970 Grant-Richland County
667,214.94 116,499.80	82,29	0 57,239.30 0 896.42	205.10	7120 Fire Disability
667,214.94	ω	9 230,820.20	992.19	5710 Sweeping Operating
	6 1,343,943.00	1 676,728.06	5,825.61	5410 Solid Waste
1.00 -5,999.70 176	0 7,901.00	0 13,900.70	3,000.00	5311 SEWER IMPACT FEES
6.00 4,694,252.35 29	5 6,635,066.00	3 1,940,813.65	212,570.43	5310 Sewer Utility
1.00 -6,000.11 138	15,801.00	21,801.11	2,000.00	5211 WATER IMPACT FEES
6.00 8,828,140.88 18	2 10,819,156.00	1,991,015.12	218,980.00	5210 Water Utility
4.00 -0.35 100	5 1,274.00	1,274.35	0.00	4070 Downtown Enhancement Capital Project
1.00 0.11 100	9 4,691.00	4,690.89	0.00	4060 Enhancement Project-CTEP-Bike Path
1.00 50,000.00 49	97,841.00	97,841.00	0.00	4040 Capital Projects - Fire Equipment
3.00 0.05 100 %	6,003.00	6,002.95	0.00	4031 Cap Proj-Street Construction
3.00 193,999.80	199,288.00	5,288.20	0.00	4030 Cap Proj-Street Equipment
2.00 13,000.48 18	15,822.00	2,821.52	0.00	4025 Police Investigative CIP
7.00 0.39 100	7,807.00	7,806.61	0.00	4020 Police CIP
2.00 49,999.95	50,822.00	822.05	0.00	4016 PARKS FACILITY CIP
25,000.01 16	29,785.00	4,784.99	0.00	4015 Parks CIP
14,999.52 15	52,946.00	7,946.48	0.00	4011 POOL CIP
0.04 100 %	6,585.00	6,584.96	0.00	4010 City Hall CIP
.00 20,843.34 63 %	55,861.00	35,017.66	0.00	3604 SID #104
.00 0.10 100 %	3,355.00	3,354.90	0.00	3400 Revolving Fund
.00 0.22 100 %	10,311.00	10,310.78	0.00	2990 ARPA
.00 -7,607.67 101 %	537,962.00	545,569.67	1,391.34	2890 Oil/Gas Severance
To Be R	Estimated Revenue	Received YTD	Received Current Month	Fund

CITY OF SIDNEY

Cash Report

For the Accounting Period: 3/25

Page: 1 of 5 Report ID: L160

			B		Transfers	Ending
Fund/Account	Beginning Balance	Received	In	Disbursed	Out	Balance
1000 General						7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7
h - Operating	1,388,262.56	310,720.66	0.00	0.00	224,058.70	1,4/4,924.32
TRICTED CASH ACCOUNT	1,063.67	0.00	0.00	0.00	0,00	-6.863.29
102000 Cash - Restricted	-6,863.29	0.00	0.00	0.00	0.00	-8,227.00
102250 Cash-Capital Equipment	-8,227.00	0.00	0.00	0.00	0.00	125.00
	125.00	0.00	0.00	0.00	0.00	150.00
	150.00	0.00	0.00		224,058.70	1,461,172.90
tal Fund	1,374,510.94	310,720.66				
2060 FLAYGROUNDS & PARKS			0	0.00	0.00	31,242.27
101000 Cash - Operating	31,242.27	0.00		•		
2061 BALLPARKS & BALLFIELDS	22 800 75	0 00	0.00	0.00	0.00	23,800.75
101000 Cash - Operating	23,800.75					
	116 500 56	0.00	0.00	0.00	0.00	116,599.56
101000 Cash - Operating	11070000					
101000 Cash - Operating	86,412.23	0.00	0.00	0.00	0.00	86,412.23
					0 00	29-968-60
101000 Cash - Operating	29,968.60	0.00	0.00			
2170 Airport	1		0 00	0.00	0.00	15,970.63
101000 Cash - Operating	15,922.52	48.11				
101000 Cash - Operating	-9,891.09	144.45	0.00	0.00	0.00	-9,746.64
2220 Library Levy						26 506 22
101000 Cash - Operating	26,489.36	16.86	0.00	0.00		0000
2260 Emergency Disaster		1	0 00	0.00	1,713,50	47,349.54
101000 Cash - Operating	49,045.22	1/.02				
	14 04	0.00	0.00	0.00	0.00	14.04
101000 Casn - Operating	-  -  -  -  -	:				
101000 Cash - Operating	17,300.82	74.28	0.00	0.00	0.00	17,3/5.10
2370 P.E.R.S Employer Contribution					13 013 73	109,196,22
101000 Cash - Operating	121,711.11	498.84	0.00		F ( )	
	1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	7 7 7 9	0.00	0.00	23,539.40	80,664.74
	103,337.57	000.07				
23/2 Permissive nearth Fry	2,843.41	1.16	0.00	0.00	0.00	2,844.57
					130 6/	CU 289 CE
101000 Cash - Operating	32,779.16	42.50	0.00	0.00	- CO	
2399 Impact Fees				0 00	0.00	316,617.50
101000 Cash - Operating	316,038.97	5/8.53	0.00			
2425 Street Lighting				0 00	11.624.85	410,740.93
101000 Cash - Operating	421,626.94	738.84	0.00		++	
2550 Tree Removal - Dutch Elm Disease			0 00	0.00	0.00	4,829.95
101000 Cash - Operating	4,829.95	0.00	0.00			,
	200	0 00	0.00	0.00	0.00	1,428.40
101000 Cash - Operating	1,428.40					

CITY OF SIDNEY

Cash Report

For the Accounting Period: 3/25

Page: 2 of 5 Report ID: L160

CITY OF SIDNEY

Cash Report

For the Accounting Period: 3/25

Page: 3 of 5 Report ID: L160

			3		Transfers	Ending
Fund/Account	Balance	Received	In	Disbursed	Out	Balance
101000 Cash - Operating	82,847.43	0.00	0.00	0.00	0.00	82,847.43
4015 Parks CIP				0.00	0.00	74,596.99
101000 Cash - Operating	74,596.99	0.00	0.00	0.00	0.00	2,000.00
102000 Cash - Restricted	2,000.00	0.00		(		76,596.99
Total Fund	76,596.99					
4016 PARKS FACILITY CIP		>	0 00	0.00	0.00	15,822.05
101000 Cash - Operating	15,822.05	0.00	· · · · · · · · · · · · · · · · · · ·	1		
4020 Police CIP			0 00	0.00	2,675.00	214,321.23
101000 Cash - Operating	216,996.23	0.00	0 .00	0 00		1,313.42
101240 UNRESTRICTED CASH ACCOUNT	1,313.42	0.00	0.00	0.00	0.00	5,275.00
101250 UNRESTRICTED CASH ACCOUNT	5,275.00	0.00	0.00	0.00	0 00	-88.934.00
102250 Cash-Capital Equipment	-88,934.00	0.00	0.00	0.00		131.975.65
Total Fund	134,650.65				2,010.00	101
4025 Police Investigative CIP		É			3, 713,00	52,915.74
101000 Cash - Operating	56,628.74	0.00	0.00	0.00		-2,322.78
102250 Cash-Capital Equipment	-2,322.78	0.00	0.00		3 713 00	50.592.96
Total Fund	54,305.96					
4030 Cap Proj-Street Equipment			0 00	0.00	0.00	84,248.59
	5 708 78	0.00	0.00	0.00	0.00	5,708.78
101240 UNRESTRICTED CASH ACCOUNT	10,625.00	0.00	0.00	0.00	0.00	10,625.00
102250 Cash-Capital Equipment	-186,658.00	0.00	0.00	0.00	0.00	-186,658.00
	-86,075.63					-00,0,0,0
4031 Cap Proj-Street Construction					0	-2.910.23
101000 Cash - Operating	-2,910.23	0.00	0.00	0.00	0.00	93,174.51
101240 UNRESTRICTED CASH ACCOUNT	93,174.51	0.00	0.00	0.00	0 00	10,625.00
101250 UNRESTRICTED CASH ACCOUNT	10,625.00	0.00	0.00	0.00	0.00	14.650.00
102250 Cash-Capital Equipment	14,650.00	0.00	0.00	0.00	0.00	115 539.28
Total Fund	115,539.28					-
4040 Capital Projects - Fire Equipment				0 00	0.00	718,992.97
Cash - Operat	718,992.97	> 00	0.00	0.00	0.00	104,851.46
UNRESTRICTED	104,851.46	0 00	0 00	0.00	0,00	19,125.00
101250 UNRESTRICTED CASH ACCOUNT	19,125.00	0.00	0.00	0.00	0.00	77,829.80
102250 Cash-Capital Equipment	77,829.80					920,799.23
Total Fund	920,799.23					
4060 Enhancement Project-CTEP-Bike Path	\ > >	0	0 00	0.00	0.00	6,890.89
101000 Cash - Operating	0,090.00	0 . 00	0 00	0.00	0.00	64,120.05
UNRESTRICTED	10 575 00	0.00	0,00	0.00	0.00	10,575.00
UNRESTRICTED	8.700.00	0.00	0.00	0.00	0.00	8,700.00
Total Fund	90,285.94					90,285.94
4070 Downtown Enhancement Capital Project	it		,			8 210 38
101000 Cash - Operating	8,210.38	0,00	0.00	0.00	0.00	8,210.00
101240 UNRESTRICTED CASH ACCOUNT	5,567.04	0.00	0.00	0.00	0.00	0,007.04
101250 UNRESTRICTED CASH ACCOUNT	8,350.00	0.00	0.00	0.00	0.00	3 400 00
102250 Cash-Capital Equipment	2,400.00	0.00	0.00	0.00	0.00	2, 400.00
Total Fund	24,527.42					75,170,47

CITY OF SIDNEY

Cash Report

For the Accounting Period: 3/25

Page: 4 of 5 Report ID: L160

20,517,505.28	636,771.89	249,505.61	636,771.89	783,927.44	19,983,083.45	Totals
17,969.24	0.00	0.00	0.00	0.00	17,969.24	7970 Grant-Richland County 101000 Cash - Operating
1,696,864.26	0.00	26,874.44	320,671.27	0.00	1,403,067.43	7930 Claims 101000 Cash - Operating
271,658.04	0.00	222,631.17	313,070.91	0.00	181,218.30	7910 Payroll 101000 Cash - Operating
-15,308.62	320.00	0.00	0.00	320.00	-15,308.62	7467 City Court - MT Law Enf. Academy 101000 Cash - Operating
-41.00	220.00	0.00	0.00	220.00	-41.00	
25,672.40	0,00	0.00	0.00	205.10	25,467.30	7120 Fire Disability 101000 Cash - Operating
6,389.30	0.00	0.00	0.00	0.00	6,389.30	7075 Swim Pool Handicapped Endowment 101000 Cash - Operating
353,172.37	21,536.44	0.00	0,00	992.19	373,716.62	5710 Sweeping Operating 101000 Cash - Operating
237,771.93	62,361.93	0.00	0.00	5,825.61	294,308.25	5410 Solid Waste 101000 Cash - Operating
155,831.53	0.00	0.00	0.00	3,000.00	152,831.53	5311 SEWER IMPACT FEES 101000 Cash - Operating
4,345,302.68	73,284.35		2,995.53	210,254.24	4,205,337.26	Total Fund
-36,633.34	0.00	0.00	0.00	0.00	-36,633.34	102240 Cash-Replacement &
-1,162.50	0.00	0.00	0.00	0.00	1 162 50	
797,668.00	0.00	0.00	0.00	0.00	1,000,230.11	
1,066,236.41	-	0.00	0.00	00 0 67.6C7'0T7	2,379,228.69	101000 Cash - Operating
2,519,194.11	73.284.35	0 00	) ) ) ) ) )			5310 Sewer Utility
307,891.38	0.00	0.00	0.00	2,000.00	305,891.38	5211 WATER IMPACT FEES 101000 Cash - Operating
0,031,203.72	130,418.33		34,18	222,007.92	6,559,639.95	Total Fund
225.00		0.00	0.00	0.00	225.00	
288,507.00	0.00	0.00	0.00	0.00	288,507.00	
192,316.00	0.00	0.00	0.00	0.00	192,316.00	102200 Cash-Restricted for Bond
-5,437.39	0.00	0.00	0.00	0.00	-5,437.39	
1,047,208.12	0.00	0.00	0.00	0.00	1.047.208.12	101235 UNRESTRICTED CASH ACCOUNT
80,736.41	0.00	0.00	0.00	0.00	90,736 41	101000 Cash - Operating
5,047,708.58	130,418.33	0.00	34.18	222.007.92	/ 956 ORA 81	5210 Water Utility
0.21	0.00	0.00	0.00	0.00	0.21	4204 SID #104 101000 Cash - Operating
1,1,2.00					1,174.96	Total Fund
1 17/ 96	0.00	0.00	0.00	0.00	-69,611.03	
2,865.50	0.00	0.00	0.00	0.00	2,865.50	101000 Cash - Operating
67,920.49	0.00	0.00	0.00	0,00	67 920 49	رن الا
						#unu/ Account
Balance	Out	Disbursed	Transfers In	Received	Beginning Balance	ゴニーレ ヘットン・オー
Ending	T Constore					

<sup>\*\*\*</sup> Transfers In and Transfers Out columns should match, with the following exceptions:

1) Cancelled electronic checks increase the Transfers In column. Disbursed column will be overstated by the same amount and will not balance to the Redeemed Checks List.
2) Payroll Journal Vouchers including local deductions with receipt accounting will reduce the Transfers Out column by the total amount of these checks.

# Denning, Downey & Associates, P.C. CERTIFIED PUBLIC ACCOUNTANTS

P.O. Box 1957 Kalispell, MT 59903-1957 (406) 756-6879 • FAX (406) 257-7879 • E-Mail <u>dda@ddaudit.com</u> Robert K. Denning, CPA, CGFM, CFF, CITP

April 10, 2025

City Council Jessi Chamberlin, City Clerk

City of Sidney 115 2<sup>nd</sup> Street SE Sidney, MT 59270

We are engaged to audit the financial statements of the governmental activities, the business-type activities, each major fund, and the aggregate remaining fund information of City of Sidney for the year ended June 30, 2024. Professional standards require that we provide you with the following information related to our audit. We would also appreciate the opportunity to meet with you to discuss this information further since a two-way dialogue can provide valuable information for the audit process.

Our Responsibility under U.S. Generally Accepted Auditing Standards, Government Auditing Standards, and The Uniform Guidance.

As stated in our engagement letter dated May 26, 2023, our responsibility, as described by professional standards, is to express opinions about whether the financial statements prepared by management with your oversight are fairly presented, in all material respects, in conformity with U.S. generally accepted accounting principles. Our audit of the financial statements does not relieve you or management of your responsibilities.

In planning and performing our audit, we will consider City of Sidney's internal control over financial reporting in order to determine our auditing procedures for the purpose of expressing our opinions on the financial statements and not to provide assurance on the internal control over financial reporting. We will also consider internal control over compliance with requirements that could have a direct and material effect on a major federal program in order to determine our auditing procedures for the purpose of expressing our opinion on compliance and to test and report on internal control over compliance in accordance with the Uniform Guidance.

As part of obtaining reasonable assurance about whether City of Sidney's financial statements are free of material misstatements, we will perform tests of City of Sidney's compliance with certain provisions of laws, regulations, contracts, and grants. However, providing an opinion on compliance with those provisions is not an objective of our audit. Also, in accordance with the Uniform Guidance, we will examine, on a test basis, evidence about City of Sidney's compliance with the types of compliance requirements described in the U.S. Office of Management and Budget (OMB) Compliance Supplement applicable to each of its major federal programs for the purpose of expressing an opinion on City of Sidney's compliance with those requirements. While our audit will provide a reasonable basis for our opinion, it will not provide a legal determination on City of Sidney's compliance with those requirements.

Our responsibility is to plan and perform the audit to obtain reasonable, but not absolute, assurance that the financial statements are free of material misstatement.

We are responsible for communicating significant matters related to the audit that are, in our professional judgment, relevant to your responsibilities in overseeing the financial reporting process. However, we are not required to design procedures specifically to identify such matters.

We are also responsible for communicating particular matters required by law, regulation, agreement, or other requirements applicable to the engagement.

We gave significant consideration to particular circumstances or relationships such as financial interests, business, or family relationships, or non-attest/non-audit services provided or expected to be provided, and have implemented safeguards to ensure independence.

Generally accepted accounting principles provide for certain required supplementary information (RSI) to supplement the basic financial statements. Our responsibility with respect to Budgetary Comparison Schedule and Budget-to-GAAP Reconciliation, Management's Discussion and Analysis (MD&A), Schedule of Changes in the Total OPEB Liability and Related Ratios, Schedule of Proportionate Share of the Net Pension Liability, and Schedule of Contributions, which supplement the basic financial statements, is to apply certain limited procedures in accordance with generally accepted auditing standards. However, the RSI will not be audited and, because the limited procedures do not provide us with sufficient evidence to express an opinion or provide any assurance, we will not express an opinion or provide any assurance on the RSI.

We have been engaged to report on the *Schedule of Expenditures of Federal Awards*, which accompany the financial statements but is not RSI. Our responsibility for this supplementary information, as described by professional standards, is to evaluate the presentation of the supplementary information in relation to the financial statements as a whole and to report on whether the supplementary information is fairly stated, in all material respects, in relation to the financial statements as a whole.

## Planned Scope, Timing of the Audit, and Other

An Audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements; therefore, our audit will involve judgment about the number of transactions to be examined and the areas to be tested.

Our audit will include obtaining an understanding to the entity and its environment, including internal control, sufficient to assess the risks of material misstatement of the financial statements and to design the nature, timing, and extent of further audit procedures. Material misstatements may result from (1) errors, (2) fraudulent financial reporting, (3) misappropriation of assets, or (4) violations of laws or governmental regulations that are attributable to the entity or to the acts by management or employees acting on behalf of the entity. We will generally communicate our significant findings at the conclusion of the audit. However, some matters could be communicated sooner, particularly if significant difficulties are encountered during the audit where assistance is needed to overcome the difficulties or if the difficulties may lead to modified opinion. We will also communicate any internal control related matters that are required to be communicated under professional standards.

We expect to begin our audit on approximately April 10, 2025 and issue our report on approximately May 31, 2025. Robert Denning is the engagement partner and is responsible for supervising the engagement and signing the report or authorizing another individual to sign it.

This information is intended solely for the use of City Council and management of City of Sidney and is not intended to be, and should not be, used by anyone other than these specified parties.

Very truly yours,

Jonathan Mahrt, CPA

Denning, Downey & Associates, P.C.

 04/18/25
 CITY OF SIDNEY
 Page: 1 of 6

 12:26:55
 Claim Approval List
 Report ID: AP100

For the Accounting Period: 4/25

Claim	Check Vendor #/Name/	Document \$/ Disc \$					Cash
	Invoice #/Inv Date/Description	on Line \$	PO #	Fund	Org Acct	Object Proj	Account
43778	3 1172 MARGIE'S CAR WASH	20.00					
	86794265 03/30/25 HERMANS CARWASH	20.00		5310	430600	300	101000
43779	3 MONTANA DAKOTA UTILITI	ES 15,861.84					
	04/07/25 FIREHALL- 115 2ND ST SE	850.87		1000	411200	340	101000
	04/07/25 1105 3RD ST NW- GENERATOR	86.19		1000	420400	300	101000
	04/07/25 1105 3RD ST NW- FIREHALL	1,961.31		1000	420400	300	101000
	04/07/25 WELL #10	129.54		5210	430500	300	101000
	04/04/25 2ND AVE & 3RD ST	21.89		2425	430263	300	101000
	04/07/25 SWIMMING POOL	370.14		1000	460445	300	101000
	04/07/25 BASEBALL FIELD	941.32		2425	430263	300	101000
	04/04/25 VILLAGE SQUARE LIGHT	17.79		2425	430263	300	101000
	04/04/25 STREET LIGHTS	8,788.65		2425	430263	300	101000
	04/07/25 PARK PAVILLION	72.85*		1000	460430	300	101000
	04/07/25 WATER TREATMENT PLANT	2,316.87		5210	430500	300	101000
	04/07/25 CENTRAL PARK LIGHTS	63.45		2425	430263	300	101000
	04/07/25 NEW TENNIS COURTS	21.81		2425	430263	300	101000
	04/07/25 CC PARK LOT	18.48		2425	430263	300	101000
	04/07/25 202 S CENTRAL AVE	21.81		2425	430263	300	101000
	04/01/25 QUILLING PARK	178.87*		1000	460430	300	101000
43780	2 LOWER YELLOWSTONE R.E.	A. 6,005.57					
	04/02/25 WATER TANK	133.83	NA	5210	430500	300	101000
	04/02/25 3-PHASE	663.04	NA	5310	430600	300	101000
	04/02/25 SIDNEY LAGOON	3,872.00	NA	5310	430600	300	101000
	04/02/25 LAGOON	1,336.70	NA	5310	430600	300	101000
43781	1398 RINKER MATERIALS	2,216.79					
	30947376 04/03/25 GRATE AND FRAME	2,216.79		5310	430600	200	101000
43782	2 402 UTILITIES UNDERGROUND	LOCATION 68.25					
	5035110 03/31/25 EXCAVATION NOTIFICAT:	IONS 39 34.13		5210	430500	300	101000
	5035110 03/31/25 EXCAVATION NOTIFICAT	IONS 34.12		5310	430600	300	101000
43783	3 1078 BIG SKY SURVEYING	1,550.00					
	33 04/03/25 LOCATE CULVERTS & CROSS SI	ECT 1,550.00		5310	430600	300	101000
43784							
	153427 04/02/25 BOWL CLEANER	384.94		5310	430600	200	101000

 04/18/25
 CITY OF SIDNEY
 Page: 2 of 6

 12:26:55
 Claim Approval List
 Report ID: AP100

For the Accounting Period: 4/25

Claim	Check Vendor #/Name/		Disc \$					Cash
	Invoice #/Inv Date/Description	Line \$		PO #	Fund Org	Acct	Object Proj	Account
43785	489 YELLOWSTONE CHIROPRACTIC (	CLINIC 100.00						
	5932 04/02/25 JARED JURGENS	100.00			5310	430600	300	101000
43786	1174 VALLI	126.39						
	99888 03/31/25 WEB POSTING	63.20			5210	430500	300	101000
	99888 03/31/25 WEB POSTING	63.19			5310	430600	300	101000
43787	38 NORMONT EQUIPMENT	337.98						
	33509 04/04/25 HOSE, CLAMPS, RAKES	337.98*			1000	430200	200	101000
25957								
43788	537 MARC	177.29						
	0844836 03/27/25 DEODORIZER	177.29*			1000	430200	200	101000
25930								
43789	1470 MCE	161.02						
	9292015 04/04/25 HOSE ASSEMBLY	161.02			5410	430830	200	101000
25931								
43790	1159 TBID	2,215.50						
	W9334 04/11/25 HOLIDAY INN TBID 1ST Q	2,215.50			2101	460440	700	101000
43791	995 MORRISON MAIERLE, INC.	33,265.84						
	258266 04/08/25 ANDERSON SUBDIVISION	1,020.00*			5310	430600	950	101000
	258232 04/07/25 PHASE IV SLUDGE REMOVAL	3,021.25			5310	430600	300	101000
	258278 04/14/25 MEADOW SUBDIVISION	21,552.59*			5310	430600	950	101000
	258055 02/06/25 MEADOW SUBDIVISION	7,672.00*			5310	430600	950	101000
43792	3 MONTANA DAKOTA UTILITIES	17.00						
	04/09/25 SWIMMING POOL	8.59			1000	460445	300	101000
	04/09/25 BIKE PATH	8.41			2425	430263	300	101000
43793	1206 FEDERAL LICENSING, INC	135.00						
	04/14/25 ANNUNAL RADIO LICENSE	135.00*		NA	5410	430830	300	101000
43794	1229 KALIL LAW FIRM	12,845.00						
	4477 04/09/25 MARCH BILLING- GENERAL	3,690.00*			2565	430200	300	101000
	4478 04/09/25 MARCH PROSECUTION 2025	9,155.00*			2565	430200	300	101000

 04/18/25
 CITY OF SIDNEY
 Page: 3 of 6

 12:26:55
 Claim Approval List
 Report ID: AP100

For the Accounting Period: 4/25

Claim	Check Vendor #/Name/	Document \$/ Disc \$					Cash
	Invoice #/Inv Date/Description	Line \$	PO #	Fund Org	Acct	Object Proj	Account
43795	E 509 CARDMEMBER SERVICE - VISA	6,017.97					
	02/25/25 NORBY- GO DADDY	22.17		1000	411200	300	101000
	03/12/25 NORBY- ZOOM	48.57		1000	411200	300	101000
	02/26/25 HINTZ- MCDONALDS	22.84		5310	430600	200	101000
	02/26/25 HINTZ- USPS	1,269.87		5210	430500	200	101000
	02/26/25 HINTZ- USPS	1,269.88		5310	430600	200	101000
	03/12/25 ELLETSON- BATTERY & AMAZON	111.14		5210	430500	200	101000
25496							
	02/25/25 ELLETSON- GREAT FALLS TRAINING	1,021.82		5210	430500	300	101000
	02/13/25 HARRIS- PETERBILT	264.26		5410	430830	200	101000
25535							
	03/06/25 HARRIS- TRAFFIC SUPPLY	62.00		2565	430200	200	101000
25916							
	02/21/25 RASMUSSEN- AMAZON	115.09		1000	420400	200	101000
	03/06/25 RASMUSSEN- IAAI & NFPA	348.00		1000	420400	300	101000
	03/13/25 BAST- EPSALES- CREDIT	-305.72		5310	430600	300	101000
CREDIT	ľ						
	03/14/25 BAST- EPSALES & TSC	655.99		5310	430600	200	101000
	03/18/25 BAST- USPS & GREAT FALLS TRAIN	103.70		5310	430600	300	101000
	02/21/25 CHAMBERLIN- AMAZON	31.90		1000	410210	200	101000
	02/21/25 CHAMBERLIN- MIICROSOFT & INTUI	467.75		5210	430500	300	101000
	02/21/25 CHAMBERLIN- MICROSOFT & INTUI	467.75		5310	430600	300	101000
	03/19/25 HINTZ- SALON 27	40.96		2565	430200	200	101000
43796	1231 ADAM SMITH	500.00					
	04/01/25 PERSONAL VEHICLE USE	500.00		1000	420400	300	101000
43797		350.00					
	04/01/25 PERSONAL VEHICLE USE	350.00		1000	420400	300	101000
43798	1362 JOHN SEITZ	350.00					
	04/01/25 PERSONAL VEHICLE USE	350.00		1000	420400	300	101000
43799		24.86					
	20411 03/31/25 FUEL- CITY UNITS	12.43		1000	420400	230	101000
	20411 03/31/25 FUEL- CITY UNITS	12.43		1000	420531	230	101000
43800	180 ADVANCED COMMUNICATIONS	411.00					
	1375 $03/11/25$ LABOR TO INSTALL OPTICOM 723	282.00		4040	420400		101000
	$1368\ 03/11/25\ \text{LABOR}$ TO INSTALL OPTICOM $729$	129.00		4040	420400	940	101000

 04/18/25
 CITY OF SIDNEY
 Page: 4 of 6

 12:26:55
 Claim Approval List
 Report ID: AP100

For the Accounting Period: 4/25

Claim	Check Vendor #/Name/	Document \$/	Disc \$					Cash
	Invoice #/Inv Date/Description	Line \$		PO #	Fund Org	Acct	Object Proj	Account
43802	E 1122 STOCKMAN BANK - BANK FEES	2,518.98						
	03/31/25 SERVICE CHARGE #0173	30.00			5310	430600	300	101000
	03/31/25 SERVICE CHARGE #0486	30.00			5310	430600	300	101000
	04/01/25 SERVICE CHARGE #5410	30.00			5310	430600	300	101000
	03/13/25 RETURN W/S NSF ITEMS	81.37			5310	430600	300	101000
	04/01/25 ACH FILE FEES & ENTRY FEES	41.00			5310	430600	300	101000
	03/03/25 SERVICE CHARGE #34889	2,306.61			5310	430600	300	101000
43803	83 ACTION AUTO INC.	1,362.82						
	75757 04/11/25 REPAIR TRANSMISSION RAM 2500	1,362.82			5210	430500	200	101000
25967								
43804	12 CROSS PETROLEUM	940.00						
		313.33			5310	430600	200	101000
25963	127807 04/01/25 DIESEL EXHAUST FLUID	313.33			5410	430830	200	101000
25963								
	127807 04/01/25 DIESEL EXHAUST FLUID	313.34			2565	430200	200	101000
43805	1192 ASPHALT ZIPPER	2,020.25						
	SO16065 03/18/25 TEETH BLOCKS AND TEETH	2,020.25			2565	430200	200	101000
25917								
43806	1483 JB INDUSTRIAL INC.	3,276.00						
	11501 04/03/25 PLEXI MELT	3,276.00*			2820	430200	200	101000
25932								
43807	1064 FASTENAL COMPANY	213.18						
	NDWAT21736 01/31/25 MISC. SUPPLIES	213.18			2565	430200	200	101000
25516								
43808		493.25						
	2550448564 03/04/25 CITY HALL RUGS CLEANED	178.63			1000	411200		101000
	2550451413 03/11/25 CITY SHOP RUGS CLEANED	157.31			5710	430252		101000
	2550445499 02/25/25 CITY SHOP RUGS CLEANED	157.31			5210	430500	300	101000
43809	307 MORRISON MAIERLE, INC.	29,029.88						
	000258134 03/06/25 MEADOW VILALGE SUB.	27,332.00			5310	430600	952	101000
	000258131 03/06/25 ANDEDSON SUBDIVISON	1,697.88			5310	430600	952	101000

 04/18/25
 CITY OF SIDNEY
 Page: 5 of 6

 12:26:55
 Claim Approval List
 Report ID: AP100

For the Accounting Period: 4/25

Claim	Check Vendor #/Name/	Document \$/	Disc \$					Cash
	Invoice #/Inv Date/Description	Line \$		PO #	Fund (	Org Acct	Object Proj	Account
43810	1041 BOB HARRIS	80.00						
	04/14/25 REIMBUSEMENT- DOT PHYS	80.00*			2565	430200	300	101000
43811	1460 APSCO LLC	1,500.00						
26005	24744 04/08/25 VENTILATOR AXIAL	1,500.00*			5310	430600	930	101000
43812	307 MORRISON MAIERLE, INC.	70.00						
	000258077 02/10/25 SEWER ON CALL- SHORT PAYMEN	70.00			5310	430600	952	101000
43813	125 MONTANA RURAL WATER SYSTEMS	400.00						
	INV413 04/16/25 2025 MEMBERSHIP RENEWAL	400.00			5210	430500	300	101000
43814	E 399 VERIZON WIRELESS	901.79						
	6110294677 04/18/25 SPD CELLULAR SERVICES	901.79			1000	420100	340	101000
43815	1429 GEM CITY MOTORS- TOWING	141.00						
	0003208 04/17/25 TOWING FEES	141.00*			4025	420100	300	101000
43816	83 ACTION AUTO INC.	116.13						
	75742 04/09/25 OIL CHANGE #6264	116.13			1000	420100	210	101000
43817	E 1038 WEX BANK	2,182.71						
	103673043 04/18/25 SPD FUEL- MARCH 2025	2,182.71			1000	420100	230	101000
43818	263 BOSS INC.	279.89						
	689078-0 03/19/25 PAPER	109.90			1000	420100	200	101000
	692324-0 04/08/25 STORAGE BOXES	169.99			1000	420100	200	101000
43819	1450 STRYKER SALES, LLC	538.00						
	9208942997 04/07/25 REPLACEMENT BATTERIES FOR	198.00			1000	420100	200	101000
	9208930056 04/04/25 REPLACEMENT ELECTRODES	340.00			1000	420100	200	101000
43820	1364 UNIFORMS2GEAR	13,210.40						
	02/0421 02/13/25 BALLISTIC SHIELDS, BODY ARMOR	10,788.96			4020	420100	940	101000
	03/0954 03/31/25 HELMETS	2,421.44			4020	420100	940	101000
43821	579 PSYCHOLOGICAL RESOURCES	300.00						
	2503106 03/06/25 PRE-EMPLOYMENT EVALUTION	300.00			1000	420100	300	101000

 04/18/25
 CITY OF SIDNEY
 Page: 6 of 6

 12:26:55
 Claim Approval List
 Report ID: AP100

For the Accounting Period: 4/25

## \* ... Over spent expenditure

2141111	Check Vendor #/Name/ Invoice #/Inv Date/Description		Disc \$	PO #	Fund O	rg Acct	Object Proj	Cash Account
43822	2 1484 FEDEX	37.65						
	879641327 03/12/25 SHIPPING FEES- FOR S	SIDNEY P 37.65			1000	420100	300	101000
43823	3 1485 SIDNEY HARDWARE INC. (AC	CE 204.97						
	015757/7 03/25/25 AIR PURIFIER & FILTER	204.97			1000	420100	200	101000
43824	4 20 EAST-MONT ENTERPRISES, I	INC. 172.40						
	INV167255 03/31/25 CANDY	172.40			1000	420100	200	101000
43825	5 1410 GUARDIAN ALLIANCE TECHNO	DLOGIES 175.00						
	28118 03/31/25 PRE-EMPLOYMENT BACKGROUP	ND 175.00			1000	420100	300	101000
43826	5 1403 SHRED ND	96.75						
	17889 04/04/25 SECRURE SHREDDING SERVICE	CES 96.75			1000	420100	300	101000
43827	7 1487 RICHLAND COUNTY SHERIFF	125.00						
	104 04/17/25 TRAINING FEE	125.00			2810	420100	300	101000
43828	3 1203 TRANS UNION LLC	13.11						
	03518494 04/01/25 PRE EMPLOYMENT CREDIT				1000	420100	300	101000
43829	9 1441 JACQUELYN K. FREE, FNP-0	75.00						
	0109 04/07/25 PRE EMPLOYMENT PHYSICAL	75.00			1000	420100	300	101000
12020	77 RICHLAND COUNTY TREASURE	ER 742.00						
43030	04/18/25 SPD PRISONER BOARD- MARCH 202				1000	420200	300	101000
43831	1 1474 PURE IMAGINATION VINYL	2,678.00						
43031	1438 04/04/25 PATROL VEHICLE GRAPHICS ‡	·			4020	420100	940	101000
43832	2 628 HIGH PLAINS VET CLINIC,	INC. 125.00						
43032	03/31/25 VET FEES- ANIMAL CRUETLY	125.00			1000	420100	300	101000
12022	3 E 1262 VISA	1,307.53						
43033	04/18/25 SUPPLIES	563.63			1000	420100	200	101000
	04/18/25 PURCHASE SERVICES	465.05			1000	420100		101000
	04/18/25 DRUG FOREFITURE- PURCHASE	105.88			2390	420100		101000
	04/18/25 BROG FOREFITORE- PURCHASE 04/18/25 K9- SUPPLIES	92.97			1000	420100		101000
	04/18/25 K9- PURCHASE SERVICES	80.00			1000	420150	300	101000

Total Electronic Claims 12,928.98 Total Non-Electronic Claims 135539.95

## **City Council Meeting 4-21-25**

RC2025-03	ON HOLD
RC2025-06	ON HOLD
RC2025-10	ON HOLD
RC2025-14	ON HOLD
RC2025-15	Cody Nichols
RC2025-16	ON HOLD
RC2025-17	Brian Cummins (BLOOM)
RC2025-18	Brian Cummins (BLOOM)
RC2025-19	QCC Holding INC
RC2025-20	Joe Stevenson
RC2025-21	Kringen Family
RC2025-22	ON HOLD
RC2025-23	B&B Builders
RC2025-24	Kayla Wick
RC2025-25	MacGrady Construction
RC2025-26	Owen Nelson
2025 002	0111015
2025-002	ON HOLD
2025-005	ON HOLD
2025-005 2025-009	ON HOLD ON HOLD
2025-005 2025-009 2025-021	ON HOLD ON HOLD ON HOLD
2025-005 2025-009 2025-021 2025-026	ON HOLD ON HOLD ON HOLD
2025-005 2025-009 2025-021 2025-026 2025-027	ON HOLD ON HOLD ON HOLD ON HOLD
2025-005 2025-009 2025-021 2025-026 2025-027 2025-031	ON HOLD ON HOLD ON HOLD ON HOLD ON HOLD ON HOLD
2025-005 2025-009 2025-021 2025-026 2025-027 2025-031 2025-044	ON HOLD
2025-005 2025-009 2025-021 2025-026 2025-027 2025-031 2025-044 2025-050	ON HOLD
2025-005 2025-009 2025-021 2025-026 2025-027 2025-031 2025-044 2025-050 2025-051	ON HOLD Janet Martineau
2025-005 2025-009 2025-021 2025-026 2025-027 2025-031 2025-044 2025-050 2025-051 2025-052	ON HOLD Janet Martineau Sidney Health Center
2025-005 2025-009 2025-021 2025-026 2025-027 2025-031 2025-044 2025-050 2025-051 2025-052 2025-053	ON HOLD Janet Martineau Sidney Health Center
2025-005 2025-009 2025-021 2025-026 2025-027 2025-031 2025-044 2025-050 2025-051 2025-052	ON HOLD Janet Martineau Sidney Health Center

2025-056 Brad Fox

2025-057 MacGrady Constrcution